UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) **OF THE SECURITIES EXCHANGE ACT OF 1934**

Date of Report (Date of earliest event reported): August 12, 2024

Inspirato Incorporated

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

001-39791 (Commission File Number)

85-2426959 (I.R.S. Employer Identification Number)

1544 Wazee Street Denver, CO

(Address of principal executive offices)

(303) 839-5060

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12) \mathbf{X}

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

	Trading	Name of each exchange
Title of each class	Symbol(s)	on which registered
Class A common stock, \$0.0001 par value per share	ISPO	The Nasdaq Stock Market LLC
Warrants to purchase Class A common stock	ISPOW	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (17 CFR §230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (17 CFR §240.12b-2 of this chapter).

Emerging growth company \boxtimes

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

80202 (Zip Code)

Item 1.01 Entry into a Material Definitive Agreement.

Investment Agreement

On August 12, 2024, Inspirato Incorporated (the "**Company**") entered into an investment agreement (the "**Investment Agreement**") with One Planet Group LLC, a Delaware limited liability company (the "**Purchaser**"), relating to the issuance and sale to the Purchaser of (i) 1,335,271 shares (the "**Tranche 1 Shares**") of Class A common stock, \$0.0001 par value per share, of the Company for an aggregate purchase price of \$4,579,980 (such transaction, the "**Tranche 1 Purchase**") and (ii) 1,580,180 shares of Class A common stock (the "**Tranche 2 Shares**") for an aggregate purchase price of \$5,420,020 and an accompanying warrant (the "**Warrant**") to purchase up to 2,915,451 shares of Class A common stock (the "**Warrant Shares**") (such transaction, the "**Tranche 2 Purchase**"). In addition, pursuant to the Investment Agreement, the Purchaser was granted an option (the "**Option**") to acquire an additional number of shares of Class A common stock with an aggregate purchase price of up to \$2,500,000, where the purchase price for each share will be the same as the per share purchase price in the Tranche 1 Purchase and the Company will deliver a number of warrants equal to the number of shares of Class A common stock being purchased as part of the Option (such shares and warrants being collectively referred to as the "**Optional Securities**").

The closing of the Tranche 1 Purchase occurred on August 13, 2024 (the "**Tranche 1 Closing**"). The closing of the Tranche 2 Purchase will take place on September 13, 2024, or as soon as practicable thereafter following the satisfaction of certain closing conditions (the "**Tranche 2 Closing**," with each of the Tranche 1 Closing and the Tranche 2 Closing being referred to as a "**Closing**"). The Tranche 2 Closing is conditioned upon the approval by the Company's stockholders at a special meeting of stockholders (the "**Special Meeting**") of a proposal to authorize the issuance of the Tranche 2 Shares, the Warrant, the Warrant Shares, and the Optional Securities, to the extent such approval is required under the rules of the Nasdaq Stock Market LLC (such proposal, the "**Nasdaq Proposal**"), in addition to other customary closing conditions.

The Investment Agreement includes customary representations, warranties and covenants by the Company. Subject to certain limitations, the Investment Agreement also provides the Purchaser with the right to designate up to three members of the Company's Board of Directors (the "**Board**") and certain registration rights with respect to the Tranche 1 Shares, the Tranche 2 Shares, the Warrant Shares and the Optional Securities.

The Investment Agreement provides that, during the period from the date of the Investment Agreement until the Tranche 2 Closing or the earlier termination of the Investment Agreement in accordance with its terms, the Company is subject to certain restrictions on its ability to solicit alternative transaction proposals from third parties, provide non-public information to third parties or engage in discussions with third parties regarding alternative transaction proposals.

The Investment Agreement provides that the Tranche 2 Purchase may be terminated in certain circumstances prior to the Tranche 2 Closing, including (i) by mutual agreement of the Purchaser and the Company, with the approval of the Board; (ii) by either the Purchaser or the Company, if the Tranche 1 Purchase or the Tranche 2 Purchase has been permanently restrained, enjoined or otherwise prohibited from being consummated; (iii) automatically, if (A) the Tranche 1 Closing has not occurred on or prior to August 17, 2024 or (B) the Tranche 2 Closing has not occurred on or prior to October 11, 2024 (in each case, unless otherwise mutually agreed by the Company and the Purchaser in writing); or (iv) by either the Purchaser or the Company, if, prior to the relevant Closing, there is an uncured breach by the other party to the Investment Agreement. Upon termination of the Investment Agreement in respect of only the Tranche 2 Shares, the Warrant, the Warrant Shares and the Optional Securities, the terms of the Investment Agreement will remain in effect insofar as they relate to the Tranche 1 Purchase. The Optional Securities will be issued only if the Tranche 2 Closing occurs.

The foregoing summary of the Investment Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Investment Agreement, a copy of which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Oakstone Ventures, Inc. an affiliate of Capital One Financial Corporation and holder of the 8% Senior Secured Convertible Note due 2028 (the "Note"), waived its rights to require the Company to repurchase all or any part of the Note in connection with the transactions contemplated by the Investment Agreement (collectively, the "Transactions").

The Warrant

The Warrant will be exercisable at an exercise price of \$3.43 per share, subject to customary adjustments set forth in the Warrant for stock splits and similar transactions. The Warrant will be exercisable in whole or in part beginning on the date of the Tranche 2 Closing (the "**Commencement Date**") and until (i) the date which is five years after the Commencement Date or (ii) in the case of Fundamental Change (as defined in the Warrant) which is publicly announced before the date described in clause (i) above but which closes after the date described in clause (i) above, the closing date of such Fundamental Change. The Purchaser may exercise part or all of the Warrant via a cashless exercise mechanism set forth in the Warrant. The exercise price of the Warrant, and the number of Warrant Shares, will be adjusted proportionately if the Company subdivides its shares of Class A common stock into a greater number of shares or combines its shares of Class A common stock into a smaller number of shares. The Purchaser will have the right to receive an instrument of a successor entity that is comparable to the Warrant, or to have its Warrant repurchased, in certain circumstances involving business combination and similar transactions.

The foregoing summary of the Warrant does not purport to be complete and is qualified in its entirety by reference to the full text of the Warrant, the form of which is filed as Exhibit A to Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Voting Agreements

Contemporaneously with the execution of the Investment Agreement, certain stockholders of the Company, including the directors and executive officers of the Company, representing approximately 49% of the voting power of the Class A common stock and the Class V common stock, \$0.0001 par value per share, of the Company (collectively, the "**Voting Shares**"), entered into voting agreements (collectively, the "**Voting Agreements**") with the Company and the Purchaser, pursuant to which, among other things, such stockholders agreed to vote their respective Voting Shares in favor of the Nasdaq Proposal and not to transfer their respective Voting Shares until the date that is one business day after the record date to be set forth in the proxy statement to be provided in connection with the Nasdaq Proposal. The Voting Agreements do not preclude any director, in his or her capacity as such, from exercising his or her fiduciary duties and electing to terminate the Investment Agreement in the circumstances permitted in the Investment Agreement.

The foregoing description of the form of Voting Agreement is qualified in its entirety by reference to the full text of the Voting Agreement, the form of which is included as Exhibit 10.2 to this Current Report on Form 8-K and incorporated herein by reference.

Item 3.02 Unregistered Sales of Equity Securities.

The information related to the Investment Agreement contained in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

The Tranche 1 Shares, the Tranche 2 Shares, the Warrant, the Warrant Shares, and the Optional Securities have not been registered under the Securities Act of 1933, as amended (the "Securities Act"), in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act and/or Regulation D promulgated thereunder.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information related to the Investment Agreement contained in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

On August 13, 2024, Brad Handler stepped down as a Class III director of the Company and the Chairman of the Board, effective immediately.

On August 13, 2024, in connection with the Tranche 1 Closing and pursuant to the Investment Agreement, (i) Eric Grosse stepped down as the Chief Executive Officer and a Class I director of the Company, effective immediately; and (ii) Payam Zamani was appointed as the Chief Executive Officer, a Class I director (with a term expiring at the Company's 2026 annual meeting of stockholders) and the Chairman of the Board, effective immediately. Chief Financial Officer Robert Kaiden temporarily assumed the duties of principal executive officer following Mr. Grosse's departure through the filing of the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2024, after which Mr. Zamani assumed those duties.

In addition, on August 15, 2024, (i) John Melicharek stepped down as a Class III director of the Company, and a member of each of the Compensation Committee and Nominating and Corporate Governance Committee of the Board, effective immediately; (ii) David Kallery, the President of the Company, was appointed as a Class III director of the Company (with a term expiring at the Company's 2025 annual meeting of stockholders), effective immediately; and (iii) May Samali was appointed as a Class III director of the Company (with a term expiring at the Company's 2025 annual meeting of stockholders), effective immediately.

Neither Mr. Zamani nor Ms. Samali has been appointed to any board committee, and there is no such committee to which Mr. Zamani or Ms. Samali is currently expected to be appointed.

The resignations of Messrs. Handler, Melicharek, and Grosse were not the result, in whole or in part, of any disagreement with the Company or its management relating to the respective operations, policies or practices of the Company.

Appointment of Mr. Zamani

Mr. Zamani, age 53, is the founder, chairman and CEO of One Planet Group LLC, a closely held private equity firm focused on operating technology and media companies, an early-stage investor and a startup business incubator, established in 2015. In 1994, Mr. Zamani co-founded Autoweb.com. Ultimately becoming a catalyst in the way people bought and sold vehicles, Autoweb made its stock market debut on March 23, 1999. In 2001, Mr. Zamani launched Reply.com, a performance-based marketing company which in 2015 was relaunched as Buyerlink and is now owned by One Planet Group. In 2020, Mr. Zamani was honored as a Best CEO for Diversity by Comparably. He received the University of California, Davis Award of Distinction in 2018, and was granted the Tahirih Justice Center's Hope Award in 2016. Mr. Zamani graduated from UC Davis with a degree in environmental toxicology in 1994.

Except for the Investment Agreement, there are no arrangements or understandings between Mr. Zamani and any other person pursuant to which Mr. Zamani was selected as a director. In addition, there are no transactions in which Mr. Zamani has an interest that would require disclosure under Item 404(a) of Regulation S-K.

In connection with the appointment of Mr. Zamani as the Company's Chief Executive Officer, Inspirato LLC, the Company's operating subsidiary, and Mr. Zamani entered into an Executive Employment Agreement, effective as of August 13, 2024 (the "Zamani Employment Agreement"). Pursuant to the Zamani Employment Agreement, Mr. Zamani will receive an annual base salary and certain travel benefits with the Company and be eligible to participate in employee benefit or group insurance plans maintained from time to time by the Company. Mr. Zamani's initial base annual salary is \$1.00, and he will not be eligible for a cash annual performance bonus related to any period commencing prior to August 13, 2025. Additionally, the Zamani Employment Agreement provides for the following:

- an initial one-time equity grant consisting of 500,000 restricted stock units ("**RSUs**"), with 25% of the RSUs subject to the award vest on the oneyear anniversary of the date of grant and the remaining 75% of the RSUs subject to the award vest in quarterly installments over the subsequent three years (the "**One-Time Equity Grant**");
- a performance-based equity award of 500,000 RSUs, which will vest in full on the trading day after the Class A common stock achieves a closing price of \$15.00 per share or more over a period of at least 30 consecutive trading days during the performance period of August 14, 2024 through August 13, 2025 (the "**Performance-Based Equity Grant**"); and
- eligibility for ongoing annual equity awards under the Company's Equity Incentive Plan commensurate with his position and in accordance with the program applicable to other similarly situated executive officers for periods after August 13, 2025.

If Mr. Zamani's employment is terminated by the Company without Cause (as defined in the Zamani Employment Agreement) or by Mr. Zamani for Good Reason (as defined in the Zamani Employment Agreement) and Mr. Zamani has completed at least 180 days as the Chief Executive Officer under the Zamani Employment Agreement, in each case regardless of whether Mr. Zamani continues in service as a member of the Board, then Mr. Zamani will also become eligible to receive the following benefits:

- a lump-sum cash severance payment equal to \$1,100,000; and
- immediate accelerated vesting of all of Mr. Zamani's unvested equity (including any unvested portion of the One-Time Equity Grant and the Performance-Based Equity Grant), *provided that* the Performance Based Equity Grant will only vest if the performance-based targets have been achieved in the aforementioned performance period.

The foregoing description of the Zamani Employment Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Zamani Employment Agreement, a copy of which is filed as Exhibit 10.3 to this Current Report on Form 8-K and incorporated herein by reference.

In accordance with the Company's customary practice, the Company will also enter into its standard form of indemnification agreement with Mr. Zamani, which agreement is filed as Exhibit 10.1 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2023.

Appointment of Ms. Samali

Except for the Investment Agreement, there are no arrangements or understandings between Ms. Samali and any other person pursuant to which she was selected as a director. In addition, there are no transactions in which Ms. Samali has an interest that would require disclosure under Item 404(a) of Regulation S-K.

In accordance with the Company's customary practice, the Company will also enter into its standard form of indemnification agreement with Ms. Samali, which agreement is filed as Exhibit 10.1 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2023.

Handler Separation Agreement

On August 13, 2024, Inspirato LLC and Mr. Handler entered into a Separation and Release Agreement, effective as of August 13, 2024 (the "Handler Separation Agreement"). Pursuant to the Handler Separation Agreement, Mr. Handler will be entitled to (i) severance payments in an aggregate gross amount of \$216,000 to be paid in 36 equal bimonthly installments beginning on January 15, 2025; (ii) accelerated vesting of all Mr. Handler's unvested equity awards; and (iii) copayment of Mr. Handler's COBRA premiums for a period of up to 18 months. In consideration for such benefits, Mr. Handler agreed to a general release of claims in favor of the Company, and to customary confidentiality and cooperation covenants.

The foregoing summary of the Handler Separation Agreement is qualified in its entirety by reference to the full text of the Handler Separation Agreement, a copy of which is filed as Exhibit 10.4 to this Current Report on Form 8-K and is incorporated herein by reference.

Grosse Separation Agreement

On August 13, 2024, Inspirato LLC and Mr. Grosse entered into a Separation and Release Agreement, effective as of August 13, 2024 (the "Grosse Separation Agreement"). Pursuant to the Grosse Separation Agreement, Mr. Grosse will be entitled to (i) severance payments in an aggregate gross amount of \$555,000, representing 12 months of Mr. Grosse's annual base salary to be paid in 36 bimonthly installments in accordance to the payment schedule provided in the Grosse Separation Agreement; (ii) a one-time grant of 166,667 RSUs, which will become fully vested on the Company's quarterly vesting date; and (iii) copayment of Mr. Grosse's COBRA premiums for a period of up to 18 months. In consideration for such benefits, Mr. Grosse agreed to a general release of claims in favor of the Company, and to customary confidentiality and cooperation covenants.

The foregoing summary of the Grosse Separation Agreement is qualified in its entirety by reference to the full text of the Grosse Separation Agreement, a copy of which is filed as Exhibit 10.5 to this Current Report on Form 8-K and is incorporated herein by reference.

Cautionary Note Regarding Forward-Looking Statements

This Current Report on Form 8-K contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended. Forward-looking statements generally relate to future events or the Company's future financial or operating performance. In some cases, you can identify forward-looking statements because they contain words such as "believe," "may," "will," "estimate," "potential," "continue," "anticipate," "intend," "expect," "could," "would," "project," "forecast," "plan," "intend," "target," or the negative of these words or other similar expressions that concern the Company's expectations, strategy, priorities, plans, or intentions. Forward-looking statements in this Current Report on Form 8-K include, but are not limited to, the Company's ability to consummate the Transactions and satisfy applicable closing conditions, including the receipt of its stockholders' approval of the Nasdaq Proposal. The Company's expectations and beliefs regarding these matters may not materialize, and actual results in future periods are subject to risks and uncertainties, including changes in the Company's stockholders not approving the Transactions, the occurrence of any event, change or other circumstances that could result in the Investment Agreement, or at all, and uncertainties as to the timing of the consummation of the Transactions; the ability of each party to consummate the Transactions; and other risks detailed in the Company's filings with the SEC, including the Company's Annual Report on Form 10-K filed with the SEC on March 12, 2024. All information provided in this Current Report on Form 8-K is as of the date hereof, and the Company's assessment as of any date subsequent to the date of this Current Report on Form 8-K.



Additional Information and Where to Find It

The Company, its directors and certain executive officers are participants in the solicitation of proxies from stockholders in connection with the Special Meeting to approve the Nasdaq Proposal. The Company plans to file a proxy statement (the "**Special Meeting Proxy Statement**") with the SEC in connection with the solicitation of proxies for the Special Meeting. Additional information regarding such participants, including their direct or indirect interests, by security holdings or otherwise, will be included in the Special Meeting Proxy Statement and other relevant documents to be filed with the SEC in connection with the Special Meeting. Information relating to the foregoing can also be found in the Company's proxy statement for its 2024 annual meeting of stockholders (the "**2024 Proxy Statement**"). To the extent that such participants' holdings of the Company's securities have changed since the amounts printed in the 2024 Proxy Statement, such changes have been or will be reflected on Statements of Change in Ownership on Form 4 filed with the SEC.

Promptly after filing the definitive Special Meeting Proxy Statement with the SEC, the Company will mail the definitive Special Meeting Proxy Statement and related proxy card to each stockholder entitled to vote at the Special Meeting. STOCKHOLDERS ARE URGED TO READ THE SPECIAL MEETING PROXY STATEMENT (INCLUDING ANY AMENDMENTS OR SUPPLEMENTS THERETO) AND ANY OTHER RELEVANT DOCUMENTS THAT THE COMPANY WILL FILE WITH THE SEC WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Stockholders may obtain, free of charge, the preliminary and definitive versions of the Special Meeting Proxy Statement, any amendments or supplements thereto, and any other relevant documents filed by the Company with the SEC in connection with the Special Meeting at the SEC's website (http://www.sec.gov). Copies of the Company's definitive Special Meeting Proxy Statement, any amendments or supplements filed by the Company with the SEC in connection with the Special Meeting at the SEC's website (http://www.sec.gov). Copies of the Company's definitive Special Meeting Proxy Statement, any amendments or supplements filed by the Company with the SEC in connection with the Special Meeting at the SEC's website (http://www.sec.gov). Copies of the Company's definitive Special Meeting Proxy Statement, any amendments or supplements thereto, and any other relevant documents filed by the Company with the SEC in connection with the Special Meeting will also be available, free of charge, at the Company's investor relations website (http://investor.inspirato.com/) or by writing to the Company at Inspirato Incorporated, 1544 Wazee Street, Denver, Colorado 80202, Attention: Investor Relations.

Item 9.01 Financial Statements and Exhibits.

(d) <u>Exhibits</u>.

Exhibit No.	Description
<u>10.1</u>	Investment Agreement, dated as of August 12, 2024, between Inspirato Incorporated and One Planet Group LLC
<u>10.2</u>	Form of Voting Agreement
<u>10.3+</u>	Executive Employment Agreement, dated as of August 13, 2024, between Inspirato LLC and Payam Zamani
10.4+	Separation and Release Agreement, dated as of August 13, 2024, between Inspirato LLC and Brad Handler
<u>10.5+</u>	Separation and Release Agreement, dated as of August 13, 2024, between Inspirato LLC and Eric Grosse
104	Cover Page Interactive Data File (formatted in Inline XBRL and included as Exhibit 101).

Indicates a management contract or any compensatory plan, contract or arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: August 15, 2024

INSPIRATO INCORPORATED

By: /s/ Robert Kaiden

Name: Robert Kaiden Title: Chief Financial Officer

INVESTMENT AGREEMENT

This INVESTMENT AGREEMENT (this "<u>Agreement</u>"), dated as of August 12, 2024 is by and between Inspirato Incorporated, a Delaware corporation (the "<u>Company</u>"), and One Planet Group LLC, a Delaware limited liability company (the "<u>Purchaser</u>"). Capitalized terms not otherwise defined where used shall have the meanings ascribed thereto in Section 1.01.

WHEREAS, the Purchaser desires to purchase from the Company, and the Company desires to issue and sell to the Purchaser, the Firm Securities, which shall be issued in accordance with the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Company desires to grant to the Purchaser an option to acquire the Optional Securities upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, prior to the execution hereof, the Board of Directors has approved and authorized the execution and delivery of this Agreement and the other Transaction Agreements and the consummation of the transactions contemplated hereby and thereby; and

WHEREAS, substantially concurrently with the execution of this Agreement, certain stockholders of the Company have entered into voting agreements (each, a "<u>Voting Agreement</u>") with the Company and the Purchaser, pursuant to which, among other things, such stockholders hold Company Common Stock representing approximately 49% of the voting power of the Company Common Stock as of the date hereof and have agreed to vote all of their shares of Company Common Stock in favor of the Nasdaq Proposal.

NOW, THEREFORE, in consideration of the premises and the representations, warranties and agreements herein contained and intending to be legally bound hereby, the parties hereby agree as follows:

ARTICLE I DEFINITIONS

1.01 <u>Definitions</u>. As used in this Agreement, the following terms shall have the meanings set forth below:

"<u>Affiliate</u>" shall mean, with respect to any Person, any other Person which directly or indirectly controls or is controlled by or is under common control with such Person. For the avoidance of doubt, with respect to the Purchaser, (i) the Company (or any of its Affiliates) shall not be considered an Affiliate of the Purchaser or any of the Purchaser's Affiliates and (ii) the term Affiliates shall include the Persons to whom the Subject Securities are transferred on or about the date of the Tranche 1 Closing. As used in this definition, "control" (including its correlative meanings, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise).

"<u>Alternative Transaction Proposal</u>" shall mean any indication of interest, inquiry, proposal, agreement or offer, whether or not in writing, from any Person (other than the Purchaser or its Affiliates) or "group," within the meaning of Section 13(d) of the Exchange Act, relating to any transaction or series of transactions that (i) would, or would reasonably be expected to, prevent, materially impair or materially delay the consummation of the Transactions or (ii) would, or would reasonably be expected to, cause a condition to the Tranche 1 Closing or the Tranche 2 Closing set forth herein to not be satisfied prior to the applicable Termination Date.

"<u>Available</u>" means, with respect to a Registration Statement, that such Registration Statement is effective and there is no stop order with respect thereto and such Registration Statement does not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading such that such Registration Statement will be available for the resale of Registrable Securities.

"Beneficially Own," or "Beneficially Owned" shall have the meaning set forth in Rule 13d-3 of the rules and regulations promulgated under the Exchange Act.

"<u>Blackout Period</u>" means in the event that the Company determines in good faith that any registration or sale pursuant to any registration statement would reasonably be expected to materially adversely affect or materially interfere with any bona fide financing of the Company or any material transaction under consideration by the Company or would require disclosure of information that has not been, and is not otherwise required to be, disclosed to the public, the premature disclosure of which the Board of Directors determines in good faith would reasonably be expected to materially adversely affect or materially interfere with such bona fide financing of the Company or material transaction under consideration by the Company or materially adversely affect the Company, a period of up to sixty (60) days; provided that a Blackout Period may not be called by the Company more than twice in any period of twelve (12) consecutive months and may not be called by the Company in consecutive fiscal quarters.

"Board of Directors" shall mean the board of directors of the Company.

"Business Day" shall mean any day, other than a Saturday, Sunday or a day on which banking institutions in the City of New York, New York are authorized or obligated by law or executive order to remain closed.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"<u>Company Board Recommendation</u>" shall mean the unanimous recommendation of the Board of Directors to the stockholders of the Company to approve the Nasdaq Proposal.

"Company Class A Common Stock" shall mean the Class A common stock,

\$0.0001 par value per share, of the Company.

"Company Class B Common Stock" shall mean the Class B non-voting common stock, \$0.0001 par value per share, of the Company.

"Company Class V Common Stock" shall mean the Class V common stock, \$0.0001 par value per share, of the Company.

"<u>Company Common Stock</u>" shall mean the Company Class A Common Stock, the Company Class B Common Stock and the Company Class V Common Stock.

"<u>Company Covered Person</u>" means, with respect to the Company as an "issuer" for purposes of Rule 506 of the Securities Act, any Person listed in the first paragraph of Rule 506(d)(1) of the Securities Act.

"Company Preferred Stock" shall mean the preferred stock, \$0.0001 par value per share, of the Company.

"<u>Company Stockholder Approval</u>" means the approval of the Nasdaq Proposal at the Company Special Meeting in accordance with the DGCL, the organizational documents of the Company and the rules and regulations of Nasdaq.

"DGCL" means the General Corporation Law of the State of Delaware.

"Exchange Act" shall mean the U.S. Securities Exchange Act of 1934, as amended.

"GAAP" shall mean U.S. generally accepted accounting principles. "Governmental Entity" shall mean any court, administrative agency or commission or other governmental authority or instrumentality, whether federal, state, local or foreign, and any applicable industry self-regulatory organization.

"Material Adverse Effect" shall mean any events, changes, developments, facts, occurrences or circumstances that, individually or in the aggregate (x) have had, or would reasonably be expected to have, a material adverse effect on the business, financial condition or results of operations of the Company or (y) prevent, materially impair or materially delay or would reasonably be expected to prevent, materially impair or materially delay the ability of the Company to consummate the Transactions, other than, solely in the case of clause (x), any event, change, development, fact, occurrence or circumstance resulting from or arising out of the following: (a) events, changes, developments, facts, occurrences or circumstances after the date hereof generally affecting the economy, the financial or securities markets, or political, legislative or regulatory conditions, in each case in the United States or elsewhere in the world, (b) events, changes, developments, facts, occurrences or circumstances after the date hereof in the industries in which the Company conducts its business, (c) any adoption, repeal or modification after the date hereof of any rule, regulation, ordinance, Order, protocol or any other law of or by any national, regional, state or local Governmental Entity, or market administrator, (d) any changes after the date hereof in GAAP or accounting standards or interpretations thereof, (e) epidemics, pandemics, earthquakes, any weather-related or other force majeure event or natural disasters or outbreak or escalation of hostilities or acts of war or terrorism, (f) the announcement or entry into this Agreement (it being understood and agreed that this clause (f) will not apply with respect to any representation or warranty the purpose of which is to address the consequences of the announcement or entry into this Agreement), (g) any taking of any action at the express written request of Purchaser, (h) any failure by the Company to meet any financial projections or forecasts or estimates of revenues, earnings or other financial metrics for any period (provided that the exception in this clause (h) shall not prevent or otherwise affect a determination that any event, change, effect, development, fact, occurrence or circumstance underlying such failure has resulted in a Material Adverse Effect so long as it is not otherwise excluded by this definition) or (i) any changes in the share price or trading volume of the Company Class A Common Stock (provided that the exception in this clause (i) shall not prevent or otherwise affect a determination that any event, change, effect, development, fact, occurrence or circumstance underlying such change has resulted in a Material Adverse Effect so long as it is not otherwise excluded by this definition); except, in each case with respect to subclauses (a) through (e), to the extent that such event, change or development materially disproportionately affects the Company relative to other similarly situated companies in the industries in which the Company operates.



"Nasdaq" shall mean the Nasdaq Stock Market LLC.

"<u>Nasdaq Proposal</u>" shall mean the proposal to authorize the issuance of the Tranche 2 Shares, the Optional Securities and the Company Common Stock upon the exercise of the Warrant, to the extent such approval is required under the rules of the Nasdaq, including Nasdaq Rule 5635(b).

"<u>Necessary Action</u>" means, with respect to a specified result, all actions (to the extent such actions are permitted by law and by the governing documents of the Company) necessary to cause such result, including (i) voting or providing a written consent or proxy with respect to the Company Common Stock, (ii) causing the adoption of shareholders' resolutions and amendments to the governing documents of the Company, (iii) causing directors (subject to any fiduciary duties that such directors may have as directors) to act in a certain manner or causing them to be removed in the event they do not act in such a manner, (iv) executing agreements and instruments, and (v) making, or causing to be made, with governmental, administrative or regulatory authorities, all filings, registrations or similar actions that are required to achieve such result.

"Order" shall mean any decision, ruling, order, writ, judgment, injunction, stipulation, decree or award entered by or with any Governmental Entity.

"Person" or "person" shall mean an individual, corporation, limited liability or unlimited liability company, association, partnership, trust, estate, joint venture, business trust or unincorporated organization, or a government or any agency or political subdivision thereof, or other entity of any kind or nature.

"Proxy Statement" shall mean the proxy statement to be provided to the Company's stockholders in connection with the Company Special Meeting.

"<u>Registrable Securities</u>" shall mean the Subject Securities; provided that any Subject Securities will cease to be Registrable Securities when (a) such Subject Securities have been sold or otherwise disposed of to a Person that is not an Affiliate of Purchaser (or the Persons to whom the Subject Securities are transferred on or about the date of the Tranche 1 Closing) or (b) such Subject Securities cease to be outstanding.

"Registration Expenses" shall mean all expenses incurred by the Company in complying with Article V, including all registration, listing and filing fees, printing expenses, fees and disbursements of counsel and independent public accountant for the Company, fees and expenses incurred by the Company in connection with complying with state securities or "blue sky" laws, fees of the Financial Industry Regulatory Authority, Inc., transfer Taxes, and fees of transfer agents and registrars, and the reasonable and out-of-pocket legal expenses of one counsel to the Purchaser in the case of an Underwritten Offering but excluding any underwriting fees, discounts and selling commissions to the extent applicable to the Registrable Securities of the Selling Holders.

"<u>Registration Statement</u>" shall mean any registration statement of the Company filed or to be filed with the SEC under the rules and regulations promulgated under the Securities Act, including the related prospectus, amendments and supplements to such registration statement, and including pre- and post-effective amendments, and all exhibits and all material incorporated by reference in such registration statement.

"<u>Representative</u>" means, with respect to any Person, such Person's Affiliates and its and their respective officers, directors, managers, partners, employees, accountants, counsel, financial advisors, consultants, temporary agency employees, independent contractors, and other advisors, agents or representatives.

"Rule 144" shall mean Rule 144 promulgated by the SEC pursuant to the Securities Act, as such rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the SEC having substantially the same effect as such rule.

"Rule 405" shall mean Rule 405 promulgated by the SEC pursuant to the Securities Act, as such rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the SEC having substantially the same effect as such rule.

"SEC" shall mean the U.S. Securities and Exchange Commission. "Securities Act" shall mean the U.S. Securities Act of 1933, as

amended.

"Solvent" means, with respect to any Person on any date of determination, that on such date (a) the fair value (to be calculated as the amount at which the assets (both tangible and intangible), in their entirety, of such Person taken as a whole would change hands between a willing buyer and a willing seller, within a commercially reasonable period of time, each having reasonable knowledge of the relevant facts, with neither being under any compulsion to act) of the assets of such Person taken as a whole exceeds the total amount of liabilities (it being understood that the term "liabilities," for the purposes of this definition, will be limited to the recorded liabilities (including contingent liabilities that would be recorded in accordance with GAAP) of such Person taken as a whole, on such date of determination, determined in accordance with GAAP consistently applied), (b) the present fair salable value (defined as the amount that could be obtained by an independent willing seller from an independent willing buyer if the assets of such Person taken as a whole are sold with reasonable promptness in an arm's-length transaction under present conditions for the sale of comparable business enterprises insofar as such conditions can be reasonably evaluated) of the assets of such person taken as a whole exceeds their liabilities, (c) such Person, taken as whole, does not have unreasonably small capital (defined as sufficient capital to reasonably ensure that such Person will continue to be a going concern for the period from the date of determination through the one-year anniversary of the Tranche 2 Closing set forth herein, based on the needs and anticipated needs for capital of the business conducted or anticipated to be conducted by such Person reflected in such Person's projected financial statements and in light of its anticipated credit capacity), and (d) such Person and its Subsidiaries, taken as a whole, will be able to pay its liabilities as they mature.

"Subject Securities" shall mean (i) the Tranche 1 Shares and the Tranche 2 Shares issued pursuant to this Agreement (or transferred to the Persons on or about the date of the Tranche 1 Closing) and any shares of Company Common Stock issued upon the Option Closing Date; (ii) the shares of Company Common Stock issuable or issued upon the exercise of the Warrant; and (iii) any securities issued as or pursuant to (or issuable upon the conversion, exercise or exchange of any warrant, right or other security that is issued as or pursuant to) a dividend, stock split, combination or any reclassification, recapitalization, merger, consolidation, exchange or any other distribution or reorganization with respect to, or in exchange for, or in replacement of, the securities referenced in clause (i) or (ii) above or this clause (iii).

"Subsidiary" shall mean, with respect to any Person, any other Person of which 50% or more of the shares of the voting securities or other voting interests are owned or controlled, or the ability to select or elect 50% or more of the directors or similar managers is held, directly or indirectly, by such first Person or one or more of its Subsidiaries, or by such first Person, or by such first Person and one or more of its Subsidiaries.

"<u>Tax</u>" or "<u>Taxes</u>" shall mean all federal, state, local, and foreign income, excise, gross receipts, gross income, ad valorem, profits, gains, property, capital, sales, transfer, use, license, payroll, employment, social security, severance, unemployment, unclaimed property, withholding, duties, windfall profits, intangibles, franchise, backup withholding, value-added, alternative or add-on minimum, estimated and other taxes, charges, levies or like assessments imposed by a Governmental Entity, together with all interest, penalties and additions to tax imposed with respect thereto.

"Tax Return" shall mean a report, return, information statement, declaration, claim for refund, election, statement or other document supplied or required to be supplied to a Governmental Entity with respect to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"<u>Third Party</u>" shall mean a Person other than Purchaser or any of its Affiliates. "<u>Underwritten Offering</u>" shall mean a sale of Registrable Securities to an underwriter or underwriters for reoffering to the public, including in a block trade offered and sold through an underwriters.

1.02 <u>General Interpretive Principles</u>. Whenever used in this Agreement, except as otherwise expressly provided or unless the context otherwise requires, any noun or pronoun shall be deemed to include the plural as well as the singular and to cover all genders. The name assigned to this Agreement and the section captions used herein are for convenience of reference only and shall not be construed to affect the meaning, construction or effect hereof. Whenever the words "include," "includes," or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The word "will" shall be construed to have the same meaning as the word "shall." The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if." The word "or" shall not be exclusive. Unless otherwise specified, the terms "hereto," "hereof," "herein" and similar terms refer to this Agreement as a whole (including the exhibits, schedules and disclosure statements hereto), references to "the date hereof" refer to the date of this Agreement and references herein to Articles or Sections refer to Articles or Sections of this Agreement. References from or through any date mean, unless otherwise specified, from and including such date or through and including such date, respectively. References to any period of days will be deemed to be the relevant number of calendar days unless otherwise specified and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the party drafting or causing any instrument to be drafted.

ARTICLE II SALE AND PURCHASE OF THE SECURITIES

2.01 <u>Sale and Purchase of the Tranche 1 Securities</u>. Subject to the terms and conditions of this Agreement, at the Tranche 1 Closing, the Company shall issue and sell to the Purchaser, and the Purchaser shall purchase and acquire from the Company, 1,335,271 shares of Class A Common Stock (the "<u>Tranche 1 Shares</u>" or the "<u>Tranche 1 Securities</u>") for an aggregate purchase price of four million five hundred seventy-nine thousand nine hundred seventy-nine dollars and fifty-three cents (\$4,579,979.53) (such price, the "<u>Tranche 1 Purchase Price</u>," and such transaction, the "<u>Tranche 1 Purchase</u>").

2.02 <u>Sale and Purchase of the Tranche 2 Securities</u>. Subject to the terms and conditions of this Agreement, at the Tranche 2 Closing, the Company shall issue and sell to the Purchaser, and the Purchaser shall purchase and acquire from the Company, 1,580,180 shares of Class A Common Stock (the "<u>Tranche 2 Shares</u>") and the Warrant, in the form attached hereto as <u>Exhibit A</u> (the "<u>Warrant</u>" and, together with the Tranche 2 Shares, the "<u>Tranche 2 Shares</u>"), for an aggregate purchase price of five million four hundred twenty thousand twenty dollars and forty- seven cents (\$5,420,020.47) (such price, the "<u>Tranche 2 Purchase Price</u>," and such transaction, the "<u>Tranche 2 Purchase</u>"). The Tranche 1 Securities and the Tranche 2 Securities are referred to collectively as the "<u>Firm Securities</u>", and the Tranche 1 Purchase Price and the Tranche 2 Purchase Price are referred to collectively as the "<u>Purchase Price</u>".

2.03 Closing.

(a) The closing of the Tranche 1 Purchase (the "<u>Tranche 1 Closing</u>") shall take place on August 13, 2024 or as soon as practicable thereafter following the satisfaction or waiver (to the extent permitted by applicable law) of all of the conditions set forth in this Article II (other than such conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions at or prior to the Closing). The closing of the Tranche 2 Purchase (the "<u>Tranche 2 Closing</u>", with each of the Tranche 1 Closing and the Tranche 2 Closing being referred to as a "<u>Closing</u>") shall take place on September 13, 2024, or as soon as practicable thereafter following the satisfaction or waiver (to the extent permitted by applicable law) of all of the conditions set forth in this Article II (other than such conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions at or prior to the Closing) (the date on which each Closing actually occurs, the "<u>Tranche 1 Closing Date</u>" or the "<u>Tranche 2 Closing Date</u>", as applicable, with each being referred to as a "<u>Closing Date</u>"), by electronic exchange of deliverables, unless another date, time or place is agreed to in writing by the parties hereto.

(b) At the Tranche 1 Closing, (i) the Company shall issue to the Purchaser the Tranche 1 Shares, (ii) the Purchaser shall cause a wire transfer to be made in immediately available funds to an account of the Company designated in writing by the Company to the Purchaser in an amount equal to the Tranche 1 Purchase Price and (iii) the Purchaser shall deliver to the Company a duly completed and executed Internal Revenue Service Form W-9 or Form W-8, as applicable.

(c) At the Tranche 2 Closing, (i) the Company shall issue to the Purchaser the Tranche 2 Shares, (ii) the Company shall execute and deliver to the Purchaser the Warrant, (iii) the Company and the Purchaser shall execute one or more certificates dated as of the date of the Tranche 2 Closing that the respective conditions to such closing have been satisfied as of such date and (iv) the Purchaser shall cause a wire transfer to be made in immediately available funds to an account of the Company designated in writing by the Company to the Purchaser in an amount equal to the Tranche 2 Purchase Price.

(d) Neither party shall be obligated to effect the Tranche 1 Purchase or the Tranche 2 Purchase, as applicable, if (i) a statute, rule or regulation that prohibits such purchase shall have been enacted, issued, enforced or promulgated and remains in effect by any Governmental Entity or if there shall be an Order or injunction of a court of competent jurisdiction prohibiting or making illegal the consummation of such purchase or (ii) in the case of the Tranche 2 Purchase, the Nasdaq Proposal has not been approved.

(e) The obligations of the Purchaser to effect the Tranche 1 Purchase or the Tranche 2 Purchase, as applicable, are subject to the satisfaction or written waiver by the Purchaser of the following conditions as of the applicable Closing:

(i) (A) the representations and warranties of the Company set forth in Section 3.01(a), Section 3.01(c), Section 3.01(g), Section 3.01(h)(i), Section 3.01(t) and Section 3.01(aa) shall be true and correct in all material respects on and as of the date hereof and the relevant Closing Date, (B) the representations and warranties of the Company set forth in Section 3.01(b) shall be true and correct in all respects on and as of the date hereof (other than *de minimis* inaccuracies), (C) the representations and warranties of the Company set forth in Section 3.01(j)(ii) shall be true and correct on and as of the date hereof and the relevant Closing Date and (D) the other representations and warranties of the Company set forth in Section 3.01 shall be true and correct on and as of the date hereof and the relevant Closing Date (Without giving effect to materiality, Material Adverse Effect, or similar phrases in the representations and warranties), except where the failure of such representations and warranties referenced in this clause (C) to be so true and correct, individually or in the aggregate, has not had and would not reasonably be expected to have a Material Adverse Effect;

(ii) the Company shall have entered into Voting Agreements in respect of Company Common Stock that represent at least 35% of the voting power of the Company Common Stock outstanding as of the Tranche 1 Closing Date; and

(iii) the Company shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be performed or complied with by it on or prior to the Closing Date.

(f) The obligations of the Company to effect the each of the Tranche 1 Purchase and the Tranche 2 Purchase are subject to the satisfaction or waiver by the Company of the following conditions as of the relevant Closing:

(i) (A) the representations and warranties of the Purchaser set forth in Section 3.02(a), Section 3.02(b)(i), Section 3.02(b) (iii) and Section 3.02(e) shall be true and correct in all material respects on and as of the date hereof and the relevant Closing Date, and (B) the other representations and warranties of the Purchaser set forth in Section 3.02 shall be true and correct on and as of the date hereof and the relevant Closing Date (without giving effect to materiality or similar phrases in the representations and warranties), except where the failure of such representations and warranties referenced in this clause (B) to be so true and correct, individually or in the aggregate, would not reasonably be expected to prevent, materially impair or materially delay the ability of the Purchaser to consummate the Transactions; and

(ii) the Purchaser shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be performed or complied with by it on or prior to the relevant Closing Date.

2.04 <u>Termination of Prior to Closing</u>. The Tranche 1 Purchase and the Tranche 2 Purchase may be abandoned at any time prior to the relevant Closing (provided that if the Tranche 1 Purchase has been consummated, the following shall apply only to the Tranche 2 Purchase and the Optional Securities):

(a) upon the mutual written agreement of the Purchaser and the Company, with the approval of the Board of Directors;

(b) by either the Purchaser or the Company, if a court of competent jurisdiction or other Governmental Entity shall have issued an Order or ruling or taken any other action, and such Order or ruling or other action will have become final and non-appealable, or there will exist any statute, rule or regulation, in each case, permanently restraining, enjoining or otherwise prohibiting the consummation of the Tranche 1 Purchase or the Tranche 2 Purchase, as applicable, provided, however, that the right to terminate this Agreement in accordance with this Section 2.04(b) will not be available to any party whose failure to comply with its obligations under this Agreement has been the primary cause of such restraint or the failure to remove such restraint;

(c) upon the failure of the Tranche 1 Closing to occur on or prior to the date this is five (5) days after the date of this Agreement or, with respect to the Tranche 2 Closing, the failure of such Closing to occur on or prior to the date that is sixty (60) days after the date of this Agreement (unless otherwise mutually agreed by the Company and Purchaser in writing) (each, a "<u>Termination Date</u>"); provided, however, that the right to terminate this Agreement under this Section 2.04(c) shall not be available to any party whose failure to comply with its obligations under this Agreement has been the primary cause of the failure of the relevant Closing to occur on or before such time;

(d) by Purchaser, if, prior to the relevant Closing, there has been a breach by the Company of, or any inaccuracy in, any representation, warranty, covenant or other agreement of the Company set forth in this Agreement such that a condition set forth in Section 2.03(e) would not be then satisfied and such breach or inaccuracy has not been cured within fifteen (15) days following notice by the Purchaser thereof or such breach or inaccuracy is not reasonably capable of being cured; provided that the Purchaser will not be entitled to terminate this Agreement pursuant to this Section 2.04(d) at any time as of which the Purchaser is in breach of any representation, warranty, covenant or agreement such that a condition set forth in Section 2.03(f) would not be then satisfied, measured as of such time; or

(e) by the Company if, prior to the relevant Closing, there has been a breach by the Purchaser of, or any inaccuracy in, any representation, warranty, covenant or other agreement of the Purchaser set forth in this Agreement such that a condition set forth in Section 2.03(f) would not be then satisfied and such breach or inaccuracy has not been cured within fifteen (15) days following notice by the Company thereof or such breach or inaccuracy is not reasonably capable of being cured; provided that the Company will not be entitled to terminate this Agreement pursuant to this Section 2.04(e) at any time as of which the Company is in breach of any representation, warranty, covenant or agreement such that a condition set forth in Section 2.03(e) would not be then satisfied, measured as of such time.

Any termination of this Agreement in accordance with this Section 2.03 will be effective immediately upon the delivery of a written notice of the terminating party to Purchaser, if the Company is the terminating party, or to the Company, if the Purchaser is the terminating party. If this Agreement is terminated in accordance with this Section 2.03, this Agreement will become null and void and be of no further force or effect with respect to all of the Securities or, as applicable, the Tranche 2 Securities and the Optional Securities, and there will be no liability on the part of any party hereto (or any of their respective Representatives); provided, however, that (x) this paragraph of this Section 2.04 shall survive any such termination, (y) nothing herein will relieve any party from liability for fraud in the making of any of its representations and warranties set forth in this Agreement or willful breach of any of its covenants or agreements set forth in this Agreement prior to such termination and (z) if the termination applies only to the Tranche 2 Securities and the Optional Securities, the terms of this Agreement will remain in effect insofar as they relate to the Tranche 1 Securities.

2.05 Option to Purchase Additional Shares. In addition, the Purchaser shall have the option to purchase up to 728.863 additional shares of Class A Common Stock and warrants to purchase an additional 728.863 shares of Class A Common Stock (the securities so purchased being referred to as the "Optional Securities") for up to an aggregate purchase price of \$2,500,000 (the "Option") where the purchase price for each share of Class A Common Stock will be the same per share purchase price as in the Tranche 1 Purchase and where the Company will deliver a number of warrants equal to the number of shares of Class A Common Stock being purchased as part of the Option. Such warrants shall have the same terms as issued in the Warrant (and may be included in the Warrant) except such warrants will not contain a limitation on exercise relating to stockholder approval of the Nasdaq Proposal. The Option may be exercised by the Purchaser on a one-time basis at any time after the Tranche 2 Closing and prior to the date (the "Option Expiration Date") that is thirty (30) days after the date of the Tranche 2 Closing (it being understood that such exercise shall be at the sole discretion of the Purchaser and without regard to any condition, including, for the avoidance of doubt, any requirement for pre-clearance under the Company's insider trading policy) (such exercise date, the "Option Notice Date"), in whole or in part, upon notice by the Purchaser to the Company (the "Option Notice") setting forth the aggregate purchase price of the Optional Securities as to which the Purchaser is exercising the Option (the "Option Purchase Amount"), the corresponding number of Optional Securities to be issued and the time and date of payment and delivery for the Optional Securities. The time and date of delivery of the Optional Securities (the "Option Closing Date" and such closing, the "Option Closing") shall be determined by the Purchaser in its sole discretion but shall not be earlier than the date of the Tranche 2 Closing Date and shall not be later than five (5) Business Days following Purchaser's delivery of the Option Notice. The number of shares of Class A Common Stock being purchased by the Purchaser at the Option Closing shall be a number equal to (i) the Option Purchase Amount divided by (ii) the per share purchase price in the Tranche 1 Purchase (rounded down to the nearest whole share). At the Option Closing, (i) the Company shall issue to the Purchaser the Optional Securities, (ii) the Company and the Purchaser shall execute one or more certificates dated as of the date of the Option Closing that the respective conditions to such closing have been satisfied as of such date and (iii) the Purchaser shall cause a wire transfer to be made in immediately available funds to an account of the Company designated in writing by the Company to the Purchaser in an amount equal to the Option Purchase Amount. Neither party shall be obligated to effect the Option Closing if a statute, rule or regulation that prohibits the Option Closing shall have been enacted, issued, enforced or promulgated and remains in effect by any Governmental Entity or if there shall be an Order or injunction of a court of competent jurisdiction prohibiting or making illegal the consummation of the Option Closing. The obligations of the Purchaser to effect the Option Closing are subject to the satisfaction or written waiver by the Purchaser of the conditions as of the Option Closing set forth in Section 2.03(e)(i) and (iii) where references therein to "Closing Date" shall be deemed to be references to "Option Closing Date". The obligations of the Company to effect the Option Closing are subject to the satisfaction or waiver by the Company of the conditions set forth in 2.04(f)(i) and (ii) where references therein to "Closing Date" shall be deemed to be references to "Option Closing Date". For purposes of this Agreement, except as the context otherwise indicates, all references to Securities shall be deemed to refer to the Firm Securities with respect to the Tranche 1 Closing and the Tranche 2 Closing, and shall be deemed to refer to the Optional Securities with respect to the Option Closing. The Purchaser agrees that if the Purchaser does not exercise the Option as contemplated by this Section 2.05 by the end of the day on the Option Expiration Date, the Company, as directed by the Board, may offer the Optional Securities to other parties.

ARTICLE III REPRESENTATIONS AND WARRANTIES

3.01 <u>Representations and Warranties of the Company</u>. Except as disclosed in the Company Reports filed with or furnished to the SEC and publicly available prior to the date hereof (excluding in each case any general disclosures set forth in the risk factors or "forward-looking statements" sections of such reports, and any other general disclosures included therein to the extent they are predictive or forward-looking in nature), the Company represents and warrants to the Purchaser, as of the date hereof, as of the Closing Date and as of the Option Closing Date, as follows:

(a) <u>Existence and Power</u>. The Company is duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation and has all requisite corporate power and authority to own, operate and lease its properties and to carry on its business as it is being conducted on the date of this Agreement, and, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, has been duly qualified as a foreign corporation for the transaction of business and is in good standing under the laws of each other jurisdiction in which it owns or leases properties, or conducts any business so as to require such qualification. The Company is not in material breach of its organizational documents.

(b) Capitalization.

The authorized share capital of the Company consists of 80,000,000 shares of Company Common Stock (consisting of (i) 50,000,000 shares of Company Class A Common Stock, 25,000,000 shares of Company Class V Common Stock, and 5,000,000 shares of Company Class B Common Stock) and 5,000,000 shares of Company Preferred Stock. As of August 8, 2024, there were 3,818,727 shares of Company Class A Common Stock, 2,857,635 shares of Company Class V Common Stock, no shares of Company Class B Common Stock and no shares of Company Preferred Stock issued and outstanding. As of August 8, 2024, there were (a) no shares of Company Class A Common Stock underlying Company restricted stock awards, (b) options to purchase an aggregate of 150,940 shares of Company Common Stock at an exercise price of \$15.60 per share issued and outstanding, (c) 869,021 shares of Company Class A Common Stock underlying the Company's restricted stock unit awards, (d) 8,624,792 shares of Company Class A Common Stock underlying the Company's warrants that are listed on the Nasdaq, which warrants have an exercise price of \$230.00 per share (the "Public Warrants") and 884,733 shares of Company Class A Common Stock issuable upon conversion of the Company's 8% Senior Secured Convertible Note (the "Capital One Note"). Since August 8, 2024, (a) the Company has only issued options, shares of restricted stock or restricted stock units to acquire shares of Company Common Stock in the ordinary course of business consistent with past practice and (b) the only shares of capital stock issued by the Company were pursuant to restricted stock, outstanding options or restricted stock units to purchase shares of Company Common Stock or issuances of shares of Company Class A Common Stock in exchange for shares of Company Class V Common Stock and common units of Inspirato LLC ("OpCo") in accordance with the organizational documents of the Company and the OpCo LLCA (as defined below). All outstanding shares of Company Common Stock are duly authorized, validly issued, fully paid and nonassessable, and are not subject to and were not issued in violation of any preemptive or similar right, purchase option, call or right of first refusal or similar right. Except as set forth above, the Company does not have any issued and outstanding securities. Except as provided in this Agreement (including the Warrant and the Optional Securities) or the Eleventh Amended and Restated Limited Liability Company Agreement of OpCo, dated October 16, 2023 (the "OpCo LLCA"), the Public Warrants and the Capital One Note and except as set forth in or contemplated by this Section 3.01(b)(i), there are no existing options, warrants, calls, preemptive (or similar) rights, subscriptions or other rights, agreements or commitments obligating the Company to issue, transfer or sell, or cause to be issued, transferred or sold, any capital stock of the Company or any securities convertible into or exchangeable for such capital stock and there are no current outstanding contractual obligations of the Company to repurchase, redeem or otherwise acquire any of its shares of capital stock.



Authorization. The execution, delivery and performance of this Agreement, the Warrant and the Voting Agreements (the (c) "Transaction Agreements") and the consummation of the transactions contemplated herein and therein (collectively, the "Transactions") have been duly authorized by the Board of Directors and all other necessary corporate action on the part of the Company other than the receipt of the Company Stockholder Approval. The execution, delivery and performance of each Transaction Agreement and the consummation of the Transactions contemplated thereby has been duly authorized by the governing body of the Company and all other necessary corporate action on the part of the Company. Assuming this Agreement constitutes the valid and binding obligation of the Purchaser, this Agreement is a valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the limitation of such enforcement by (A) the effect of bankruptcy, insolvency, reorganization, receivership, conservatorship, arrangement, moratorium or other laws affecting or relating to creditors' rights generally, (B) the rules governing the availability of specific performance, injunctive relief or other equitable remedies and general principles of equity, regardless of whether considered in a proceeding in equity or at law and (C) public policy limitations on indemnification, advancement and contribution (the "Enforceability Exceptions"). Pursuant to resolutions previously provided to the Purchaser, the Board of Directors has, by resolutions unanimously adopted by the Board of Directors: (i) determined that the Tranche 1 Purchase, the Tranche 2 Purchase and the other Transactions are advisable and in the best interests of the Company and the Company's stockholders; (ii) approved and declared it advisable to enter into this Agreement and the other Transaction Agreements; (iii) directed that the Nasdaq Proposal be submitted to a vote of the Company's stockholders at the Company Special Meeting; and (iv) resolved to recommend that that the Company stockholders approve the Nasdag Proposal. The Company Board Recommendation has not been amended, rescinded or modified. Approval of the Nasdaq Proposal at the Company Special Meeting requires the affirmative vote of a majority of the voting power of the shares of Company Common Stock (not including the Class B Common Stock) present in person or represented by proxy at the Company Special Meeting and entitled to vote on the Nasdaq Proposal.

(d) <u>General Solicitation; No Integration</u>. Neither the Company nor any other Person or entity authorized by the Company to act on its behalf has engaged in a general solicitation or general advertising (within the meaning of Regulation D of the Securities Act) of investors in connection with the offer or sale of the Securities. The Company has not, directly or indirectly, sold, offered for sale, solicited offers to buy or otherwise negotiated in respect of, any security (as defined in the Securities Act) which, to its knowledge, is or will be integrated with the Securities sold pursuant to this Agreement.

(e) <u>No Registration</u>. Assuming the accuracy of the representations of the Purchaser set forth herein, no registration under the Securities Act is required for the offer and sale of the Securities by the Company to the Purchaser in the manner contemplated by this Agreement. The Securities are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any state securities laws.

(f) <u>No Disqualification Event</u>. No disqualifying event described in Rule 506(d)(1)(i-viii) of the Securities Act (a "<u>Disqualification</u> <u>Event</u>") is applicable to the Company or, to the Company's knowledge, any Company Covered Person, except for a Disqualification Event as to which Rule 506(d)(2)(i-iv) or (d)(3) of the Securities Act is applicable. The Company has complied, to the extent applicable, with any disclosure obligations under Rule 506(e) under the Securities Act.

(g) <u>Valid Issuance</u>. The Securities have been duly authorized by all necessary corporate action of the Company. When issued and sold against receipt of the consideration therefor, the Warrant will be a valid and legally binding obligation of the Company, enforceable in accordance with its terms, subject to the limitation of such enforcement by the Enforceability Exceptions. The Company has available for issuance the maximum number of shares of Company Common Stock issuable upon exercise of the Warrant. All of the shares of Company Common Stock to be issued upon exercise of the Warrant have been duly authorized and will, upon such issuance, be validly issued, fully paid and nonassessable and free of pre-emptive or similar rights.

(h) <u>Non-Contravention/No Consents</u>. The execution, delivery and performance of the Transaction Agreements, the issuance of the shares of Company Common Stock upon exercise of the Warrant in accordance with its terms and the consummation by the Company of the Transactions, does not conflict with, violate or result in a breach of any provision of, or constitute a default under, or result in the termination of or accelerate the performance required by, or result in a right of termination or acceleration under, (i) the certificate of incorporation, bylaws, limited liability company agreement, operating agreement, partnership agreement or other applicable organizational documents of the Company, (ii) any mortgage, note, indenture, deed of trust, lease, license, loan agreement or other agreement binding upon the Company or (iii) any permit, license, judgment, order, decree, ruling, injunction, statute, law, ordinance, rule or regulation applicable to the Company, other than in the cases of clauses (ii) and (iii) as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Assuming the accuracy of the representations of the Purchaser set forth herein, other than (A) requirements or regulations in connection with the issuance of shares of Company Common Stock upon the exercise of the Warrant, (B) any required filings pursuant to the Exchange Act or the rules of the SEC or Nasdaq or (C) as have been obtained prior to the date of this Agreement, no consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity is required on the part of the Company in connection with the execution, delivery and performance by the Company of the Transaction Documents and the consummation by the Company of the Transactions, except for any consent, approval, order, authorization, registration, declaration, filing, exemption or review the failure of which to be obtained or made, individually or in the aggregate, would not reasonably be expected

(i) <u>Reports; Financial Statements</u>.

(i) The Company has filed or furnished, as applicable, (A) its annual report on Form 10-K for the fiscal year ended December 31, 2023, (B) its quarterly report on Form 10-Q for its fiscal quarter ended March 31, 2024, (C) its proxy statement relating to the annual meeting of the stockholders of the Company held in 2024 and (D) all other forms, reports, schedules and other statements required to be filed or furnished by it with the SEC under the Exchange Act or the Securities Act since January 1, 2023 (collectively, the "<u>Company Reports</u>"). As of its respective date, and, if amended, as of the date of the last such amendment, each Company Report complied in all material respects as to form with the applicable requirements of the Securities Act and the Exchange Act, and any rules and regulations promulgated thereunder applicable to such Company Report. As of its respective date, and, if amended, as of the date of the last such amendment, no Company Report contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading.

(ii) Each of the consolidated balance sheets, and the related consolidated statements of operations and comprehensive loss, consolidated statements of equity, and consolidated statements of cash flows included in the Company Reports filed with the SEC under the Exchange Act (A) have been prepared from, and are in accordance with, the books and records of the Company, (B) fairly present in all material respects the consolidated financial position of the Company and its consolidated Subsidiaries as of the respective dates shown, (C) have been prepared in accordance with GAAP consistently applied during the periods involved, except as otherwise set forth therein or in the notes thereto and (D) otherwise comply with the requirements of the SEC.

(iii) The Company (i) makes and keeps accurate books and records in all material respects and (ii) maintains systems of internal accounting controls sufficient to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP, including, but not limited to internal accounting controls sufficient to provide reasonable assurance that (A) transactions are executed in accordance with management's general or specific authorization, (B) transactions are recorded as necessary to permit preparation of its financial statements in accordance with GAAP and to maintain accountability for its assets, (C) access to its assets is permitted only in accordance with management's general or specific authorization, (D) the recorded accountability for its assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences. Since the end of the Company's most recent audited fiscal year, there has been (i) no material weakness in the Company's internal control over financial reporting (whether or not remediated) and (ii) no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

(j) <u>Absence of Certain Changes</u>. Since December 31, 2023, (i) the Company has conducted its businesses in all material respects in the ordinary course of business, and (ii) no events, changes, developments, facts, occurrences or circumstances have occurred that, individually or in the aggregate, have had or would reasonably be expected to have a Material Adverse Effect.

(k) <u>No Undisclosed Liabilities, Etc.</u> There are no liabilities of the Company that would be required by GAAP to be reflected on the face of the balance sheet, except (i) liabilities reflected or reserved against in the financial statements or disclosed in the notes thereto contained in the Company Reports, (ii) liabilities incurred since June 30, 2024 in the ordinary course of business, (iii) liabilities incurred in connection with the Transactions and (iv) liabilities that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(1) <u>Compliance with Applicable Law</u>. The Company has complied in all respects with, and is not in default or violation in any respect of, any law, statute, order, rule, regulation, policy or guideline of any federal, state or local governmental authority, binding industry standard or posted privacy policy applicable to such entity, other than such non- compliance, defaults or violations that, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect.

(m) <u>Permits</u>. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, the Company possesses all certificates, authorizations and permits issued by the appropriate federal, state or foreign regulatory authorities necessary to conduct its businesses, and the Company has not received any notice of proceedings relating to the revocation or modification of any such certificate, authorization or permit.

(n) <u>Legal Proceedings</u>. The Company is not a party to any, and there are no pending, or to the knowledge of the Company, threatened, legal, administrative, arbitral or other proceedings, claims, actions or governmental investigations of any nature against any the Company (i) that, individually or in the aggregate, have had or would reasonably be expected to have a Material Adverse Effect or (ii) that challenge the validity of or seek to prevent the Transactions. The Company is not subject to any Order, judgment or decree of a Governmental Entity that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have a Material Adverse Effect. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have a Material Adverse Effect, there is no investigation or review pending or, to the knowledge of the Company, threatened by any Governmental Entity with respect to the Company.

(o) Anti-Corruption, Anti-Money Laundering, and Economic Sanctions Compliance.

(i) The Company, each of its officers and directors and, to the Company's knowledge, their respective employees and agents acting on behalf of the Company are, and for the past five (5) years have been, in material compliance with: (A) anti-bribery and anti-corruption laws applicable to the Company or any of its Subsidiaries, including the Foreign Corrupt Practices Act of 1977 and the UK Bribery Act of 2010 (collectively, "<u>Anti-Corruption Laws</u>"); (B) the anti-money laundering statutes of all relevant jurisdictions, the rules and regulations promulgated thereunder and any other rules or regulations relating to anti-money laundering issued, administered or enforced by any relevant Governmental Entity (collectively, the "<u>Anti-Money Laundering Laws</u>"); and (C) economic sanctions administered or enforced by the Office of Foreign Assets Control and the U.S. Department of State, the United Nations Security Council, the European Union, Her Majesty's Treasury, or any other relevant sanctions authority (collectively, "<u>Sanctions</u>"). In the past five (5) years, neither (x) the Company or any of its officers or directors nor (y) to the Company's knowledge, any of the Company's employees or agents acting on behalf of the Company has made any offer or promise of, or has otherwise authorized, any direct or indirect payment or benefit to any foreign or domestic government official in violation of any Anti-Corruption Laws, Anti-Money Laundering Laws and applicable Sanctions.



(ii) In the past five (5) years, the Company has not engaged in any transactions or business dealings with any Person that is the subject or target of Sanctions, or in or with any country or territory that is the subject or target of comprehensive Sanctions in each case at the time of such transaction or business dealing (at the time of this Agreement, Crimea, Donetsk People's Republic and Luhansk People's Republic regions of Ukraine, Cuba, Iran, North Korea, Russia, Belarus and Syria).

(iii) To the knowledge of Company, no Governmental Entity is investigating or, in the past five (5) years, conducted, initiated or threatened any investigation of or action against the Company in connection with an alleged or potential violation of any applicable Anti-Corruption Laws, Anti-Money Laundering Laws or Sanctions.

(p) <u>Property; Insurance</u>. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) the Company has good and valid title in fee simple to all real property and good and valid title to all personal property owned by it which is used in the business of the Company, in each case free and clear of all liens, encumbrances and defects; and (ii) any real property and buildings held under lease by the Company is held by it under valid, subsisting and enforceable leases, subject to the Enforceability Exceptions. The Company is insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as are prudent and customary in the businesses in which it is engaged.

(q) <u>Environmental Laws</u>. The Company (i) is in compliance with any and all applicable foreign, federal, state and local laws and regulations relating to the protection of human health and safety, the environment or hazardous or toxic substances or wastes, pollutants or contaminants ("<u>Environmental Laws</u>"), (ii) has received all permits, licenses or other approvals required of them under applicable Environmental Laws to conduct its businesses and (iii) is in compliance with all terms and conditions of any such permit, license or approval, except where such noncompliance with Environmental Laws, failure to receive required permits, licenses or other approvals or failure to comply with the terms and conditions of such permits, licenses or approvals would not, individually or in the aggregate, have or would reasonably be expected to have a Material Adverse Effect.

(r) <u>Intellectual Property</u>. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) the Company exclusively owns its proprietary patents, trademarks, service marks, trade names, domain names and other source indicators, copyrights, know-how, trade secrets and other intellectual property rights (collectively, "<u>Intellectual Property</u>") and has the right to use all other Intellectual Property used in the conduct of its businesses as currently conducted; (ii) the conduct of the Company's businesses does not infringe or violate any Intellectual Property of any Person and no Person is infringing or violating any Intellectual Property owned by the Company; (iii) the Company has not distributed, conveyed or made available to third parties any software that is subject to any open source or similar license that requires the licensing or availability of material proprietary source code in such circumstances and (iv) no Person (other than employees or service providers working on behalf of the Company and subject to reasonable confidentiality arrangements) has the current or contingent right to access or possess any of their proprietary source code.

(s) <u>Data Security; Privacy</u>. The software, systems, networks, databases and other information technology assets ("<u>IT Assets</u>") used by the Company are adequate for the operation of their businesses as currently conducted and are, to the knowledge of the Company, free of defects, malware, viruses or other corruptants. The Company takes, and has taken, commercially reasonable actions (including implementing organizational, physical, administrative and technical measures) to protect and maintain the integrity, security, operation and redundancy of the IT Assets used by or on behalf of the Company, whether proprietary or those of third parties (including all data, including personal and confidential data, stored thereon and processed thereby), and there have been no violations, outages, breaches, interruptions, or unauthorized accesses to same, other than those that would not reasonably be expected to have a Material Adverse Effect. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) the Company has complied and is in compliance with all internal and external privacy policies, contractual obligations, industry standards, applicable laws, statutes, judgments, Orders, rules and regulations of any Governmental Entity and any other legal obligations, in each case, relating to the collection, use, transfer, import, export, storage, protection, disposal and disclosure by the Company of personal, personally identifiable, household, sensitive, confidential or regulated data ("<u>Data Security Obligations</u>"); (ii) the Company has not received any notification of or complaint regarding and is unaware of any other facts that, individually or in the aggregate, would reasonably indicate non-compliance with any Data Security Obligation; and (iii) there is no action, suit or proceeding by or before any Governmental Entity pending or threatened against the Company alleging non-compliance with any Data Security Obligation.

(t) <u>Investment Company Act; Margin Rules</u>. The Company is not, and immediately after the Closing will not be, required to register as an "investment company" within the meaning of the Investment Company Act of 1940, as amended. The Company is not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying Margin Stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System of the United States (the "<u>FRB</u>") as from time to time in effect), or extending credit for the purpose of purchasing or carrying Margin Stock and no proceeds of the sale of the Securities will be used to purchase or carry any Margin Stock or to extend credit to others for the purpose of purchasing or carrying any Margin Stock. Neither the purchase or issuance of the Securities hereunder nor the use of proceeds thereof will violate any regulations of the FRB, including the provisions of Regulations T, U or X of the FRB.

(u) <u>Taxes and Tax Returns</u>. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have a Material Adverse Effect:

(i) the Company has timely filed (taking into account all applicable extensions) all Tax Returns required to be filed by it, and all such Tax Returns are true, correct and complete in all respects, and the Company has duly and timely paid (or has had duly and timely paid on its behalf) to the appropriate Governmental Entity all Taxes that are required to be paid by it;

(ii) the Company has deducted, withheld and collected, and timely remitted to the appropriate Governmental Entity, all Taxes required to have been withheld, deducted or collected and remitted by it in connection with any amounts paid or owing to or from any employee, creditor, customer, shareholder, independent contractor or other Third Party;

(iii) there are no audits, disputes, claims, investigations, or other proceedings pending or asserted in writing in respect of Taxes of the Company;

(iv) the Company (A) is not nor has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group the common parent of which was the Company) or (B) does not have any liability for the Taxes of any person under Treasury Regulations Section 1.1502-6 (or any similar provision of any law), as a transferee or successor, by contract or otherwise;

(v) the Company has not been, within the past two (2) years, a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intending to qualify for tax-free treatment under Section 355 of the Code; and

4(b).

(vi) the Company has not participated in a "listed transaction" within the meaning of Treasury Regulations Section 1.6011-

(v) <u>Brokers and Finders</u>. The Company has not retained, utilized or been represented by, or otherwise become obligated to, any broker, placement agent, financial advisor or finder in connection with the transactions contemplated by this Agreement whose fees the Purchaser would be required to pay.

(w) <u>Proxy Statement</u>. The Proxy Statement on the date filed, mailed, distributed or disseminated, as applicable, to the Company's stockholders and at the time of the Company Special Meeting, will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. The Proxy Statement, including any amendments thereof and supplements thereto, will comply in all material respects with the requirements of applicable law, except that the Company makes no representation or warranty with respect to statements made in the Proxy Statement, including any amendments thereof and supplements thereof and supplements thereof.

(x) Exchange Listing. The issued and outstanding shares of Company Class A Common Stock are registered pursuant to Section 12(b) of the Exchange Act, and are listed for trading on Nasdaq under the symbol "ISPO." Except as disclosed in the Company Reports, there is no suit, action, proceeding or investigation pending or, to the knowledge of the Company, threatened against the Company by Nasdaq or the SEC, respectively, to prohibit or terminate the listing of the Company Class A Common Stock on Nasdaq, suspend trading of the Company Class A Common Stock on Nasdaq or to deregister the Company Class A Common Stock under the Exchange Act. The Company has taken no action that is designed to terminate or expected to result in the termination of the registration of the Company Class A Common Stock under the Exchange Act.

(y) <u>Solvency</u>. On the Closing Date, after giving effect to the Transactions, the Company and its Subsidiaries, on a consolidated basis, will be Solvent.

(z) <u>Reserved</u>.

(aa) <u>Membership Subscriptions and Property Inventory</u>. The information set forth in the Company Reports with respect to the number of membership subscriptions to the Company and the Company's property inventory is complete, true and accurate in all material respects as of the respective dates set forth therein.

(bb) No Additional Representations.

(i) The Company acknowledges that the Purchaser makes no representation or warranty as to any matter whatsoever except as expressly set forth in Section 3.02, in any certificate delivered by the Purchaser pursuant to this Agreement or as expressly set forth in any other Transaction Agreement, and the Company has not relied on or been induced by such information or any other representations or warranties (whether express or implied or made orally or in writing) not expressly set forth in Section 3.02, in any certificate delivered by the Purchaser pursuant to this Agreement or as expressly set forth in any other Transaction Agreement.

(ii) The Company acknowledges and agrees that, except for the representations and warranties expressly set forth in Section 3.02, in any certificate delivered by the Purchaser pursuant to this Agreement or as expressly set forth in any other Transaction Agreement, (i) no person has been authorized by the Purchaser to make any representation or warranty relating to the Purchaser or otherwise in connection with the transactions contemplated hereby, and if made, such representation or warranty must not be relied upon by the Company as having been authorized by the Purchaser, and (ii) any materials or information provided or addressed to the Company or any of its Representatives are not and shall not be deemed to be or include representations or warranties of the Purchaser unless any such materials or information are the subject of any express representation or warranty set forth in Section 3.02 of this Agreement, in any certificate delivered by the Purchaser pursuant to this Agreement or as expressly set forth in any other Transaction Agreement.

3.02 <u>Representations and Warranties of the Purchaser</u>. The Purchaser represents and warrants to the Company, as of the date hereof, as of the Closing Date and as of any Option Closing Date, as follows:

(a) <u>Organization; Ownership</u>. The Purchaser is a limited liability company, duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite limited liability company power and authority to own, operate and lease its properties and to carry on its business as it is being conducted on the date of this Agreement.

(b) Authorization; Sufficient Funds; No Conflicts.

(i) The Purchaser has full limited liability company power and authority to execute and deliver this Agreement and to consummate the Transactions to which it is a party. The execution, delivery and performance by the Purchaser of this Agreement, the Transaction Documents to which it is party and the consummation of the Transactions to which it is a party have been duly authorized by all necessary limited liability company action on behalf of such Purchaser. No other proceedings on the part of the Purchaser are necessary to authorize the execution, delivery and performance by such Purchaser of this Agreement and consummation of the Transactions to which it is a party. This Agreement has been duly and validly executed and delivered by the Purchaser. Assuming this Agreement constitutes the valid and binding obligation of the Company, this Agreement is a valid and binding obligation of the Purchaser, enforceable against the Purchaser in accordance with its terms, subject to the limitation of such enforcement by the Enforceability Exceptions.

(ii) The Purchaser will have as of the Closing Date cash in immediately available funds sufficient to pay the Tranche 1 Purchase Price or the Tranche 2 Purchase Price, as applicable.

(iii) The execution, delivery and performance of this Agreement by the Purchaser, the consummation by the Purchaser of the Transactions and the compliance by the Purchaser with any of the provisions hereof and thereof will not conflict with, violate or result in a breach of any provision of, or constitute a default under, or result in the termination of or accelerate the performance required by, or result in a right of termination or acceleration under, (A) any provision of the Purchaser's certificate of formation, operating agreement or other organizational document or (B) any permit, license, judgment, Order, decree, ruling, injunction, statute, law, ordinance, rule or regulation applicable to the Purchaser or any of its Affiliates, other than in the case of clause (B) as would not reasonably be expected to, individually or in the aggregate, prevent, materially impair or materially delay the ability of the Purchaser to consummate the Transactions.

(c) <u>Consents and Approvals</u>. Assuming the accuracy of the representations of the Company set forth herein, no consent, approval, Order or authorization of, or registration, declaration or filing with, or exemption or review by, any Governmental Entity is required on the part of the Purchaser in connection with the execution, delivery and performance by the Purchaser of this Agreement and the consummation by the Purchaser of the Transactions, except for any requirements or regulations in connection with the issuance of shares of Company Common Stock upon the exercise of the Warrant and any consent, approval, Order, authorization, registration, declaration, filing, exemption or review the failure of which to be obtained or made, individually or in the aggregate, would not reasonably be expected to prevent, materially impair or materially delay the ability of the Purchaser to consummate the Transactions.

(d) <u>Securities Act Representations</u>. The Purchaser is an accredited investor (as defined in Rule 501 of the Securities Act) and is aware that the sale of the Securities is being made in reliance on a private placement exemption from registration under the Securities Act. The Purchaser is acquiring the Securities (including any shares of Company Common Stock issuable upon exercise of the Warrant) for its own account, and not with a view toward, or for sale in connection with, any distribution thereof in violation of any federal or state securities or "blue sky" law, or with any present intention of distributing or selling the Securities (including any shares of Company Common Stock issuable upon exercise of the Warrant) in violation of the Securities Act. The Purchaser has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the Securities (including any shares of Company Common Stock issuable upon exercise of the Warrant) and is capable of bearing the economic risks of such investment. The Purchaser has been provided a reasonable opportunity to undertake and has undertaken such investigation and has been provided with and has evaluated such documents and information as it has deemed necessary to enable it to make an informed and intelligent decision with respect to the execution, delivery and performance of this Agreement.

(e) <u>Brokers and Finders</u>. The Purchaser has not retained, utilized or been represented by, or otherwise become obligated to, any broker, placement agent, financial advisor or finder in connection with the transactions contemplated by this Agreement whose fees the Company would be required to pay.

(f) <u>Proxy Statement</u>. None of the information supplied by Purchaser in writing to the Company about Purchaser or its Affiliates specifically for inclusion in the Proxy Statement will, on the date the Proxy Statement is filed, mailed, distributed or disseminated, as applicable, to the Company's stockholders and at the time of the Company Special Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, except that no representation or warranty is made by Purchaser with respect to information supplied by or on behalf of the Company for inclusion in the Proxy Statement.

(g) Legends. The Purchaser acknowledges that, upon issuance, the Securities will bear a legend substantially in the following form:

THE ISSUANCE AND SALE OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR APPLICABLE STATE SECURITIES LAWS. THE SECURITIES MAY NOT BE OFFERED FOR SALE, SOLD, TRANSFERRED OR ASSIGNED EXCEPT (A) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT FOR THE SECURITIES UNDER THE ACT, OR (B) IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE ACT, INCLUDING PURSUANT TO RULE 144 THEREUNDER, WHERE THE COMPANY HAS RECEIVED AN OPINION OF COUNSEL, IN A GENERALLY ACCEPTABLE FORM, THAT REGISTRATION IS NOT REQUIRED UNDER THE ACT.

(h) No Additional Representations.

(i) The Purchaser acknowledges that the Company does not make any representation or warranty as to any matter whatsoever except as expressly set forth in Section 3.01, in any certificate delivered by the Company pursuant to this Agreement or as expressly set forth in any other Transaction Agreement, and specifically (but without limiting the generality of the foregoing), that, except as expressly set forth in Section 3.01, in any certificate delivered by the Company pursuant to this Agreement or as expressly set forth in any other Transaction Agreement, the Company makes no representation or warranty with respect to (A) any matters relating to the Company, its business, financial condition, results of operations, prospects or otherwise, (B) any projections, estimates or budgets delivered or made available to the Purchaser (or any of its Affiliates or other Representatives) of future revenues, results of operations (or any component thereof), cash flows or financial condition (or any component thereof) of the Company or (C) the future business and operations of the Company, and the Purchaser has not relied on or been induced by such information or any other representations or warranties (whether express or implied or made orally or in writing) not expressly set forth in Section 3.01, in any certificate delivered by the Company pursuant to this Agreement or as expressly set forth in any other Transaction Agreement.

(ii) The Purchaser has conducted its own independent review and analysis of the business, operations, assets, liabilities, results of operations, financial condition and prospects of the Company and acknowledges such Purchaser has been provided with sufficient access for such purposes. The Purchaser acknowledges and agrees that, except for the representations and warranties expressly set forth in Section 3.01, in any certificate delivered by the Company pursuant to this Agreement or as expressly set forth in any other Transaction Agreement, (i) no person has been authorized by the Company to make any representation or warranty relating to itself or its business or otherwise in connection with the transactions contemplated hereby, and if made, such representation or warranty must not be relied upon by such Purchaser as having been authorized by the Company, and (ii) any estimates, projections, data, financial information, memoranda, presentations or any other materials or information provided or addressed to the Purchaser or any of its Affiliates or Representatives are not and shall not be deemed to be or include representations or warranties of the Company unless any such materials or information are the subject of any express representation or warranty set forth in Section 3.01 of this Agreement, in any certificate delivered by the Company pursuant to this Agreement, in any certificate delivered by the Company pursuant or as expressly set forth in any other Transaction Agreement or as expressly set forth in any other Transaction Agreement.

ARTICLE IV ADDITIONAL AGREEMENTS

4.01 <u>Taking of Necessary Action</u>. Each party hereto agrees to use its reasonable best efforts to promptly take or cause to be taken all action and promptly to do or cause to be done all things necessary, proper or advisable under applicable laws and regulations to consummate and make effective the Tranche 1 Purchase and the Tranche 2 Purchase, subject to the terms and conditions hereof and compliance with applicable law. In case at any time before or after the Tranche 1 Closing or the Tranche 2 Closing any further action is necessary or desirable to carry out the purposes of the Tranche 1 Purchase or the Tranche 2 Purchase, as applicable, the proper officers, managers and directors of each party to this Agreement shall take all such Necessary Action as may be reasonably requested by the requesting party.

4.02 Exclusive Dealing. The Company shall immediately cease and cause to be terminated all existing discussions and negotiations with any parties with respect to any proposal that constitutes or may be reasonably expected to constitute or lead to an Alternative Transaction Proposal. From the date of this Agreement until the earlier of the Tranche 2 Closing or the termination of this Agreement in accordance with its terms, the Company shall not, and shall use reasonable best efforts to cause its Representatives not to, directly or indirectly: (i) solicit, initiate, knowingly encourage (including by means of furnishing or disclosing information), knowingly facilitate, discuss or negotiate, directly or indirectly, any inquiry, proposal or offer (written or oral) with respect to an Alternative Transaction Proposal; (ii) furnish or disclose any non-public information to any Person in connection with, or that would reasonably be expected to lead to, an Alternative Transaction Proposal; (iii) enter into any contract or other arrangement or understanding regarding an Alternative Transaction Proposal; or (iv) otherwise cooperate in any way with, or assist or participate in, or knowingly facilitate or knowingly encourage any effort or attempt by any Person to do or seek to do any of the foregoing. The Company agrees to (A) notify Purchaser promptly upon receipt of any Alternative Transaction Proposal in reasonable detail and (B) keep Purchaser reasonably informed on a current basis of any material modifications to such offer or information. Notwithstanding the foregoing, if the Company (x) reasonably determines that the conditions to the parties' obligations to the Purchaser of such determination, the Company may discuss with hird parties potential alternative financing transactions (it being understood that the Company shall not thereby be relieved of any of its other obligations hereunder, including under Section 4.01).

4.03 Proxy Statement. The Company will, no later than the close of business on the third Business Day after the Tranche 1 Closing, prepare and file with the SEC the Proxy Statement in preliminary form, and the Company will use its reasonable best efforts to respond as promptly as reasonably practicable to any comments of the SEC with respect thereto. The Company will notify Purchaser promptly (and in any case no later than 24 hours) of the receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements to the Proxy Statement or for additional information and will supply Purchaser with copies of all correspondence between the Company or any of its Representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to the Proxy Statement. The Company covenants and agrees that the information in the Proxy Statement will not, at the time that the Proxy Statement or any amendment or supplement thereto is filed with the SEC or is first mailed to the stockholders of the Company contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading. If at any time prior to receipt of the Company Stockholder Approvals there will occur any event that should be set forth in an amendment or supplement to the Proxy Statement, including correcting any information that has become false or misleading in any material respect, the Company will promptly prepare and mail to its stockholders such an amendment or supplement. Purchaser and its counsel will be given a reasonable opportunity to review the Proxy Statement before it is filed with the SEC and the Company will give due consideration to all reasonable additions, deletions, or changes thereto suggested by Purchaser and its counsel. The Company will (a) establish a record date, (b) commence a broker search pursuant to Section 14a-13 of the Exchange Act in connection therewith and (c) thereafter commence mailing the Proxy Statement to the Company's stockholders as promptly as practicable after (i) the first Business Day after the date that is ten (10) calendar days after filing the Proxy Statement in preliminary form if, prior to such date, the SEC does not provide comments or indicates that it does not plan to provide comments or (ii) the date on which the Company shall have been informed by the SEC staff that it has no further comments on the document. The Proxy Statement shall include the Company Board Recommendation and shall not contain any proposals other than (i) the Nasdaq Proposal, (ii) any proposal that either the SEC or Nasdaq (or the respective staff members thereof) indicates is necessary in its comments to the Proxy Statement or in correspondence related thereto, (iii) each other proposal reasonably agreed to by the Company and Purchaser as necessary or appropriate in connection with the consummation of the Transactions; and (iv) a proposal for the postponement or adjournment of the Company Special Meeting, if necessary to permit further solicitation of proxies because there are not sufficient votes to approve and adopt any of the foregoing (collectively, the "Transaction Proposals").

4.04 Company Special Meeting. The Company shall duly call, give notice of, convene and hold a meeting of its stockholders (the "Company Special Meeting") for the purpose of seeking the Company Stockholder Approval and use its reasonable best efforts to solicit from the stockholders of the Company proxies in favor of the approval of the Nasdaq Proposal and to take all other action necessary or advisable to secure the vote or consent of the stockholders of the Company required by the Company's organizational documents, the rules of Nasdaq and the DGCL to obtain the Company Stockholder Approval. The Company will schedule the Company Special Meeting to be held within twenty-one (21) Business Days of the initial mailing of the Proxy Statement; provided that, the Company shall be permitted to postpone or adjourn the Company Special Meeting, but only (a) if the Company is unable to obtain a quorum of its stockholders at such time, to the extent necessary in order to obtain a quorum of its stockholders, (b) if there are not sufficient affirmative votes in Person or represented by proxy at such meeting to obtain the Company Stockholder Approval, to allow reasonable time for solicitation of proxies for purposes of obtaining the Company Stockholder Approval, (c) if the Board of Directors has determined in good faith, after consultation with Purchaser, that such delay is required by applicable law to comply with comments made by the SEC with respect to the Proxy Statement or to allow reasonable time for the mailing of any supplemental or amended disclosure required thereby or (d) if the Company is required to do so by a court of competent jurisdiction; provided, further, that: (i) to the extent permitted by applicable law, the Company may not postpone or adjourn the Company Special Meeting by more than 10 calendar days past the originally scheduled date without Purchaser's prior consent (such consent not to be unreasonably delayed, conditioned or withheld), (ii) if applicable, the Company shall respond as promptly as reasonably practicable to resolve any SEC comments and (iii) the Company shall reconvene the Company Special Meeting at the earliest practicable date on which Board of Directors reasonably expects to have sufficient affirmative votes to obtain the Company Stockholder Approval. The Company shall not present or allow any proposal to be presented at the Company Special Meeting other than the Transaction Proposals. Neither the Board of Directors nor any committee thereof may directly or indirectly (A) withhold, withdraw (or amend, qualify or modify in a manner adverse to Purchaser), or publicly propose to withdraw (or amend, qualify or modify in a manner adverse to Purchaser), the Company Board Recommendation; (B) propose publicly to recommend, adopt or approve any Alternative Transaction Proposal; or (C) fail to include the Company Board Recommendation in the Proxy Statement (any of the foregoing clauses (A) through (C), a "Company Board Recommendation Change"), in each case, except as required by applicable law. Notwithstanding any Company Board Recommendation Change, unless this Agreement has been validly terminated pursuant to Section 2.03, the Company shall duly call, give notice of, convene and hold the Company Special Meeting for the purpose of seeking the Company Stockholder Approval in compliance with this Agreement.

4.05 <u>Transaction Litigation</u>. The Company shall notify the Purchaser in writing promptly after learning of any stockholder or equityholder demands or other stockholder or equityholder proceedings (including derivative claims) relating to this Agreement or any of the Transactions (collectively, the "<u>Transaction Litigation</u>") commenced against the Company or any of its Representatives. The Company shall (i) keep the Purchaser reasonably informed regarding any Transaction Litigation, (ii) give the Purchaser the opportunity to, at its own cost and expense, participate in the defense, settlement and compromise of any such Transaction Litigation, and (iii) consider in good faith the Purchaser's advice with respect to any such Transaction Litigation. In no event shall the Company or any of its Representatives settle or compromise any Transaction Litigation without the prior written consent of the Purchaser (not to be unreasonably withheld, conditioned or delayed).

4.06 Board Representation.

(a) Effective as of the Tranche 1 Closing, the Company shall have appointed Payam Zamani as its Chief Executive Officer and Chairman pursuant to the terms of the employment agreement in the form as agreed as of the date hereof, which shall become effective as of the Tranche 1 Closing, and Mr. Zamani shall have received a grant of restricted stock units as set forth in such employment agreement.

(b) Effective as of the Tranche 1 Closing and for so long as the Purchaser and its Affiliates collectively Beneficially Own 15% or more of the Company Common Stock (the "<u>Tranche 1 Ownership Condition</u>"), the Purchaser shall have the right to designate an aggregate of two members of the Board of Directors (together with the designees contemplated by Section 4.06(c), the "<u>Purchaser Designees</u>"), one of whom shall serve as Chair of the Board of Directors, and the Company shall use its reasonable best efforts to promptly take or cause to be taken all Necessary Action as may be required to effect approval and designation of such Purchaser Designees to the Board of Directors (it being understood that as of the Tranche 1 Closing Mr. Zamani in his capacity as Chairman of the Board of Directors shall be deemed to be one of the Purchaser Designees).

(c) Effective immediately following the Tranche 2 Closing and for so long as the Purchaser and its Affiliates collectively Beneficially Own 40% or more of the Company Common Stock (the "<u>Tranche 2 Ownership Condition</u>"), the Purchaser shall have the right to designate an aggregate of three members of the Board of Directors (including Mr. Zamani) and the Company shall use its reasonable best efforts to promptly take or cause to be taken all Necessary Action as may be required to effect approval and designation of such Purchaser Designees to the Board of Directors.

(d) From and after the date of the Tranche 1 Closing, and for so long as the Purchaser and its Affiliates satisfy the Tranche 1 Ownership Condition, the Board of Directors shall be composed of seven (7) directors (or such other number of directors as approved by the Purchaser and its Affiliates), one of whom shall be the Chief Executive of the Company who shall serve as Chairman of the Board of Directors for so long as Payam Zamani is serving in the role of Chief Executive Officer of the Company.

(e) Any Purchaser Designee may be removed or have their nomination withdrawn (in each case of removal or withdrawal, with or without cause) from time to time and at any time by the Purchaser upon notice to the Company, and may otherwise only be removed for cause. The Purchaser and the Company acknowledge that Nasdaq rules impose on the Company certain governance rules including as it relates to the composition of the Board of Directors when the Purchaser's Beneficial Ownership of Company Common Stock is reduced over time, and the Purchaser and the Company agree to cooperate in good faith and shall cooperate to the extent necessary to comply with such rules.

(f) The Company shall enter into indemnification agreements and maintain customary D&O liability insurance for the benefit of each Purchaser Designee elected or appointed to the Board of Directors with respect to all periods during which such individual is a member of the Board of Directors, on terms, conditions and amounts substantially similar to the terms, conditions and amounts of the Company's current D&O liability insurance policy, and shall use is reasonable best efforts to cause such indemnification and insurance to be maintained in full force and effect.

(g) The Company shall reimburse the Purchaser Designees for all reasonable out-of-pocket expenses incurred in connection with their attendance at meetings of the Board of Directors and any committees thereof as contemplated by its reimbursement policies as in effect from time to time.

4.07 Confidentiality. Until one year after the Purchaser no longer has the right to designate any Purchaser Designees, the Purchaser will, and will direct its Affiliates and its and their respective Representatives to, keep confidential any non-public, confidential information concerning the Company that may be furnished to the Purchaser, its Affiliates, their respective Representatives or the Purchaser Designees by or on behalf of the Company or their Representatives pursuant to this Agreement (collectively referred to as the "Confidential Information"), provided that the Confidential Information shall not include information that (i) was or becomes available to the public other than as a result of a disclosure by the Purchaser, its Affiliates or any of their respective Representatives in violation of this Section 4.07, (ii) was or becomes available to the Purchaser, its Affiliates or any of their respective Representatives from a source other than the Company or their Representatives, provided that such source is believed by the Purchaser not to be disclosing such information in violation of an obligation of confidentiality to the Company, (iii) at the time of disclosure is already in the possession of the Purchaser, its Affiliates or any of their respective Representatives, provided that such information is believed by the Purchaser not to be subject to an obligation of confidentiality (whether by agreement or otherwise) to the Company, or (iv) was independently developed by the Purchaser, its Affiliates or any of their respective Representatives without use of any Confidential Information. The Purchaser agrees, on behalf of itself and its Affiliates, its and their respective Representatives and the Purchaser Designees, that Confidential Information may be disclosed solely (i) to the Purchaser's Affiliates and its and their respective Representatives on a need-to-know basis, and (ii) in the event that the Purchaser, its Affiliates or any of its or their respective Representatives are requested or required by applicable law, regulation, judgment, stock exchange rule or other applicable judicial or governmental process, regulatory review or examination (including by deposition, interrogatory, request for documents, subpoena, civil investigative demand or similar process) to disclose any Confidential Information, in each of which instances the Purchaser, its Affiliates and its and their respective Representatives, as the case may be, shall, to the extent legally permitted, provide notice to the Company sufficiently in advance of any such disclosure so that the Company will have a reasonable opportunity to timely seek to limit, condition or quash such disclosure.

4.08 Indemnification.

(a) The Purchaser, its Affiliates and its and their respective Representatives (each, an "Indemnitee") shall be indemnified to the fullest extent permitted by law by the Company for any and all Losses to which such Indemnitees may become subject as a result of, arising in connection with, or relating to any actual or threatened claim, suit, action, arbitration, cause of action, complaint, allegation, demand letter, or proceeding, whether at law or at equity, direct or derivative and whether public or private, before any Governmental Entity, any arbitrator or other tribunal (each, an "Action") brought by any stockholder of the Company (other than the Purchaser, its Affiliates, or any stockholder in their capacity as a holder of the Subject Securities) (a "Stockholder Claim"), and regardless of whether such Stockholder Claim is against an Indemnitee, related to the Transactions; provided that the Company will not be liable to indemnify any Indemnitee for any such Losses to the extent that such Losses (i) have resulted from Purchaser's breach of this Agreement; (ii) have resulted from an Indemnitee's willful misconduct or fraud in connection with the Transactions; or (iii) relate to a proceeding before any Governmental Entity relating to the Purchaser's Tax or accounting treatment of the Transactions or disclosure regarding the same. The parties agree, for the avoidance of doubt, that this Section 4.08 shall not apply to any matter for which indemnification is otherwise provided in Section 5.05.

(b) Each Indemnitee shall give the Company prompt written notice (an "<u>Indemnification Notice</u>") of any Action it has actual knowledge of that might give rise to Losses for which an Indemnitee would reasonably be likely to be entitled to indemnification under this Section 4.08, which notice shall set forth a description of those elements of such Action of which such Indemnitee has knowledge and promptly deliver to the Company any complaints with respect to such Action or other documents provided to such Indemnitee in connection therewith; provided that any delay or failure to give such Indemnification Notice shall not affect the indemnification obligations of the Company hereunder except to the extent the Company is materially and actually prejudiced by such delay or failure.

(c) The Company shall have the right, exercisable by written notice to the applicable Indemnitee(s) within thirty (30) days of receipt of the applicable Indemnification Notice, to select counsel to defend and control the defense of any Stockholder Claim set forth in such Indemnification Notice and the Company shall pay all reasonable fees and expenses of such counsel; provided that the Company shall not be entitled to so select counsel or control the defense of any claim to the extent that (i) such claim seeks primarily non-monetary or injunctive relief against the Indemnitee or alleges any violation of criminal law, (ii) the Company does not, subsequent to its assumption of such defense in accordance with this clause (c), conduct the defense of such claim in good faith, (iii) any of the Indemnitees reasonably determines upon the advice of counsel that representation of all such Indemnitees by the same counsel would be prohibited by applicable codes of professional conduct, or (iv) in the event that, based on the reasonable advice of counsel for the applicable Indemnitee(s), there are one or more material defenses available to the applicable Indemnitee(s) that are not available to other defendants. If the Company does not assume the defense of any Stockholder Claim in accordance with this clause (c), the applicable Indemnitee(s) may continue to defend such claim at the sole cost of the Company and the Company may still participate in, but not control, the defense of such Stockholder Claim at the Company's sole cost and expense. In no event shall the Company, in connection with any Action or separate but substantially similar Actions arising out of the same general allegations, be liable for the fees and expenses of more than one separate firm of attorneys at any time for all Indemnitees chosen by the Purchaser together with its Affiliates, and one separate firm of local counsel, in addition to regular counsel, to the extent required in order to effectively defend the Action.

(d) No Indemnitee shall consent to a settlement of, or the entry of any judgment arising from, any Stockholder Claim for which such Indemnitee is entitled to indemnification pursuant to this Section 4.08 without the prior written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed). Except with the prior written consent of the applicable Indemnitee(s), the Company, in the defense of any Stockholder Claim for which such Indemnitee is entitled to indemnification pursuant to this Section 4.08, shall not consent to the entry of any judgment or enter into any settlement that (i) provides for injunctive or other nonmonetary relief affecting any Indemnitee or involves any finding or admission of any violation of law or admission of any wrongdoing by any Indemnitee, (ii) does not include as an unconditional term thereof the giving by each claimant or plaintiff to each such Indemnitee(s) of an unconditional release of such Indemnitee(s) from all liability with respect to such Action or (iii) imposes any material burden on Indemnified not fully indemnified hereunder. In any such Stockholder Claim where the Company has assumed control of the defense thereof pursuant to clause (c), the Company shall keep the applicable Indemnitee(s) reasonably informed as to the status of such Stockholder Claim at all stages thereof (including all settlement negotiations and offers), promptly submit to such Indemnitee(s) copies of all pleadings, responsive pleadings, motions and other similar legal documents and paper received or filed in connection therewith, permit such Indemnitee(s) and their respective counsels to confer with the Company and its counsel with respect to the conduct of the defense thereof, and permit such Indemnitee(s) and their respective counsel(s) a reasonable opportunity to review all legal papers to be submitted prior to their submission; provided that the Company and OpCo shall not be obligated to provide materials, documents or information the disclosure of which would reasonably be likely to jeopardize the attorney-client privilege between the Company and its counsel or violate applicable law. Nothing in this Section 4.08(d) shall in any way limit, affect or otherwise modify an Indemnitee's rights to indemnification under the Company's certificate of incorporation, by-laws, any applicable policies of the Company or its Subsidiaries or any other agreement between the Indemnitee and the Company or its Subsidiaries.

4.09 <u>Tax Matters</u>. The Company shall bear and pay any and all issue, transfer, stamp, documentary and other similar Taxes that may be payable in respect of any issuance or delivery of the Securities pursuant to this Agreement or any Company Common Stock or other securities issuable upon the exercise of the Warrant; provided that the Company shall not be required to pay any such Tax to the extent such Tax is payable in respect of any transfer involved in the issuance and delivery of shares of Company Common Stock upon the exercise of the Warrant in a name other than that in which the Warrant so converted was or were registered at the request of the registered holder.

ARTICLE V REGISTRATION RIGHTS

5.01 Registration Statement.

(a) As soon as reasonably practicable after the Tranche 2 Closing, and in any event within 15 days of such closing (or, if the Option has been exercised, within 15 days of the Option Closing), the Company will prepare and file and use reasonable best efforts to cause to be declared effective or otherwise become effective pursuant to the Securities Act (such effectiveness date, the "<u>Registration Date</u>") a Registration Statement or post-effective amendment to an existing Registration Statement in order to provide for resales of all Registrable Securities to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act, and will use its reasonable best efforts to keep the Registration Statement continuously effective under the Securities Act at all times until the Registration Date. Any Registration Statement filed pursuant to this Section 5.01 shall cover only Registrable Securities and shall be on Form S- 3 (or a successor form) if the Company is eligible to use such form.

(b) Subject to the provisions of Section 5.02, the Company will use its reasonable best efforts to keep the Registration Statement (or any replacement Registration Statement) continuously effective until the earlier of (such earlier date, the "<u>Registration Termination Date</u>"): (i) the date on which all Registrable Securities covered by the Registration Statement have been sold thereunder in accordance with the plan and method of distribution disclosed in the prospectus included in the Registration Statement and (ii) there otherwise cease to be any Registrable Securities.

5.02 Registration Limitations and Obligations.

Subject to Section 5.01, the Company will use reasonable best efforts to prepare such supplements or amendments (including a (a) post-effective amendment), if required by applicable law, to each applicable Registration Statement and file any other required document so that such Registration Statement will be Available at all times during the period for which such Registration Statement is, or is required pursuant to this Agreement to be, effective; provided that no such supplement, amendment or filing will be required during a Blackout Period or during such periods when SEC rules or staff guidance prohibit such Registration Statement from being Available. Notwithstanding anything to the contrary contained in this Agreement, the Company shall be entitled, from time to time, by providing written notice to the holders of Registrable Securities, to require such holders of Registrable Securities to suspend the use of the prospectus for sales of Registrable Securities under the Registration Statement during any Blackout Period. No sales may be made under the applicable Registration Statement during any Blackout Period. In the event of a Blackout Period, the Company shall (x) deliver to the holders of Registrable Securities a certificate signed by the chief executive officer, chief financial officer, general counsel or treasurer of the Company confirming that the conditions described in the definition of Blackout Period are met, which certificate shall contain an approximation of the anticipated delay, and (y) notify each holder of Registrable Securities promptly upon each of the commencement and the termination of each Blackout Period, which notice of termination shall be delivered to each holder of Registrable Securities no later than the close of business of the last day of the Blackout Period and shall include a reasonably detailed description of the event and/or reason for the Blackout Period. In connection with the expiration of any Blackout Period and without any further request from a holder of Registrable Securities, the Company to the extent necessary and as required by applicable law shall as promptly as reasonably practicable prepare supplements or amendments, including a post-effective amendment, to the Registration Statement or the prospectus, or any document incorporated therein by reference, or file any other required document so that the Registration Statement will be Available. A Blackout Period will be deemed to have expired when the Company has notified the holders of Registrable Securities that the Blackout Period is over and the Registration Statement is Available. Notwithstanding anything in this Agreement to the contrary, the absence of an Available Registration Statement at any time from and after the Registration Date shall be considered a Blackout Period and subject to the limitations described in the definition of Blackout Period, unless the Registration Statement is otherwise prohibited from being Available by an applicable SEC rule or staff guidance.

At any time that a Registration Statement is effective and prior to the Registration Termination Date, if a holder of Registrable (b) Securities delivers a notice to the Company (a "Take-Down Notice") stating that it, together with any other Persons, intend to sell at least \$5,000,000 in aggregate of Registrable Securities held by such holder and such other Persons; provided that, if the Purchasers and their Affiliates do not collectively own at least \$5,000,000 of Registrable Securities, they shall be permitted to deliver a Take-Down Notice to sell all of the Registrable Securities held by them (but such amount may not in any case be less than \$5,000,000 collectively of Registrable Securities), in each case, pursuant to the Registration Statement, then, the Company shall amend or supplement the Registration Statement as may be necessary and to the extent required by law so that the Registration Statement remains Available in order to enable such Registrable Securities to be distributed in an Underwritten Offering. In connection with any Underwritten Offering of Registrable Securities for which a holder delivers a Take-Down Notice and satisfies the dollar thresholds set forth in first sentence above, and where the Take-Down Notice contemplates marketing efforts not to exceed twenty-four (24) hours by the Company and the underwriters, the Company will use reasonable best efforts to cooperate and make its senior officers available for participation in such marketing efforts (which marketing efforts will not, for the avoidance of doubt, include a "road show" requiring such officers to travel outside of the city in which they are primarily located). The holder of Registrable Securities that delivered the applicable Take-Down Notice shall select the underwriter(s) for each Underwritten Offering; provided that the managing underwriter(s) (if there is only one underwriter, such underwriter shall be deemed to be the managing underwriter) of a marketed Underwritten Offering shall be reasonably acceptable to the Company. The Company shall select the counsel for the managing underwriter(s); provided that such counsel shall be reasonably acceptable to the underwriter(s) and the holder of Registrable Securities that delivered the applicable Take-Down Notice. Such holder shall determine the pricing of the Registrable Securities offered pursuant to any such Registration Statement, including the underwriting discount and fees payable by such holder to the underwriters in such Underwritten Offering. Such holder shall reasonably determine the timing of any such registration and sale. Such holder shall determine the applicable underwriting discount and other financial terms, and the holders of the Registrable Securities sold in the Underwritten Offering shall be solely responsible for all such discounts and fees payable to such underwritters in such Underwritten Offering. Without the consent of the holder of Registrable Securities that delivered the applicable Take-Down Notice, no Underwritten Offering pursuant to this Agreement shall include any securities other than Registrable Securities.

(c) Subject to Section 5.02(d), if the Company proposes to file a Registration Statement for equity securities or securities convertible into equity securities, whether on its own behalf or in connection with the exercise of any registration rights by any holder of registrable securities (other than (i) a registration relating solely to an employee benefit plan or employee stock plan, a dividend reinvestment plan, or a merger or a consolidation, (ii) a registration on Form S-4 or any successor form, (iii) a registration on Form S-4 or any successor form, (iii) a registration on Form S-4 or any successor form, (iii) a registration on Form S-8 or any successor form, or (iv) any amendments or supplements to a Registration Statement previously filed by the Company with the SEC), then the Company shall give prompt notice (the "Initial Notice"), no later than the date that is twenty (20) days prior to the intended filing date of such Registration Statement, to the holders of Registrable Securities, and the holders of Registrable Securities shall be entitled to include in such Registration Statement the Registration Rights"), to register such number of shares of Registrable Securities as each such holders may request and shall set forth (A) the anticipated filing date of such Registration Statement and (B) the aggregate number of Registrable Securities that is proposed to be included in such Registration Statement. Subject to Section 5.02(d), the Company shall include in such Registration Statement. Subject to Section 5.02(d), the Company shall include in such Registration Statement. Subject to Section 5.02(d), the specific statement and (B) the aggregate number of Registration Statement such Registrable Securities for which it has received written requests to register within ten (10) days after the Initial Notice has been given.

(d) If a registration pursuant to Section 5.02(b) or Section 5.02(c) involves an Underwritten Offering and the managing underwriter(s) of such proposed Underwritten Offering advises the Company or the holders of Registrable Securities that the total securities that the holders of Registrable Securities and any other Persons intend to include in such offering, or that the inclusion of certain holders of the Registrable Securities in such offering, would be reasonably likely to adversely affect the price, timing or distribution of the securities offered in such offering, then the number of securities proposed to be included in such registration shall be allocated among the Company, the holders of Registrable Securities and the holders of other registrable securities in such offering, such that the number of securities that each such Person shall be entitled to sell in the Underwritten Offering shall be included in the following order:

(i) in the case of an exercise of any registration rights by holders of Registrable Securities pursuant to Section 5.02(b):
(i) first, the securities held by the Person(s) exercising such registration rights pursuant to Section 5.02(b), *pro rata* based upon the number of Registrable Securities requested to be registered by each such Person in connection with such registration; (ii) second, the securities held by other Person(s), if any, validly exercising demand registration rights, *pro rata* based upon the number of Registrable Securities requested to be registered by each such Person in connection with such registration rights, *pro rata* based upon the number of Registrable Securities requested to be registered by each such Person in connection with such registration; and (iii) third, securities to be issued and sold by the Company in such registration;

(ii) In the case of an exercise of registration rights by Persons other than the holders of Registrable Securities: (i) first, the securities held by such other Person(s), *pro rata* based upon the number of registrable securities requested to be registered by each such Person in connection with such registration; (ii) second, the securities held by holders of Registrable Securities exercising their Piggyback Registration Rights, *pro rata* based upon the number of Registrable Securities requested to be registered by each such Person in connection with such registration; and (iii) third, securities to be issued and sold by the Company in such registration; and

(iii) in the case of any other registration: (i) first, the securities to be issued and sold by the Company in such registration; and (ii) second, the securities held by holders of Registrable Securities exercising their Piggyback Registration Rights and any other holders of registrable securities validly exercising their piggyback registration rights, *pro rata* based upon the number of securities requested to be registered by each such Person in connection with such registration.

(e) In connection with a distribution of Registrable Securities in which the holders of Registrable Securities are selling an aggregate of at least \$5,000,000 of Registrable Securities, the Company shall, to the extent requested by managing underwriter(s) of such a distribution, be subject to a restricted period of the same length of time as such holder agrees with the managing underwriter(s) (but not to exceed 90 days) during which the Company may not offer, sell or grant any option to purchase Company Common Stock and any debt securities of the Company other than issuances pursuant to the Company's employee or director stock plans and issuances of shares upon the exercise of options or other equity awards under such stock plans.

5.03 Registration Procedures.

(a) If and whenever the Company is required to use reasonable best efforts to effect the registration of any Registrable Securities under the Securities Act and in connection with any distribution of Registrable Securities pursuant thereto as provided in this Agreement (including any sale referred to in any Take-Down Notice), the Company shall as promptly as reasonably practicable, subject to the other provisions of this Agreement:

(i) use reasonable best efforts to prepare and file with the SEC a Registration Statement to effect such registration in accordance with the intended method or methods of distribution of such securities and thereafter use reasonable efforts to cause such Registration Statement to become and remain effective pursuant to the terms of this Section 5.03; provided, however, that the Company may discontinue any registration of its securities which are not Registrable Securities at any time prior to the effective date of the Registration Statement relating thereto; provided, further, that before filing such Registration Statement or any amendments or supplements thereto, including any prospectus supplements in connection with a sale referred to in a Take-Down Notice, the Company will furnish to the holders which are including Registrable Securities in such registration ("Selling Holders") and the lead managing underwriter(s), if any, copies of all such documents proposed to be filed, which documents will be subject to the review and reasonable comment (which comments will be considered in good faith by the Company) of the counsel (if any) to such holders and counsel (if any) to such underwriter(s), and other documents reasonably requested by any such counsel, including any comment letter from the SEC, and, if requested by any such counsel, provide such counsel and the lead managing underwriter(s), if any, reasonable opportunity to participate in the preparation of such Registration Statement and each prospectus (including any prospectus supplement) included or deemed included therein and such other opportunities to conduct a customary and reasonable due diligence investigation (in the context of a registered Underwritter Offering) of the Company, including reasonable access to (including responses to any reasonable inquiries by the lead managing underwriter(s) and their counsel) the Company's books and records, officers, accountants and other advisors;

(ii) prepare and file with the SEC such amendments and supplements to such Registration Statement and the prospectus used in connection therewith as may be necessary and to the extent required by applicable law to keep such Registration Statement effective and Available pursuant to the terms of this Section 5.03;

(iii) if requested by the lead managing underwriter(s), promptly include in a prospectus supplement or post-effective amendment such information as the lead managing underwriter(s), if any, and such holders may reasonably request in order to permit the intended method of distribution of such securities and make all required filings of such prospectus supplement or such post-effective amendment as soon as reasonably practicable after the Company has received such request; provided, however, that the Company shall not be required to take any actions under this Section 5.03(a)(iii) that are not, in the opinion of counsel for the Company, in compliance with applicable law;

(iv) furnish to the Selling Holders and each underwriter, if any, of the securities being sold by such Selling Holders such number of conformed copies of such Registration Statement and of each amendment and supplement thereto, such number of copies of the prospectus and any prospectus supplement contained in or deemed part of such Registration Statement (including each preliminary prospectus supplement) and each free writing prospectus (as defined in Rule 405 of the Securities Act) (a "Free Writing Prospectus") utilized in connection therewith and any other prospectus filed under Rule 424 under the Securities Act, in conformity with the requirements of the Securities Act, and such other documents as such Selling Holders and underwriter(s), if any, may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities owned by such Selling Holders;

(v) use reasonable best efforts to cause such Registrable Securities to be listed on each securities exchange on which similar securities issued by the Company are then listed;

(vi) use reasonable best efforts to provide and cause to be maintained a transfer agent and registrar for all Registrable Securities covered by such Registration Statement from and after a date not later than the effective date of such Registration Statement;

(vii) as promptly as practicable notify in writing the holders of Registrable Securities and the underwriters, if any, of the following events: (A) the filing of the Registration Statement, any amendment thereto, the prospectus or any prospectus supplement related thereto or post-effective amendment to such Registration Statement or any Free Writing Prospectus utilized in connection therewith, and, with respect to such Registration Statement or any post-effective amendment thereto, when the same has become effective; (B) any request by the SEC or any other U.S. or state governmental authority for amendments or supplements to such Registration Statement or the prospectus or for additional information; (C) the issuance by the SEC of any stop order suspending the effectiveness of such Registration Statement or the initiation of any proceedings by any person for that purpose; (D) the receipt by the Company of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the securities or "blue sky" laws of any jurisdiction or the initiation or threat of any proceeding for such purpose; (E) if at any time the representations and warranties of the Company contained in any agreement (including any underwriting agreement) related to such registration cease to be true and correct in any material respect; and (F) upon the happening of any event that makes any statement made in such Registration Statement or related prospectus or any document incorporated or deemed to be incorporated therein by reference untrue in any material respect or that requires the making of any changes in such Registration Statement, prospectus or documents so that, in the case of such Registration Statement, it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and that in the case of the prospectus, it will not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading;

(viii) use reasonable efforts to obtain the withdrawal of any order suspending the effectiveness of such Registration Statement, or the lifting of any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction at the earliest reasonable practicable date, except that the Company shall not for any such purpose be required to (A) qualify generally to do business as a foreign corporation or as a dealer in securities in any jurisdiction wherein it would not but for the requirements of this clause (viii) be obligated to be so qualified, (B) subject itself to taxation in any such jurisdiction or (C) file a general consent to service of process in any such jurisdiction;

(ix) cooperate with each seller of Registrable Securities and each underwriter or agent participating in the disposition of such Registrable Securities and their respective counsel in connection with any filings required to be made with the Financial Industry Regulatory Authority, Inc.;

(x) prior to any public offering of Registrable Securities, use reasonable efforts to register or qualify or cooperate with the Selling Holders in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities for offer and sale under the applicable state securities or "blue sky" laws of those jurisdictions within the United States as any holder reasonably requests in writing to keep each such registration or qualification (or exemption therefrom) effective until the Registration Termination Date; provided that the Company will not be required to (A) qualify generally to do business as a foreign corporation or as a dealer in securities in any jurisdiction wherein it would not but for the requirements of this clause (x) be obligated to be so qualified, (B) subject itself to taxation in any such jurisdiction or (C) file a general consent to service of process in any such jurisdiction;

(xi) use reasonable efforts to cooperate with the holders to facilitate the timely preparation and delivery of certificates or book-entry securities representing Registrable Securities to be delivered to a transferee pursuant to the Registration Statements, which certificates or book-entry securities shall be free, to the extent permitted by applicable law, of all restrictive legends, and to enable such Registrable Securities to be in such denominations and registered in such names as any such holders may request in writing; and in connection therewith, if required by the Company's transfer agent, the Company will promptly after the effectiveness of the Registration Statement cause to be delivered to its transfer agent when and as required by such transfer agent from time to time, any authorizations, certificates, directions and other evidence required by the transfer agent which authorize and direct the transfer agent to issue such Registrable Securities without legend upon sale by the holder of such shares of Registrable Securities under the Registration Statement; and

(xii) agrees with each holder of Registrable Securities that, in connection with any Underwritten Offering or other resale pursuant to the Registration Statement in accordance with the terms hereof, it will use reasonable efforts to negotiate in good faith and execute all customary indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements (in each case on terms reasonably acceptable to the Company), including using reasonable efforts to procure customary legal opinions and auditor "comfort" letters.

(b) The Company may require each Selling Holder and each underwriter, if any, to (i) furnish the Company in writing such information regarding each Selling Holder or underwriter and the distribution of such Registrable Securities as the Company may from time to time reasonably request in writing to complete or amend the information required by such Registration Statement and/or any other documents relating to such registered offering, and (ii) execute and deliver, or cause the execution or delivery of, and to perform under, or cause the performance under, any agreements and instruments reasonably requested by the Company to effectuate such registered offering, including, without limitation, opinions of counsel and questionnaires. If the Company requests that the holders of Registrable Securities take any of the actions referred to in this Section 5.03(b), such holders shall take such action promptly and as soon as reasonably practicable following the date of such request.

(c) Each Selling Holder agrees that upon receipt of any notice from the Company of the happening of any event of the kind described in clauses (B), (C), (D), (E) and (F) of Section 5.03(a)(vii), such Selling Holder shall forthwith discontinue such Selling Holder's disposition of Registrable Securities pursuant to the applicable Registration Statement and prospectus relating thereto until such Selling Holder is advised in writing by the Company that the use of the applicable prospectus may be resumed, and has received copies of any additional or supplemental filings that are incorporated or deemed to be incorporated by reference in such prospectus. The Company shall use reasonable efforts to cure the events described in clauses (B), (C), (D), (E) and (F) of Section 5.03(a)(vii) so that the use of the applicable prospectus may be resumed at the earliest reasonably practicable moment.

5.04 <u>Expenses</u>. The Company shall pay all Registration Expenses in connection with a registration pursuant to this Article V, provided that each holder of Registrable Securities participating in an offering shall pay all applicable underwriting fees, discounts, selling commissions and similar charges.

5.05 <u>Registration Indemnification</u>.

The Company agrees, without limitation as to time, to indemnify and hold harmless, to the fullest extent permitted by law, each (a) Selling Holder and its Affiliates and their respective Representatives and each Person who controls (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) such Selling Holder or such other indemnified Person and the Representatives of each such controlling Person, each underwriter, if any, and each Person who controls (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) such underwriter (collectively, the "Indemnified Persons"), from and against all losses, claims, damages, liabilities, costs, expenses (including reasonable expenses of investigation and reasonable attorneys' fees and expenses), judgments, fines, penalties, charges and amounts paid in settlement (collectively, the "Losses"), as incurred, arising out of, caused by, resulting from or relating to any untrue statement (or alleged untrue statement) of a material fact contained in any Registration Statement, prospectus or preliminary prospectus or Free Writing Prospectus, in each case related to such Registration Statement, or any amendment or supplement thereto or any omission (or alleged omission) of a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading and (without limitation of the preceding portions of this Section 5.05(a)) will reimburse each such Selling Holder, each of its Affiliates, and each of their respective Representatives and each such Person who controls each such Selling Holder and the Representatives of each such controlling Person, each such underwriter and each such Person who controls any such underwriter, for any reasonable legal and any other reasonable expenses incurred in connection with investigating and defending or settling any such claim, Loss, damage, liability or action, except insofar as the same are caused by any information regarding a holder of Registrable Securities or underwriter furnished in writing to the Company expressly for use therein by any such person, any Affiliate or controlling Person thereof.

(b) In connection with any Registration Statement in which a Selling Holder is participating, without limitation as to time, each such Selling Holder shall, severally and not jointly, indemnify the Company, its directors and officers, and each Person who controls (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) the Company, from and against all Losses, as incurred, arising out of, caused by, resulting from or relating to any untrue statement (or alleged untrue statement) of material fact contained in the Registration Statement, prospectus or preliminary prospectus or Free Writing Prospectus or any amendment or supplement thereto or any omission (or alleged omission) of a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and (without limitation of the preceding portions of this Section 5.05(b)) will reimburse the Company, its directors and officers and each Person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) for any reasonable legal and any other reasonable expenses incurred in connection with investigating and defending or settling any such claim, Loss, damage, liability or action, in each case solely to the extent, but only to the extent, that such untrue statement or omission is made in such Registration Statement, prospectus or preliminary prospectus or any amendment or supplement thereto in reliance upon and in conformity with written information regarding the Selling Holder furnished to the Company by such Selling Holder for inclusion in such Registration Statement, prospectus or Free Writing Prospectus or any amendment or supplement thereto.

(c) Any Person entitled to indemnification hereunder shall give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification; provided, however, the failure to give such notice shall not release the indemnifying party from its obligation, except to the extent that the indemnifying party has been actually and materially prejudiced by such failure to provide such notice on a timely basis.

(d) In any case in which any such action is brought against any indemnified party, the indemnified party shall promptly notify in writing the indemnifying party of the commencement thereof, and the indemnifying party will be entitled to participate therein, and, to the extent that it may wish, to assume the defense thereof, with counsel reasonably satisfactory to such indemnified party, and after notice from the indemnifying party to such indemnified party of its election so to assume the defense thereof and acknowledging the obligations of the indemnifying party with respect to such proceeding, the indemnifying party will not (so long as it shall continue to have the right to defend, contest, litigate and settle the matter in question in accordance with this paragraph) be liable to such indemnified party hereunder for any legal or other expense subsequently incurred by such indemnified party in connection with the defense thereof other than reasonable costs of investigation, supervision and monitoring (unless (i) such indemnified party reasonably objects to such assumption on the grounds that there may be defenses available to it which are different from or in addition to the defenses available to such indemnifying party and, as a result, a conflict of interest exists or (ii) the indemnifying party shall have failed within a reasonable period of time to assume such defense and the indemnified party is or would reasonably be expected to be materially prejudiced by such delay, in either event the indemnified party shall be promptly reimbursed by the indemnifying party for the expenses incurred in connection with retaining one separate legal counsel (for the avoidance of doubt, for all indemnified parties in connection therewith)). For the avoidance of doubt, notwithstanding any such assumption by an indemnifying party, the indemnified party shall have the right to employ separate counsel in any such matter and participate in the defense thereof, but the fees and expenses of such counsel shall be at the expense of such indemnified party except as provided in the previous sentence. An indemnifying party shall not be liable for any settlement of an action or claim effected without its consent (which consent shall not be unreasonably withheld, conditioned or delayed).

No matter shall be settled by an indemnifying party without the consent of the indemnified party (which consent shall not be unreasonably withheld, conditioned or delayed), unless such settlement (x) includes an unconditional release of such indemnified party from all liability on claims that are the subject matter of such claim or proceeding, (y) does not include any statement as to or any admission of fault, culpability or a failure to act by or on behalf of any indemnified party and (z) is settled solely for cash for which the indemnified party would be entitled to indemnification hereunder. The failure of an indemnified party to give notice to an indemnifying party of any action brought against such indemnified party shall not relieve the indemnifying party of its obligations or liabilities pursuant to this Agreement, except to the extent such failure materially and adversely prejudices the indemnifying party.

(e) The indemnification provided for under this Agreement shall survive the sale or other transfer of the Registrable Securities and the termination of this Agreement.

If recovery is not available under the foregoing indemnification provisions for any reason or reasons other than as specified therein, (f) any Person who would otherwise be entitled to indemnification by the terms thereof shall nevertheless be entitled to contribution with respect to any Losses with respect to which such Person would be entitled to such indemnification but for such reasons, in such proportion as is appropriate to reflect the relative fault of the indemnifying party, on the one hand, and such indemnified party, on the other hand, in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party, the Persons' relative knowledge and access to information concerning the matter with respect to which the claim was asserted, the opportunity to correct and prevent any statement or omission, and other equitable considerations appropriate under the circumstances. It is hereby agreed that it would not necessarily be equitable if the amount of such contribution were determined by pro rata or per capita allocation that does not take into account the equitable considerations referred to in the immediately preceding sentence. Notwithstanding any other provision of this Agreement, no holder of Registrable Securities shall be required to contribute, in the aggregate, any amount in excess of its net proceeds from the sale of the Registrable Securities subject to any actions or proceedings over the amount of any damages, indemnity or contribution that such holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not found guilty of such fraudulent misrepresentation.

(g) The indemnification and contribution agreements contained in this Section 5.05 are in addition to any liability that the indemnifying party may have to the indemnified party and do not limit other provisions of this Agreement that provide for indemnification.

5.06 <u>Facilitation of Sales Pursuant to Rule 144</u>. For as long as the Purchaser or its Affiliates Beneficially Owns the Securities, including any Company Common Stock issued or issuable upon exercise or conversion thereof, to the extent it shall be required to do so under the Exchange Act, the Company shall use reasonable best efforts to timely file the reports required to be filed by it under the Exchange Act or the Securities Act (including the reports under Sections 13 and 15(d) of the Exchange Act referred to in subparagraph (c)(1) of Rule 144) and submit all required Interactive Data Files (as defined in Rule 11 of Regulation S-T of the SEC), and shall use reasonable best efforts to take such further necessary action as any holder of the Securities may reasonably request in connection with the removal of any restrictive legend on the Securities being sold, all to the extent required from time to time to enable such holder to sell the Securities without registration under the Securities Act within the limitations of the exemption provided by Rule 144.

ARTICLE VI MISCELLANEOUS

6.01 <u>Survival of Representations and Warranties</u>. Except for the warranties and representations contained in clauses Section 3.01(a), Section 3.01(b), Section 3.01(c), Section 3.01(g), Section 3.01(h)(i), Section 3.02(a), Section 3.02(b)(i), Section 3.02(b)(iii)(A) and Section 3.02(e), which shall survive indefinitely, the warranties and representations made herein shall survive for one (1) year following the Tranche 1 Closing Date or, if the Tranche 2 Closing occurs, the Tranche 2 Closing Date, and shall then expire; provided that nothing herein shall relieve any party of liability for any inaccuracy or breach of such representation or warranty to the extent that any good faith allegation of such inaccuracy or breach is made in writing prior to such expiration.

6.02 <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given if delivered personally, sent by overnight courier or sent via email (with receipt confirmed) as follows:

(a) If to the Purchaser, to:

One Planet Group LLC 1820 Bonanza Street Walnut Creek, CA 94596 Attention: Payam Zamani Email: payam@zamani.org

with a copy (which will not constitute actual or constructive notice) to:

Davis Polk & Wardwell LLP 450 Lexington Avenue New York, NY 10017 Attention: Derek Dostal Email: derek.dostal@davispolk.com

(b) If to the Company, to:

Inspirato Incorporated 1544 Wazee Street Denver, CO 80202 Attention: Brent Wadman Legal@inspirato.com

with a copy (which will not constitute actual or constructive notice) to:

Davis Graham & Stubbs LLP 1550 17th Street, Suite 500 Denver, CO 80202 Attention: John Elofson john.elofson@davisgraham.com

or to such other address or addresses as shall be designated in writing. All notices shall be deemed effective (a) when delivered personally (with written confirmation of receipt, by other than automatic means, whether electronic or otherwise), (b) when sent by email (with written confirmation of receipt, by other than automatic means, whether electronic or otherwise), (b) when sent by email (with written confirmation of receipt, by other than automatic means, whether electronic or otherwise) or (c) one (1) Business Day following the day sent by overnight courier. Any party may change its address for notice by providing notice of such change in address to the other parties in accordance with the foregoing.

6.03 Entire Agreement; Third-Party Beneficiaries; Amendment. This Agreement, together with the Warrant and the Voting Agreements, sets forth the entire agreement between the parties hereto with respect to the Transactions, and are not intended to and shall not confer upon any person other than the parties hereto, their successors and permitted assigns any rights or remedies hereunder; provided that (i) Section 5.05 shall be for the benefit of and fully enforceable by each of the Indemnified Persons and (ii) Section 6.12 shall be for the benefit of and fully enforceable by each of the Specified Persons. Any provision of this Agreement may be amended or modified in whole or in part at any time by an agreement in writing between the parties hereto executed in the same manner as this Agreement. No failure on the part of any party to exercise, and no delay in exercising, any right shall operate as a waiver thereof nor shall any single or partial exercise by any party of any right preclude any other or future exercise thereof or the exercise of any other right.

6.04 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed to constitute any original, but all of which together shall constitute one and the same document. Signatures to this Agreement transmitted by facsimile transmission, by electronic mail in "portable document format" (".pdf") form, or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document will have the same effect as physical delivery of the paper document bearing the original signature. The words "execution," "signed," "signature," "delivery," and words of like import in or relating to this Agreement or any document to be signed in connection with this Agreement shall be deemed to include electronic signatures, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be, and the parties hereto consent to conduct the transactions contemplated hereunder by electronic means.

6.05 <u>Public Announcements</u>. No press release or public announcement related to this Agreement or the transactions contemplated herein shall be issued or made by the Purchaser or its Affiliates without the prior written approval of the Company or by the Company or its Affiliates without the prior written approval of the Purchaser, unless in either case such announcement is required by law (based on the advice of counsel) in which case the Company or the Purchaser (as applicable) shall have the right to review and reasonably comment on such press release, announcement or communication prior to issuance, distribution or publication. The restrictions of this Section 6.05 shall not apply to any communication if the information contained therein substantially reiterates (or is consistent with) previous press release or public announcement made in compliance with this Section 6.05 or substantially reiterates (or is consistent with) any press release or public announcement made by the Company. For the avoidance of doubt, this Section 6.05 shall not be deemed to restrict any filings required by securities laws, including without limitation filings required to be made with the SEC on Schedule 13D, Form 3 and/or Form 4 with respect to the consummation of transactions contemplated by this Agreement.

6.06 <u>Expenses</u>. Except as otherwise expressly provided herein, each party hereto shall bear its own costs and expenses (including attorneys' fees) incurred in connection with this Agreement and the Transactions contemplated hereby, except that promptly following the Tranche 2 Closing Date the Company shall reimburse the Purchaser for all of its reasonable out-of-pocket costs and expenses (including attorneys' fees) in an amount not to exceed \$125,000.

6.07 <u>Successors and Assigns</u>. Except as otherwise expressly provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, each party's successors and assigns, and no other person; provided that no party hereto may assign its respective rights or delegate its respective obligations under this Agreement, whether by operation of law or otherwise, and any assignment by a party hereto in contravention of this Section 6.07 shall be null and void; provided that (i) Purchaser may assign all of its rights and obligations under this Agreement to one or more Affiliates; provided, further that no such assignment will relieve the Purchaser of its obligations hereunder, (ii) any Affiliate of Purchaser who after the date hereof or the Tranche 2 Closing Date, as applicable, executes and delivers a joinder (in form and substance reasonably satisfactory to the Company) and is a transferee of the Warrant or shares of Company Common Stock shall be deemed a Purchaser hereunder and have all the rights and obligations of such Purchaser or any portion thereof, (iii) the rights of a holder of Registrable Securities under Article V may be assigned, but only together with a transfer of the Warrant (or a portion thereof). For the avoidance of doubt, no Third Party to whom any the shares of Company Common Stock issued upon exercise of the Warrant are transferred shall have any rights or obligations under this Agreement except as expressly agreed by the Company.

6.08 Governing Law; Jurisdiction; Waiver of Jury Trial.

This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to (a) any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. In addition, each of the parties hereto irrevocably agrees that any legal action or proceeding with respect to this Agreement and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Court of Chancery in the State of Delaware and, if such court declines jurisdiction, any other state court of the State of Delaware or the United States District Court for the District of Delaware. Each of the parties hereto hereby irrevocably submits with regard to any such action or proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts and agrees that it will not bring any action relating to this Agreement or any of the transactions contemplated by this Agreement in any court other than the aforesaid courts. Each of the parties hereto hereby irrevocably waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement, (i) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason other than the failure to serve in accordance with this Section 6.08(a), (ii) any claim that it or its property is exempt or immune from the jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (iii) to the fullest extent permitted by the applicable law, any claim that (A) the suit, action or proceeding in such court is brought in an inconvenient forum, (B) the venue of such suit, action or proceeding is improper or (C) this Agreement, or the subject matter hereof, may not be enforced in or by such courts. Each of the parties hereby agrees that service of any process, summons, notice or document by U.S. registered mail to the respective addresses set forth in Section 6.02 shall be effective service of process for any suit or proceeding in connection with this Agreement or the transactions contemplated hereby.

(b) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS CONTAINED IN THIS SECTION 6.08.

6.09 <u>Severability</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable, the remaining provisions of this Agreement shall remain in full force and effect; provided that the economic and legal substance of, any of the Transactions is not affected in any manner materially adverse to any party. In the event of any such determination, the parties agree to negotiate in good faith to modify this Agreement to fulfill as closely as possible the original intent and purpose hereof. To the extent permitted by law, the parties hereby to the same extent waive any provision of law that renders any provision hereof prohibited or unenforceable in any respect.

6.10 <u>Specific Performance</u>. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, each party agrees that in the event of any breach or threatened breach by any other party of any covenant or obligation contained in this Agreement, the non-breaching party shall be entitled (in addition to any other remedy that may be available to it, whether in law or equity) to (i) a decree or Order of specific performance to enforce the observance and performance of such covenant or obligation, and (ii) an injunction restraining such breach or threatened breach. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that any other party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity. Any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with any such Order or injunction.

6.11 <u>Headings</u>. The headings of Articles and Sections contained in this Agreement are for reference purposes only and are not part of this Agreement.

6.12 <u>Non-Recourse</u>.

(a) This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby may only be brought against the entities that are expressly named as parties hereto and their respective successors and assigns. Except as set forth in the immediately preceding sentence, no past, present or future director, officer, employee, incorporator, member, partners, stockholder, Affiliate, agent, attorney or Representative of any party hereto (collectively, the "Specified Persons") shall have any liability for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby.

(b) Notwithstanding anything to the contrary in this Agreement, the Purchaser's aggregate liability for any liability, loss, damage or recovery of any kind (including special, exemplary, consequential, indirect or punitive damages or damages arising from loss of profits, business opportunities or goodwill, diminution in value or any other losses or damages, whether at law, in equity, in contract, in tort or otherwise) arising under or in connection with any breach of this Agreement (whether willfully, intentionally, unintentionally or otherwise) or the failure of the Tranche 1 Closing or the Tranche 2 Closing to occur for any reason or otherwise in connection with the Transactions or this Agreement or in respect of any oral representations made or alleged to have been made in connection therewith shall be no greater than the Purchase Price and the Purchaser shall not have any further liability or obligation relating to or arising out of this Agreement, the Transactions or any other agreement or document relating thereto in excess of such amount.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, this Agreement has been executed by the parties hereto or by their respective duly authorized officers, ali as of the date first above written.

INSPIRATO INCORPORATED

By: /s/Eri se Name: Eri se Title: Chief Executive Officer

[Signature Page to Investment Agreement]

IN WITNESS WHEREOF, this Agreement has been executed by the parties hereto or by their respective duly authorized officers, all as of the date first above written.

ONE PLANET GROUP LLC

By: /s/ Payam Zamani Name: Payam Zamani

Title: President and Chief Executive Officer

[Signature Page to Investment Agreement]

EXHIBIT A FORM OF WARRANT

Exhibit A

THIS WARRANT AND THE UNDERLYING SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "*ACT*"), OR UNDER THE SECURITIES LAWS OF ANY STATE. THESE SECURITIES MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED EXCEPT AS PERMITTED UNDER THE ACT AND APPLICABLE STATE SECURITIES LAWS IN ACCORDANCE WITH APPLICABLE REGISTRATION REQUIREMENTS OR AN EXEMPTION THEREFROM. THE ISSUER OF THESE SECURITIES MAY REQUIRE AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE, TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT AND ANY APPLICABLE STATE SECURITIES LAWS. THIS WARRANT MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OF ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

WARRANT TO PURCHASE SHARES OF CLASS A COMMON STOCK

of

INSPIRATO INCORPORATED

Dated as of [·], 2024¹

Warrant to Purchase up to 2,915,451 Shares of Class A Common Stock (subject to adjustment)

THIS CERTIFIES THAT, for value received, One Planet Group LLC, a Delaware limited liability company, or its registered assigns (the "*Holder*"), is entitled, subject to the provisions and upon the terms and conditions set forth herein, to purchase from Inspirato Incorporated, a Delaware corporation (the "*Company*"), shares of the Company's Class A Common Stock, \$0.0001 par value per share ("*Class A Common Stock*" and such shares, the "*Shares*"), in the amounts, at such times and at the price per share set forth in <u>Section 1</u>. The term "*Warrant*" as used herein shall include this Warrant and any warrants delivered in substitution or exchange therefor as provided herein. This Warrant is issued in connection with that certain Investment Agreement between the Company and the Holder dated August 12, 2024 (the "*Investment Agreement*"). The following is a statement of the rights of the Holder and the conditions to which this Warrant is subject, and to which the Holder, by acceptance of this Warrant, agrees:

1. Number and Price of Shares; Exercise Period.

(a) *Number of Shares.* Subject to any previous exercise of the Warrant, the Holder shall have the right to purchase up to 2,915,451 Shares, as may be adjusted pursuant hereto, prior to (or in connection with) the expiration of this Warrant as provided in <u>Section 8</u>; *provided, however*, that if the Holder exercises the Option, the Company agrees to promptly amend this Warrant following the Option Closing to increase the number of shares issuable pursuant to this Warrant by the number of warrants included in the Optional Securities (the terms "Option," "Option Closing" and "Optional Securities" being defined in the Investment Agreement).

¹ To be issued and dated as of the Tranche 2 Closing Date (as defined in the Investment Agreement).

(b) *Exercise Price*. The exercise price per Share shall be equal to \$3.43, subject to adjustment pursuant hereto (the "*Exercise Price*").

(c) *Exercise Period.* This Warrant shall be exercisable, whether in whole or in part, beginning on the date hereof (the "*Commencement Date*") and until the expiration of this Warrant as set forth in <u>Section 8</u>.

2. Exercise of the Warrant.

(a) *Exercise.* The purchase rights represented by this Warrant may be exercised at the election of the Holder, in whole or in part, in accordance with <u>Section 1</u>, by (i) in the case of a cash exercise, (1) the tender to the Company at its principal office (or such other office or agency as the Company may designate) of a notice of exercise in the form of <u>Exhibit A</u> (the "*Notice of Exercise*"), duly completed and executed by or on behalf of the Holder, together with the surrender of this Warrant; and (2) the payment to the Company of an amount equal to (x) the Exercise Price multiplied by (y) the number of Shares being purchased, by cash or a wire transfer of immediately available funds payable to the Company or (ii) in the case of a Cashless Exercise, duly completed and executed by or on behalf of the Holder, together with the surrender of this Warrant in accordance with <u>Section 2(d)</u>.

(b) *Stock Certificates.* The rights under this Warrant shall be deemed to have been exercised and the Shares issuable upon such exercise shall be deemed to have been issued immediately prior to the close of business on the date this Warrant is exercised in accordance with its terms, and the person entitled to receive the Shares issuable upon such exercise shall be treated for all purposes as the holder of record of such Shares as of the close of business on such date. As promptly as reasonably practicable on or after such date, the Company shall issue and deliver to the person or persons entitled to receive the same a notice of uncertificated shares for that number of shares issuable upon such exercise. In the event that the rights under this Warrant are exercised in part and have not expired, the Company shall execute and deliver a new Warrant reflecting the number of Shares that remain subject to this Warrant.

(c) *No Fractional Shares or Scrip.* No fractional shares or scrip representing fractional shares shall be issued upon the exercise of the rights under this Warrant, and the Holder hereby waives any rights to any such fractional shares without consideration.

(d) **Cashless Exercise.** This Warrant may be exercised at the election of the Holder, in whole or in part, in accordance with <u>Section 1</u>, by tender to the Company at its principal office (or such other office or agency as the Company may designate) of a Notice of Exercise that includes the Holder's intention to effect a cashless exercise, including a calculation of the number of Shares to be issued upon such exercise in accordance with the terms hereof (a "**Cashless Exercise**"). In the event of a Cashless Exercise, in lieu of paying the Exercise Price in cash, the Holder shall surrender this Warrant for that number of Shares determined by multiplying (i) the number of Shares to which it would otherwise be entitled by (ii) a fraction, the numerator of which shall be the difference between the then current Market Price per share of the Class A Common Stock, and the Exercise Price per share of the Class A Common Stock, and the denominator of which shall be the then current Market Price per share of Class A Common Stock. For example, if the Holder is exercising this Warrant to purchase 100,000 Shares with an Exercise Price of \$3.43 per Share through a Cashless Exercise when the Class A Common Stock's Market Price per share is \$10.00 per share, then upon such Cashless Exercise the Company shall deliver to the Holder 65,700 Shares. Solely for purposes of this <u>Section 2(d)</u>, the "*Market Price*" shall mean the 10-day VWAP of the Class A Common Stock for the ten (10) consecutive trading days ending on the trading day prior to the date on which the Notice of Exercise is sent to the Company shall be.



(e) **Reservation of Stock.** The Company agrees that during the term the rights under this Warrant are exercisable to reserve and keep available from its authorized and unissued shares of Class A Common Stock for the purpose of effecting the exercise of this Warrant such number of shares as shall from time to time be sufficient to effect the exercise of the rights under this Warrant; and if at any time the number of authorized but unissued shares of Class A Common Stock shall not be sufficient for purposes of the exercise of this Warrant in accordance with its terms, without limitation of such other remedies as may be available to the Holder, the Company will use commercially reasonable efforts to take such corporate action as may, in the opinion of counsel, be necessary to increase its authorized and unissued shares of its Class A Common Stock to a number of shares as shall be sufficient for such purposes.

3. **Replacement of the Warrant.** Subject to the receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form and substance to the Company or, in the case of mutilation, on surrender and cancellation of this Warrant, the Company at the expense of the Holder shall execute and deliver, in lieu of this Warrant, a new warrant of like tenor and amount.

4. Transfer of the Warrant.

(a) *Warrant Register.* The Company shall maintain a register (the "*Warrant Register*") containing the name and address of the Holder or Holders. Until this Warrant is transferred on the Warrant Register in accordance herewith, the Company may treat the Holder as shown on the Warrant Register as the absolute owner of this Warrant for all purposes, notwithstanding any notice to the contrary. Any Holder of this Warrant (or of any portion of this Warrant) may change its address as shown on the Warrant Register by written notice to the Company requesting a change.

(b) *Warrant Agent.* The Company may appoint an agent for the purpose of maintaining the Warrant Register referred to in <u>Section 4(a)</u>, issuing the Shares or other securities then issuable upon the exercise of the rights under this Warrant, exchanging this Warrant, replacing this Warrant or conducting related activities.

(c) **Transferability of the Warrant.** Subject to the provisions of this Warrant with respect to compliance with the Securities Act of 1933, as amended (the "Securities Act"), and limitations on assignments and transfers, including without limitation compliance with the restrictions on transfer set forth in Section 5, title to this Warrant may be transferred by endorsement (by the transferor and the transferee executing the assignment form attached as Exhibit B (the "Assignment Form")) and delivery in the same manner as a negotiable instrument transferable by endorsement and delivery.

-3-

(d) *Exchange of the Warrant upon a Transfer.* On surrender of this Warrant (and a properly endorsed Assignment Form) for exchange, subject to the provisions of this Warrant with respect to compliance with the Securities Act and limitations on assignments and transfers, the Company shall issue to or on the order of the Holder a new warrant or warrants of like tenor, in the name of the Holder or as the Holder (on payment by the Holder of any applicable transfer taxes) may direct, for the number of shares issuable upon exercise hereof, and the Company shall register any such transfer upon the Warrant Register. This Warrant (and the securities issuable upon exercise of the rights under this Warrant) must be surrendered to the Company or its warrant or transfer agent, as applicable, as a condition precedent to the sale, pledge, hypothecation or other transfer of any interest in any of the securities represented hereby.

(e) *Taxes.* In no event shall the Company be required to pay any tax which may be payable in respect of any transfer involved in the issue and delivery of any certificate, or a book entry, in a name other than that of the Holder, and the Company shall not be required to issue or deliver any such certificate, or make such book entry, unless and until the person or persons requesting the issue or entry thereof shall have paid to the Company the amount of such tax or shall have established to the satisfaction of the Company that such tax has been paid or is not payable.

5. Restrictions on Transfer of the Warrant and Shares; Compliance with Securities Laws. By acceptance of this Warrant, the Holder agrees to comply with the following:

(a) **Restrictions on Transfers.** Except as set forth in Section 5(b), this Warrant may not be transferred or assigned in whole or in part without the Company's prior written consent, and any attempt by the Holder to transfer or assign any rights, duties or obligations that arise under this Warrant without such permission shall be void. Any transfer of this Warrant or the Shares (the "*Warrant Securities*") must be in compliance with all applicable federal and state securities laws. The Holder agrees not to make any sale, assignment, transfer, pledge or other disposition of all or any portion of the Warrant Securities unless and until the transferee thereof has agreed in writing for the benefit of the Company to take and hold such Warrant Securities subject to, and to be bound by, the terms and conditions set forth in this Warrant to the same extent as if the transferee were the original Holder hereunder, and

(i) there is then in effect a registration statement under the Securities Act covering such proposed disposition and such disposition is made in accordance with such registration statement (or such proposed disposition is exempt from the registration requirements under the U.S. securities laws), or

-4-

(ii) (A) such Holder shall have given prior written notice to the Company of such Holder's intention to make such disposition and shall have furnished the Company with a detailed description of the manner and circumstances of the proposed disposition, (B) the transferee shall have confirmed to the satisfaction of the Company in writing that the Warrant Securities are being acquired (i) solely for the transferee's own account and not as a nominee for any other party, (ii) for investment and (iii) not with a view toward distribution or resale, and shall have confirmed such other matters related thereto as may be reasonably requested by the Company, and (C) if requested by the Company, such Holder shall have furnished the Company, at the Holder's expense, with (i) an opinion of counsel or other evidence, reasonably satisfactory to the Company, to the effect that such disposition will not require registration of such Warrant Securities under the Securities Act or (ii) a "no action" letter from the SEC to the effect that the transfer of such Warrant Securities without registration will not result in a recommendation by the staff of the SEC that action be taken with respect thereto, whereupon such Holder shall be entitled to transfer such Warrant Securities in accordance with the terms of the notice delivered by the Holder to the Company.

(b) **Permitted Transfers.** Notwithstanding the provisions of <u>Section 5(a)</u>, the Holder is permitted to transfer, in whole or in part and from time to time, the Warrant Securities to one or more Affiliates of the Holder in accordance with the terms hereof, and without the Company's prior written consent; *provided*, in each case, that the Holder shall give written notice to the Company of the Holder's intention to effect such disposition and shall have furnished the Company with a detailed description of the manner and circumstances of the proposed disposition.

(c) *Securities Law Legend.* Each certificate, instrument or book entry representing the Warrant Securities shall (unless otherwise permitted by the provisions of this Warrant) be notated with a legend substantially similar to the following (in addition to any legend required by state securities laws):

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "*ACT*"), OR UNDER THE SECURITIES LAWS OF CERTAIN STATES. THESE SECURITIES MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED EXCEPT AS PERMITTED UNDER THE ACT AND APPLICABLE STATE SECURITIES LAWS IN ACCORDANCE WITH APPLICABLE REGISTRATION REQUIREMENTS OR AN EXEMPTION THEREFROM. THE ISSUER OF THESE SECURITIES MAY REQUIRE AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT AND ANY APPLICABLE STATE SECURITIES LAWS. THIS CERTIFICATE MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OF ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

(d) *Instructions Regarding Transfer Restrictions.* The Holder consents to the Company making a notation on its records and giving instructions to any transfer agent in order to implement the restrictions on transfer established in this <u>Section 5</u>.

(e) **Removal of Legend.** The legend referring to federal and state securities laws identified in <u>Section 5(c)</u> notated on any certificate, instrument or book entry representing the Warrant Securities and the stock transfer instructions and record notations with respect to such securities shall be removed and the Company shall issue a certificate, instrument or book entry representing the Warrant Securities are registered under the Securities Act, or (ii) such holder provides the Company with an opinion of counsel reasonably acceptable to the Company to the effect that a sale or transfer of such securities may be made without registration, qualification or legend.



6. Adjustments. Subject to the expiration of this Warrant pursuant to <u>Section 8</u>, the number and kind of shares purchasable hereunder and the Exercise Price therefor are subject to adjustment from time to time, as follows:

(a) **Reclassification of Shares.** If the securities issuable upon exercise of this Warrant are changed into the same or a different number of securities of any other class or classes by reclassification, capital reorganization or otherwise (other than as otherwise provided for herein) (a "**Reclassification**"), then, in any such event, in lieu of the number of Shares which the Holder would otherwise have been entitled to receive, the Holder shall have the right thereafter to exercise this Warrant for a number of shares of such other class or classes of stock that a holder of the number of securities deliverable upon exercise of this Warrant immediately before that change would have been entitled to receive in such Reclassification, all subject to further adjustment as provided herein with respect to such other shares. For the avoidance of doubt, with respect to any such Reclassification in which an adjustment is made under this section, no adjustment shall be made under <u>Section 7</u> that would result in a duplicative adjustment.

(b) **Subdivisions and Combinations.** In the event that the outstanding shares of Class A Common Stock are subdivided (by stock split, by payment of a stock dividend or otherwise) into a greater number of shares of such securities, the number of Shares issuable upon exercise of the rights under this Warrant immediately prior to such subdivision shall, concurrently with the effectiveness of such subdivision, be proportionately increased, and the Exercise Price shall be proportionately decreased, and in the event that the outstanding shares of Class A Common Stock are combined (by reclassification or otherwise) into a lesser number of shares of such securities, the number of Shares issuable upon exercise of the rights under this Warrant immediately prior to such combination shall, concurrently with the effectiveness of such combination, be proportionately decreased, and the Exercise Price shall be proportionately with the effectiveness of such combination, be proportionately decreased, and the Exercise Price shall be proportionately accessed.

(c) Notice of Adjustments. Upon any adjustment in accordance with this <u>Section 6</u>, the Company shall give notice thereof to the Holder, which notice shall state the event giving rise to the adjustment, the Exercise Price as adjusted and the number of securities or other property purchasable upon the exercise of the rights under this Warrant, setting forth in reasonable detail the method of calculation of each. The Company shall, upon the written request of any Holder, furnish or cause to be furnished to such Holder a certificate setting forth (i) such adjustments, (ii) the Exercise Price at the time in effect and (iii) the number of securities and the amount, if any, of other property that at the time would be received upon exercise of this Warrant.

-6-

7. Fundamental Change Transactions.

If, at any time while this Warrant is outstanding, in one or more related transactions, (i) the Company directly or indirectly effects (a) any merger or consolidation of the Company with or into another Person (other than a merger or consolidation in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding shares of Class A Common Stock), (ii) the Company or any subsidiary thereof directly or indirectly effects any sale, lease, license, assignment, transfer, conveyance or other disposition of all or substantially all of the Company's assets, (iii) any, direct or indirect, purchase offer, tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Class A Common Stock are permitted to sell, tender or exchange their shares for other securities, cash or property and has been accepted by the holders of more than 50% of the voting power of the common equity of the Company and the shares of Class A Common Stock are converted into or exchanged for other securities, cash or property, (iv) the Company, directly or indirectly, in one or more related transactions effects any reclassification, reorganization or recapitalization of the Class A Common Stock or any compulsory share exchange pursuant to which the Class A Common Stock is effectively converted into or exchanged for other securities, cash or property, or (v) the Company directly or indirectly consummates a stock or share purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, merger or scheme of arrangement) with another Person or group of Persons whereby such other Person or group acquires more than 50% of the outstanding shares of Class A Common Stock or more than 50% of the voting power of the common equity of the Company and the shares of Class A Common Stock are converted into or exchanged for other securities, cash or property (each a "Fundamental Change"), then, upon any subsequent exercise of this Warrant, the Holder shall have the right to receive, for each Share that would have been issuable upon such exercise immediately prior to the occurrence of such Fundamental Change, at the option of the Holder, the number of shares of common stock of the successor or acquiring corporation or of the Company, if it is the surviving corporation, and any additional consideration (the "Alternate Consideration") receivable as a result of such Fundamental Change by a holder of the number of shares of Class A Common Stock for which this Warrant is exercisable immediately prior to such Fundamental Change; provided, however, that (i) if the holders of the Class A Common Stock were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets constituting the Alternate Consideration for which the Warrant shall become exercisable shall be deemed to be the weighted average of the kind and amount received per share by the holders of Class A Common Stock in such consolidation or merger that affirmatively make such election, and (ii) if a tender, exchange or redemption offer shall have been made to and accepted by the holders of Class A Common Stock under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act (or any successor rule)) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act (or any successor rule)) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act (or any successor rule)) more than 50% of the outstanding shares of Class A Common Stock, the Holder shall be entitled to receive the highest amount of cash, securities or other property to which the Holder would actually have been entitled as a stockholder if the Holder had exercised the Warrant prior to the expiration of such tender or exchange offer, accepted such offer and all of the Class A Common Stock held by the Holder had been purchased pursuant to such tender or exchange offer, subject to adjustments (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in this Section 7(a); provided further that if less than 70% of the consideration receivable by the holders of the Class A Common Stock in the applicable event is payable in the form of common stock in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the Holder properly exercises the Warrant within thirty (30) days following the public disclosure of the consummation of such applicable event by the Company pursuant to a Current Report on Form 8-K filed with the Commission, the Exercise Price shall be reduced by an amount (in dollars) equal to the difference (but in no event less than zero) of (i) the Exercise Price in effect prior to such reduction minus (ii) (A) the Per Share Consideration (as defined below) minus (B) the Black-Scholes Warrant Value (as defined below). The "Black-Scholes Warrant Value" means the value of the Warrant immediately prior to the consummation of the applicable event based on the Black-Scholes Warrant Model for a Capped American Call on Bloomberg Financial Markets ("Bloomberg"). For purposes of calculating such amount, (1) the price of each share of Class A Common Stock shall be the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event, (2) the assumed volatility shall be the 90 day volatility obtained from the HVT function on Bloomberg determined as of the trading day immediately prior to the day of the announcement of the applicable event and (3) the assumed risk-free interest rate shall correspond to the U.S. Treasury rate for a period equal to the remaining term of the Warrant. "Per Share Consideration" means (i) if the consideration paid to holders of the Class A Common Stock consists exclusively of cash, the amount of such cash per share of Class A Common Stock, and (ii) in all other cases, the volume weighted average price of the Class A Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the effective date of the applicable event. In no event will the Exercise Price be reduced to less than the par value per share issuable upon exercise of the Warrant.

-7-

(b) The Company shall cause any successor entity in a Fundamental Change in which the Company is not the survivor (the "Successor Entity") to assume in writing all of the obligations of the Company under this Warrant in accordance with the provisions of this Section 7 pursuant to written agreements in form and substance satisfactory to the Holder and approved by the Holder prior to such Fundamental Change and shall, at the option of the Holder, deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant which is exercisable for a corresponding number of shares of capital stock of such Successor Entity (or its parent entity) equivalent to the shares of Class A Common Stock acquirable and receivable upon exercise of this Warrant prior to such Fundamental Change, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (but taking into account the relative value of the shares of Class A Common Stock pursuant to such Fundamental Change and the value of such shares of capital stock, such number of shares of capital stock and such exercise price being for the purpose of protecting the economic value of this Warrant immediately prior to the consummation of such Fundamental Change), and which is satisfactory in form and substance to the Holder. Upon the occurrence of any such Fundamental Change, the Successor Entity shall succeed to, and be substituted for (so that from and after the date of such Fundamental Change, the provisions of this Warrant referring to the "Company" shall refer instead to the Successor Entity), and may exercise every right and power of the Company and shall assume all of the obligations of the Company under this Warrant with the same effect as if such Successor Entity had been named as the Company herein.

-8-

(c) The provisions of this <u>Section 7</u> shall apply similarly and equally to successive Fundamental Changes and shall be applied as if this Warrant (and any such subsequent warrants) were fully exercisable.

8. Expiration of the Warrant. This Warrant shall expire and shall no longer be exercisable as of the later of 5:00 p.m., Mountain time, on (i) the date which is five (5) years after the Commencement Date or (ii) in the case of a Fundamental Change which is publicly announced before the date described in (i) but which closes after the date described in (i), the closing date of such Fundamental Change (such time, the "*Expiration Time*").

For the avoidance of doubt, to the extent that the Warrant or any portion thereof is not exercised prior to the Expiration Time, it shall be automatically cancelled with no action required by the Company, and with no further rights thereunder, upon such expiration.

9. No Rights as a Stockholder. Nothing contained herein shall entitle the Holder to any rights as a stockholder of the Company or cause the Holder to be deemed the holder of any securities that may at any time be issuable on the exercise of the rights hereunder for any purpose nor shall anything contained herein be construed to confer upon the Holder, as such, any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, change of par value or change of stock to no par value, consolidation, merger, conveyance or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or any other rights of a stockholder of the Company until the rights under the Warrant shall have been exercised and the Shares purchasable upon exercise of the rights hereunder shall have become deliverable as provided herein.

10. Miscellaneous.

(a) *Amendments.* Except as expressly provided herein, neither this Warrant nor any term hereof may be amended, waived, discharged or terminated other than by a written instrument referencing this Warrant and signed by the Company and the Holder.

(b) *Waivers.* No waiver of any single breach or default shall be deemed a waiver of any other breach or default theretofore or thereafter occurring.

(c) Notices. Any notice, request, instruction or other document to be given hereunder by any party to the other shall be in writing and shall be deemed to have been duly given (i) if sent by United Parcel Service or FedEx on an overnight basis, signature receipt required, one business day after mailing, (ii) if sent by email, with a copy mailed on the same day (or next day, if such day is not a business day) in the manner provided in clause (i) of this Section 10(c), when transmitted and receipt is confirmed, or (iii) if otherwise personally delivered, when delivered with signature receipt required. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice.

-9-

If to the Company, to:

Name: Address:	Inspirato Incorporated 1544 Wazee Street			
	Denver, CO 80202			
Attn:	Brent Wadman			
Email:	legal@inspirato.com			
with a copy to (which copy alone shall not constitute notice):				
Name:	Davis Graham & Stubbs LLP			
Address:	1550 17th Street			
	Denver, CO 80202			
Attn:	John Elofson			
Email:	john.elofson@davisgraham.com			
If to the Holder, to:				
Name:	One Planet Group LLC			
Address:	1820 Bonanza Street			
	Walnut Creek, CA 94596			
Attn:	Payam Zamani			
Email:	payam@zamani.org			
with a copy to (which copy a	lone shall not constitute notice):			
Name:	Davis Polk & Wardwell LLP			
Address:	450 Lexington Avenue			
	New York, NY 10017			
Attn:	Derek Dostal			
Email:	derek.dostal@davispolk.com			

(d) Governing Law, Submission to Jurisdiction; Waiver of Jury Trial. This Warrant and all actions arising out of or in connection with this Warrant shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to the conflicts of law provisions or rules (whether of the State of Delaware or of any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. In addition, each of the parties expressly, irrevocably and unconditionally (a) submits to the personal jurisdiction and venue of the Court of Chancery of the State of Delaware, or if such court is unavailable, the United States District Court for Delaware (the "Chosen Courts"), in the event any dispute (whether in contract, tort or otherwise) arises out of this Warrant or the transactions contemplated hereby, (b) agrees that it shall not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court and waives any claim of lack of personal jurisdiction or improper venue and any claims that such courts are an inconvenient forum or that this Warrant or the transactions contemplated hereby may not be enforced in or by any such court, and (c) agrees that it shall not bring any claim, action, or proceeding relating to this Warrant or the transactions contemplated hereby in any court other than the Chosen Courts, and in stipulated preference ranking, of the preceding clause (a). Each party agrees that service of process upon such party in any such claim, action, or proceeding shall be effective if notice is given in accordance with the provisions of this Warrant. EACH PARTY HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY CLAIM, ACTION, OR PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER, OR IN CONNECTION WITH THIS WARRANT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY (i) CERTIFIES THAT NO REPRESENTATIVE, AGENT, OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (ii) ACKNOWLEDGES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS WARRANT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10(d).

-10-

(e) *Titles and Subtitles.* The titles and subtitles used in this Warrant are used for convenience only and are not to be considered in construing or interpreting this Warrant. All references in this Warrant to sections, paragraphs and exhibits shall, unless otherwise provided, refer to sections and paragraphs hereof and exhibits attached hereto.

(f) **Severability.** If any provision of this Warrant becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, portions of such provision, or such provision in its entirety, to the extent necessary, shall be severed from this Warrant, and such illegal, unenforceable or void provision shall be replaced with a valid and enforceable provision that will achieve, to the extent possible, the same economic, business and other purposes of the illegal, unenforceable or void provision. The balance of this Warrant shall be enforceable in accordance with its terms.

(g) *Saturdays, Sundays and Holidays.* If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall be a Saturday, Sunday or U.S. federal holiday, then such action may be taken or such right may be exercised on the next succeeding day that is not a Saturday, Sunday or U.S. federal holiday.

(h) *Rights and Obligations Survive Exercise of the Warrant.* Except as otherwise provided herein, the rights and obligations of the Company and the Holder under this Warrant shall survive exercise of this Warrant.

(i) **Entire Agreement.** Except as expressly set forth herein, the Investment Agreement and this Warrant (including the exhibits attached hereto) constitute the entire agreement and understanding of the Company and the Holder with respect to the subject matter hereof and supersede all prior agreements and understandings relating to the subject matter hereof.

(j) **Defined Terms Used in this Warrant.** In addition to the terms defined in this Warrant, the following terms used in this Warrant shall be construed to have the meanings set forth or referenced below.

-11-

(i) "*Affiliates*" means, with respect to a Person, any other Person that, directly or indirectly, through one (1) or more intermediaries, controls, is controlled by or is under common control with such Person. For purposes of this definition, "control" and, with correlative meanings, the terms "controlled by" and "under common control with" means (a) the possession, directly or indirectly, of the power to direct the management or policies of a Person, whether through the ownership of voting securities, by contract relating to voting rights or corporate governance, or otherwise; or (b) the ownership, directly or indirectly, of more than fifty percent (50%) of the voting securities or other ownership interest of a Person (or, with respect to a limited partnership or other similar entity, its general partner or controlling entity). The parties acknowledge that in the case of certain entities organized under the laws of certain countries outside of the United States, the maximum percentage ownership permitted by law for a foreign investor may be less than fifty percent (50%), and that, in such case, such lower percentage shall be substituted in the preceding sentence, *provided* that such foreign investor has the power to direct the management or policies of such entity.

- (ii) "Person" means any individual, corporation, partnership, trust, limited liability company, association or other entity.
- (iii) "SEC" means the U.S. Securities and Exchange Commission.

(iv) "*WWAP*" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Class A Common Stock is then listed or quoted on a trading market, the daily volume weighted average price of the Class A Common Stock for such date (or the nearest preceding date) on the trading market on which the Class A Common Stock is then listed or quoted as reported by Bloomberg (based on a trading day from 9:30 a.m. (Eastern time) to 4:00 p.m. (Eastern time)), (b) if OTCQB or OTCQX is not a trading market, the volume weighted average price of the Class A Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Class A Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Class A Common Stock are then reported on the Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Class A Common Stock so reported, or (d) in all other cases, the fair market value of a share of Class A Common Stock as determined by an independent appraiser selected by the Holder and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

11. Dispute Resolution.

(a) Submission to Dispute Resolution.

(i) In the case of a dispute relating to the Exercise Price, the Market Price, VWAP, or fair market value or the arithmetic calculation of the number of Shares issuable pursuant to this Warrant (as the case may be) (including, without limitation, a dispute relating to the determination of any of the foregoing), the Company or the Holder (as the case may be) shall submit the dispute to the other party via email (A) if by the Company, within five (5) business days after the occurrence of the circumstances giving rise to such dispute or (B) if the Holder, within five (5) business days after the locentrence of the circumstances giving rise to such dispute or (B) if the Holder, within five (5) business days after the Holder learned of the circumstances giving rise to such dispute. If the Holder and the Company are unable to promptly resolve such dispute relating to such Exercise Price, such Market Price, VWAP, or such fair market value or such arithmetic calculation of the number of Shares issuable pursuant to this Warrant (as the case may be), at any time after the second (2nd) business day following such initial notice by the Company or the Holder (as the case may be) of such dispute to the Company or the Holder (as the case may be), then the Company and the Holder may jointly select an independent, reputable accounting firm to resolve such dispute.

(ii) The Holder and the Company shall each deliver to such accounting firm (A) a copy of the initial dispute submission so delivered in accordance with the first sentence of this <u>Section 11</u> and (B) written documentation supporting its position with respect to such dispute, in each case, no later than 5:00 p.m. (Eastern time) by the fifth (5^{th}) business day immediately following the date on which the Company and the Holder jointly selected such accounting firm (the "*Dispute Submission Deadline*") (the documents referred to in the immediately preceding clauses (A) and (B) are collectively referred to herein as the "*Required Dispute Documentation*") (it being understood and agreed that if either the Holder or the Company fails to so deliver all of the Required Dispute Documentation by the Dispute Submission Deadline, then the party who fails to so submit all of the Required Dispute and such accounting firm shall resolve such dispute based solely on the Required Dispute Documentation that was delivered to such accounting firm prior to the Dispute Submission Deadline). Unless otherwise agreed to in writing by both the Company and the Holder or otherwise requested by such accounting firm, neither the Company nor the Holder shall be entitled to deliver or submit any written documentation).

(iii) The Company and the Holder shall cause such accounting firm to determine the resolution of such dispute and notify the Company and the Holder of such resolution no later than ten (10) business days immediately following the Dispute Submission Deadline. The fees and expenses of such accounting firm shall be borne equally by the Company and the Holder, and such accounting firm's resolution of such dispute shall be final and binding upon all parties absent manifest error.

(b) *Miscellaneous*. The Company and the Holder each expressly acknowledges and agrees that (i) this <u>Section 11</u> constitutes an agreement to arbitrate between the Company and the Holder (and constitutes an arbitration agreement) under the rules then in effect under § 7501, et seq. of the New York Civil Practice Law and Rules ("CPLR") and that the Company and the Holder are each authorized to apply for an order to compel arbitration pursuant to CPLR § 7503(a) in order to compel compliance with this <u>Section 11</u>, (ii) the terms of this Warrant shall serve as the basis for the selected accounting firm's resolution of the applicable dispute, such accounting firm shall be entitled (and is hereby expressly authorized) to make all findings, determinations and the like that such accounting firm determines are required to be made by such accounting firm in connection with its resolution of such dispute and in resolving such dispute such accounting firm shall apply such findings, determinations and the like to the terms of this Warrant, and (iii) nothing in this <u>Section 11</u> shall limit the Holder from obtaining any injunctive relief or other equitable remedies (including, without limitation, with respect to any matters described in this <u>Section 11</u>).

(signature page follows)

-13-

The Company and the Holder sign this Warrant as of the date stated on the first page.

INSPIRATO INCORPORATED

By: _____ Name: Title:

AGREED AND ACKNOWLEDGED: ONE PLANET GROUP LLC

By: Name: Title:

(Signature Page to Warrant to Purchase Shares Class A Common Stock of Inspirato Incorporated)

EXHIBIT A

FORM OF NOTICE OF EXERCISE

TO: INSPIRATO INCORPORATED (the "Company")

Attention: Chief Financial Officer

The undersigned, pursuant to the provisions set forth in the attached Warrant, hereby elects to purchase the number of Shares set forth below covered by such Warrant. The undersigned, in accordance with Section 2 of the Warrant, hereby agrees to pay the aggregate Exercise Price for such shares of Class A Common Stock. Upon surrender of the Warrant, duly endorsed, to the offices of the Company, a new warrant evidencing the remaining Shares covered by such Warrant but not yet exercised for and purchased, if any, should be issued in the name of the Holder. Capitalized terms used herein without definition are used as defined in the Warrant.

The undersigned represents and warrants that the aforesaid shares are being acquired for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, the distribution thereof, and that the undersigned has no present intention of selling, granting any participation in, or otherwise distributing the shares, nor does it have any contract, undertaking, agreement or arrangement for the same, and all representations and warranties of the undersigned set forth in Section 3.02 of the Investment Agreement are true and correct as of the date hereof.

Number of Shares with respect to which the Warrant is being exercised:

Aggregate Exercise Price to be paid in cash or by wire transfer:

Holder:			
By: Name:			
-			
Name:			

(Exhibit A – Form of Exercise)

EXHIBIT B

ASSIGNMENT FORM

ASSIGNOR:		
COMPANY:	INSPIRATO INCORPORAT)
WARRANT:	THE WARRANT TO PURC	ASE SHARES OF CLASS A COMMON STOCK ISSUED ON [•], 2024 (THE "WARRANT")
DATE:		
		holder of the Warrant (" <i>Assignor</i> ") assigns and transfers to the assignee named below (" <i>Assignee</i> ") all of , with respect to the number of shares set forth below:
	Name of Assignee:	
	Address of Assignee:	
	Number of Shares Assigned:	
	-	oint as attorney to make such transfer on the books of Inspirato Incorporated, ver of substitution in the premises.

- (2) **Obligations of Assignee.** Assignee agrees to take and hold the Warrant and any shares of stock to be issued upon exercise of the rights thereunder (the "*Warrant Securities*") subject to, and to be bound by, the terms and conditions set forth in the Warrant to the same extent as if Assignee were the original holder thereof.
- (3) **Investment Representations.** Assignee represents and warrants that the Warrant Securities are being acquired for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, the distribution thereof, and that Assignee has no present intention of selling, granting any participation in, or otherwise distributing the shares, nor does it have any contract, undertaking, agreement or arrangement for the same, and all representations and warranties set forth in Section 3.02 of the Investment Agreement are true and correct as to Assignee as of the date hereof.

Assignor and Assignee are signing this Assignment Form on the date first set forth above.

Assignor:	
By: Name:	
Name:	
Title:	
Assignee	
By:	
Assignee: By: Name:	
Title:	
	_

(Exhibit B – Form of Assignment)

VOTING AGREEMENT

This VOTING AGREEMENT (this "<u>Agreement</u>"), dated as of August [], 2024, is entered into by and among (i) One Planet Group LLC, a Delaware limited liability company ("<u>Purchaser</u>"), (ii) Inspirato Incorporated, a Delaware corporation (the "<u>Company</u>"), and (iii) the stockholders of the Company listed on <u>Schedule A</u> and the signature pages hereto (each, a "<u>Stockholder</u>" and, collectively, the "<u>Stockholders</u>"). Each of Purchaser, the Company and the Stockholders are sometimes referred to as a "<u>Party</u>."

RECITALS

A. Concurrently with the execution and delivery of this Agreement, Purchaser and the Company are entering into an Investment Agreement (as it may be amended, supplemented or otherwise modified from time to time, the "<u>Investment Agreement</u>") that, among other things and subject to the terms and conditions set forth therein, provides for the issuance and sale by the Company to Purchaser, and the purchase and acquisition by Purchaser from the Company, of (i) shares of the Company's Class A common stock, par value \$0.0001 per share ("<u>Class A common stock</u>"); (ii) warrants to purchase shares of Class A common stock; and (iii) a convertible note of the Company (collectively, the "<u>Purchase</u>").

B. As of the date hereof, each Stockholder is the record and/or "beneficial owner" (within the meaning of Rule 13d-3 under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) of the number of (i) shares of Class A common stock and/or (ii) shares of Class V common stock, par value, par value \$0.0001 per share, of the Company ((i) and (ii), collectively, the "Common Shares") set forth next to such Stockholder's name on <u>Schedule A</u> hereto, representing all of the Common Shares owned of record or beneficially by such Stockholder as of the date hereof (with respect to such Stockholder, the "Owned Shares", and the Owned Shares together with any additional Common Shares that such Stockholder may acquire record and/or beneficial ownership of after the date hereof (including pursuant to a stock split, reverse stock split, stock dividend or distribution or any change in Common Shares by reason of any recapitalization, reorganization, combination, reclassification, exchange of shares or similar transaction), such Stockholder's "Covered Shares").

C. In connection with the execution by Purchaser of the Investment Agreement, each Stockholder has agreed to enter into this Agreement with respect to such Stockholder's Covered Shares.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth below and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, do hereby agree as follows:

1. <u>Definitions</u>. Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in the Investment Agreement. When used in this Agreement, the following terms shall have the meanings assigned to them in this <u>Section 1</u>.

1.1. "Effective Time" shall mean the time at which the Purchase is consummated.

1.2. "Expiration Time" shall mean the earlier to occur of (a) the Effective Time, and (b) such time as the Investment Agreement is terminated in accordance with the terms of Section [__] of the Investment Agreement.

1.3. "<u>Transfer</u>" shall mean (a) any direct or indirect offer, sale, assignment, encumbrance, pledge, hypothecation, disposition, or other transfer (by operation of law or otherwise), either voluntary or involuntary, or entry into any option or other contract, arrangement or understanding with respect to any offer, sale, assignment, encumbrance, pledge, hypothecation, disposition or other transfer (by operation of law or otherwise), of any Covered Shares or any interest in any Covered Shares (in each case other than this Agreement); (b) the deposit of such Covered Shares into a voting trust, the entry into a voting agreement or arrangement (other than this Agreement) with respect to such Covered Shares or the grant of any proxy or power of attorney (other than this Agreement) with respect to such Covered Shares; or (c) any contract or commitment (whether or not in writing) to take any of the actions referred to in the foregoing clauses (a), or (b) above.

2. Agreement to Not Transfer the Covered Shares; No Inconsistent Arrangements

2.1. <u>No Transfer of Covered Shares</u>. Until the date that is 1 business day following the record date set forth in the proxy statement to be provided to the Company's stockholders in connection with the proposals described in clause (a) of Section 3.1 (the "<u>Proxy Statement</u>"), each Stockholder agrees (i) not to Transfer or cause or permit the Transfer of any Covered Shares of any Stockholder, other than with the prior written consent of Purchaser. Any Transfer or attempted Transfer of any Covered Shares in violation of this Section 2.1 shall be null and void and of no effect whatsoever.

2.2. <u>No Inconsistent Arrangements</u>. Until the Expiration Time, each Stockholder agrees not to take any action, in its capacity as the record holder or beneficial owner of Covered Shares, that would directly or indirectly have the effect of preventing, materially delaying or materially impairing such Stockholder from performing any of its obligations under this Agreement or that would, or would reasonably be expected to, have the effect of preventing, materially delaying or materially impairing, the consummation of the Purchase or the other transactions contemplated by the Investment Agreement or the performance by the Company of its obligations under the Investment Agreement.

3. Agreement to Vote the Covered Shares.

3.1. <u>Voting Agreement</u>. Until the Expiration Time, at every meeting of the Company's stockholders at which any of the following matters are to be voted on (and at every adjournment or postponement thereof), and on any action or approval of the Company's stockholders by written consent with respect to any of the following matters, each Stockholder shall vote (including via proxy) all of such Stockholder's Covered Shares (or cause the holder of record on any applicable record date to vote (including via proxy) all of such Stockholder's Covered Shares) (a) in favor of any proposal to approve the issuance of Class A common stock pursuant to the Investment Agreement, exercise of the Warrant and the conversion of the Note, to the extent such approval is required under the rules of the Nasdaq, including Nasdaq Rule 5635(b) and (d); and (b) against (1) any action or agreement that would reasonably be expected to result in any of the conditions to the Company's obligations to effect the Tranche 2 Closing set forth in the Investment Agreement not being fulfilled or result in a breach of any covenant, representation or warranty or any other obligation or agreement of the Company contained in the Investment Agreement; and (3) any merger, sale of stock or substantial assets of, or any other business combination or similar transaction involving, the Company (each, an "<u>Alternative Transaction Proposal</u>"), excluding bankruptcy transactions with a stalking horse bid and debt financing transactions in the normal course of business that are not in lieu of the Purchase (the proposals referred to in clauses (a) and (b) are collectively referred to as the "<u>Covered Proposals</u>").

3.2. <u>Quorum</u>. Until the Expiration Time, at every meeting of the Company's stockholders (and at every adjournment or postponement thereof), each Stockholder shall be represented in person or by proxy at such meeting (or cause the holders of record on any applicable record date to be represented in person or by proxy at such meeting) in order for the Covered Shares to be counted as present for purposes of establishing a quorum.

3.3. <u>Return of Proxy</u>. Each Stockholder shall execute and deliver (or cause the holders of record to execute and deliver), within 48 hours of receipt, any proxy card or voting instructions it receives that are sent to stockholders of the Company soliciting proxies with respect to any matter described in <u>Section 3.1</u>, which shall be voted in the manner described in <u>Section 3.1</u>.

4. <u>Fiduciary Duties; Legal Obligations</u>. Each Stockholder is entering into this Agreement solely in its capacity as the record holder or beneficial owner of Covered Shares. Nothing in this Agreement shall in any way limit or affect any actions taken by any Stockholder in its capacity as a director or officer of the Company or from complying with such Stockholder's fiduciary duties or other legal obligations while acting in such capacity as a director or officer of the Company.

6. <u>Representations and Warranties of the Stockholder</u>. Each Stockholder hereby represents and warrants to Purchaser and the Company that:

6.1. Due Authority. Such Stockholder has the full power and capacity to make, enter into and carry out the terms of this Agreement. If such Stockholder is not a natural person, (a) the Stockholder is duly organized, validly existing and in good standing in accordance with the Laws of its jurisdiction of formation, as applicable and (b) the execution and delivery of this Agreement, the performance of the Stockholder's obligations hereunder, and the consummation of the transactions contemplated hereby have been validly authorized, and no other consents or authorizations are required to give effect to this Agreement or the transactions contemplated by this Agreement. This Agreement has been duly and validly executed and delivered by the Stockholder and constitutes a valid and binding obligation of the Stockholder enforceable against it in accordance with its terms, except that such enforcement may be subject to (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights and remedies generally and (ii) Laws relating to the availability of specific performance, injunctive and other forms of equitable remedies, equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

6.2. <u>Ownership of the Covered Shares</u>. (a) Such Stockholder is, as of the date hereof, the beneficial or record owner of the Covered Shares set forth on <u>Schedule A</u>, free and clear of any and all liens other than those (i) created by this Agreement or (ii) arising under applicable securities laws, and (b) except as disclosed on Schedule A, such Stockholder has sole voting power over all of the Covered Shares beneficially owned by such Stockholder. Such Stockholder has not entered into any agreement to Transfer any Covered Shares. As of the date hereof, such Stockholder does not own, beneficially or of record, any Common Shares or other voting shares of the Company (or any securities convertible, exercisable or exchangeable for, or rights to purchase or acquire, any Common Shares or other voting shares of the Company) other than the Owned Shares.

6.3. No Conflict; Consents.

a. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of its obligations under this Agreement and the compliance by such Stockholder with any provisions hereof does not and will not: (a) conflict with or violate any laws, or (b) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a lien on any of the Covered Shares beneficially owned by such Stockholder pursuant to any contract or obligation to which any Stockholder is a party or by which any Stockholder is subject.

b. No consent, approval, order or authorization of, or registration, declaration or, except as required by the rules and regulations promulgated under the Exchange Act, filing with, any Governmental Entity or any other Person, is required by or with respect to the Stockholder in connection with the execution and delivery of this Agreement or the consummation by them of the transactions contemplated hereby.

6.4. <u>Absence of Litigation</u>. As of the date hereof, there is no legal action pending against, or, to the knowledge of such Stockholder, threatened against or affecting such Stockholder that would reasonably be expected to impair the ability of such Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby on a timely basis.

7. Miscellaneous.

7.1. <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given if delivered personally, sent by overnight courier or sent via email (with receipt confirmed) as follows:

a. if to a Stockholder, to the address for notice set forth on Schedule A hereto.

b. if to Purchaser, to:

One Planet Group LLC 1820 Bonanza Street Walnut Creek, CA 94596 Attention: Payam Zamani

with a copy (which shall not constitute actual constructive notice) to:

[____]

c. if to the Company, to:

Inspirato Incorporated 1544 Wazee Street Denver, CO 80202 Attention: Brent Wadman

with a copy (which shall not constitute notice) to:

Davis Graham & Stubbs LLP 1550 17th Street, Suite 500 Denver, Colorado 80202 Attn: John Elofson

or to such other address or addresses as shall be designated in writing. All notices shall be deemed effective (a) when delivered personally (with written confirmation of receipt, by other than automatic means, whether electronic or otherwise), (b) when sent, if sent by electronic mail during normal business hours of the recipient, and if not sent during normal business hours, then on the recipient's next business day or (c) one (1) Business Day following the day sent by overnight courier. "Business Day" shall mean any day, other than a Saturday, Sunday or a day on which banking institutions in the City of New York, New York are authorized or obligated by law or executive order to remain closed.

7.2. Governing Law; Jurisdiction; Waiver of Jury Trial.

a. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. In addition, each of the Parties hereto irrevocably agrees that any legal action or proceeding with respect to this Agreement and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder, or for recognition, any other state court of the State of Delaware or the United States District Court of Chancery in the State of Delaware and, if such court declines jurisdiction, any other state court of the State of Delaware or the United States District Court for the District of Delaware. Each of the Parties hereto hereby irrevocably submits with regard to any such action or proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts and agrees that it will not bring any action relating to this Agreement or any of the transactions contemplated by this Agreement in any court other than the aforesaid courts. Each of the Parties hereto are or document by U.S. registered mail to the respective addresses set forth in Section 7.1 shall be effective service of process for any suit or proceeding in connection with this Agreement or the transactions contemplated hereby.

b. EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS CONTAINED IN THIS SECTION 7.2.

7.3. <u>Publicity</u>. No press release or public announcement related to this Agreement or the transactions contemplated herein shall be issued or made by any Stockholder without the prior written approval of the Purchaser, unless required by law (based on the advice of counsel) in which case Purchaser shall have the right to review and reasonably comment on such press release, announcement or communication prior to issuance, distribution or publication. The restrictions of this Section 7.3 shall not apply to any communication if the information contained therein substantially reiterates (or is consistent with) previous press release or public announcement made in compliance with this Section 7.3. Each Stockholder consents to and authorizes the publication and disclosure by Purchaser and the Company of such Stockholder's identity and holding of the Covered Shares and the terms of this Agreement (including, for the avoidance of doubt, the disclosure of this Agreement), in any press release, the Proxy Statement and any other disclosure document required in connection with the Investment Agreement, the Purchase and the transactions contemplated by the Investment Agreement, and each Stockholder acknowledges that Purchaser and the Company may, in their sole discretion, file this Agreement or a form hereof with the SEC or any other Governmental Entity or securities exchange. Each Stockholder agrees to promptly give the Company and Purchaser any information it may reasonably require for the preparation of any such disclosure documents, and each Stockholder agrees to promptly notify the Company and Purchaser of any required corrections with respect to any information supplied by such Stockholder specifically for use in any such disclosure document, if and to the extent that any such information shall have become false or misleading in any material respect.

7.4. <u>Further Assurances</u>. Each Stockholder agrees, from time to time, at the reasonable request of Purchaser and without further consideration, to execute and deliver such additional documents and take all such further action as may be reasonably required to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated by this Agreement.

7.5. Specific Performance. The Parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, each Party agrees that in the event of any breach or threatened breach by any other Party of any covenant or obligation contained in this Agreement, the non-breaching party shall be entitled (in addition to any other remedy that may be available to it, whether in law or equity) to (i) a decree or order of specific performance to enforce the observance and performance of such covenant or obligation, and (ii) an injunction restraining such breach or threatened breach. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that any other Party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity. Any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with any such order or injunction.

7.6. Entire Agreement; Third Party Beneficiaries; Amendment. This Agreement, together with the Investment Agreement, the Warrants and the Note, sets forth the entire agreement between the parties hereto with respect to the Purchase and the Covered Proposals, and are not intended to and shall not confer upon any Person other than (i) the Parties hereto, their successors and permitted assigns any rights or remedies hereunder and (ii) One Planet Group LLC. Any provision of this Agreement may be amended or modified in whole or in part at any time by an agreement in writing between the Parties hereto and One Planet Group LLC that is executed in the same manner as this Agreement. No failure on the part of any party to exercise, and no delay in exercising, any right shall operate as a waiver thereof nor shall any single or partial exercise by any party of any right preclude any other or future exercise thereof or the exercise of any other right.

7.7. <u>Assignment</u>. This Agreement will not be assigned by any of the Parties without the prior written consent of the other Parties. Subject to the preceding sentence, but without relieving any Party of any obligation hereunder, this Agreement will be binding upon, inure to the benefit of and be enforceable by the Parties and their respective successors and assigns.

7.8. <u>Severability</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable, the remaining provisions of this Agreement shall remain in full force and effect provided that the economic and legal substance of any of the transactions contemplated hereby is not affected in any manner materially adverse to any party. In the event of any such determination, the Parties agree to negotiate in good faith to modify this Agreement to fulfill as closely as possible the original intent and purpose hereof. To the extent permitted by Law, the Parties hereby to the same extent waive any provision of law that renders any provision hereof prohibited or unenforceable in any respect.

7.14. <u>Non-Survival of Representations and Warranties</u>. None of the representations and warranties in this Agreement shall survive the Effective Time.

7.15. <u>Termination</u>. This Agreement shall automatically terminate without further action by any of the Parties and shall have no further force or effect as of the Expiration Time; <u>provided</u> that the provisions of this Article VII, to the extent relevant, shall survive any such termination. Notwithstanding the foregoing, termination of this Agreement shall not prevent any Party from seeking any remedies (at Law or in equity) against any other Party for that Party's breach of any of the terms of this Agreement prior to the date of termination.

7.16. <u>Counterparts and Signature</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed to constitute any original, but all of which together shall constitute one and the same document. Signatures to this Agreement transmitted by facsimile transmission, by electronic mail in "portable document format" (".pdf") form, or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document will have the same effect as physical delivery of the paper document bearing the original signature. The words "execution," "signed," "signature," "delivery," and words of like import in or relating to this Agreement or any document to be signed in connection with this Agreement shall be deemed to include electronic signatures, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be, and the parties hereto consent to conduct the transactions contemplated hereunder by electronic means.

(signature page follows)

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed and delivered on the date and year first above written.

PURCHASER:

ONE PLANET GROUP LLC

By: Name: Payam Zamani Title: President & CEO

(Signature Page to Voting Agreement)

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed and delivered on the date and year first above written.

COMPANY:

INSPIRATO INCORPORATED

By:			
Name:			
Title:			

(Signature Page to Voting Agreement)

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed and delivered on the date and year first above written.

STOCKHOLDER:

By:			
Name:			
Title:			
STOCI	KHOLDER:		
[NAMI	E]		

(Signature Page to Voting Agreement)

<u>Schedule A</u>

		Owned Shares*		
Name	Address	Class A Common Stock	Class V Common Stock	

*If any additional Common Shares are owned by any of the Stockholders as of the date hereof, such shares shall be automatically deemed to be "Owned Shares" notwithstanding the contents of this Schedule A.

EXECUTIVE EMPLOYMENT AGREEMENT

This Executive Employment Agreement ("*Agreement*") is made by and between **Inspirato LLC**, a Delaware limited liability company (the "*Company*"), and **Payam Zamani**, a California resident ("*Executive*"), effective as of August 13, 2024 (the "*Effective Date*").

WHEREAS, the Company desires to employ Executive upon the terms and conditions set forth herein, and Executive desires to accept employment with the Company upon such terms and conditions.

NOW, THEREFORE, for and in consideration of the above recitals and the mutual promises contained herein, and for other good and valuable consideration, the sufficiency of which is hereby acknowledged, the Company and Executive agree as follows:

1. EMPLOYMENT AND DUTIES.

- 1.1 Position. Executive shall be employed by the Company as the Chief Executive Officer and Chair of the Board of Directors of Inspirato Incorporated ("Parent"), reporting to the Parent's Board of Directors (the "Board"). In such position, the Executive shall have such duties, authority, and responsibilities of such a position and as granted by the bylaws of the Company. Executive's duties may include providing services and serving as a member of the Board, an officer, or director, without additional compensation, to any or all of the Company's direct or indirect parent, subsidiary or affiliated companies (collectively, the "Companies").
- 1.2 **Full Attention.** Executive shall devote substantial business time to affairs of the Company and the discharge of his duties and responsibilities hereunder. Provided that such activities do not, as determined by the Board in its reasonable discretion, create an actual or apparent conflict of interest, violate any provision of this Agreement or any other contract between Executive and the Company, or otherwise materially interfere with the performance of Executive's duties under this Agreement, nothing in this Agreement shall limit Executive's ability to (a) engage in any new business activity or serve in any new industry, trade, professional, governmental or academic position or during the Term or (b) participate in reasonable levels of charitable, civic, trade organization, and similar activities, passive personal investment activities, and to join as a member the board of directors of other companies.
- 1.3 **Confidentiality Agreement.** As a condition of employment, Executive shall sign and comply with all provisions of the Employee Proprietary Rights and Inventions Assignment Agreement, which includes confidentiality provisions, attached as Exhibit 1 hereto, as well as any successor agreement thereto signed by Executive regarding the same subject matter (the "*Confidentiality Agreement*"), and Executive acknowledges that his employment is adequate consideration for such agreement.

Page 1 of 19

- 2. TERM. This Agreement shall be effective on the Effective Date. Executive's employment with the Company pursuant to this Agreement shall commence on August 13, 2024 (the "*Start Date*") and shall continue until terminated as provided in Section 4 of this Agreement. The period of Executive's employment pursuant to this Agreement shall be the "*Term*."
- 3. **COMPENSATION AND BENEFITS.** During the Term, the Company shall provide the following compensation and benefits to Executive:
 - 3.1 **Base Salary.** For a period of at least one (1) year from the Effective Date, Executive shall receive an annual base salary ("*Base Salary*") of \$1 (One Dollar).
 - 3.2 **Annual Bonus.** For a period of at least one (1) year from the Effective Date, Executive shall not be eligible for a cash annual performance bonus, and rather will receive an annual equity-based performance bonus.

3.3 Equity.

- 3.3.1 One-Time Equity Grant. The Company shall, on the Start Date, cause Parent to grant Executive a one-time service-based grant of 500,000 Restricted Stock Units ("*RSUs*") of Class A common stock of Parent (the "*Parent Stock*") (the "*One-Time Equity Grant*"). The One-Time Equity Grant shall vest as follows: 25% of the One-Time Equity Grant will vest on the first anniversary of the Start Date and the remaining 75% of the One-Time Equity Grant will vest in equal quarterly installments over the subsequent three years. The One-time Equity Grant will be governed by and subject to the terms of Parent's Equity Incentive Plan and the applicable grant agreement, which will be consistent with the terms of this Section 3.3.1. For the avoidance of doubt, Executive will not pay any purchase price for the RSUs.
- 3.3.2 Performance Based Equity Grant. The Company shall, on the Start Date, cause Parent to grant Executive a performance-based equity award of 500,000 RSUs of the Parent Stock, (the "Performance Based Equity Grant"). Subject to the conditions outlined herein, the award will vest in full on the trading day after the Parent Stock achieves a closing price of \$15 per share or more over a period of at least 30 consecutive trading days during the Performance Period. The "Performance Period" is from August 14, 2024, through August 13, 2025. The performance stock price goal will be adjusted for any stock split or reverse stock split during the performance period. If the performance stock price goal is not met during the Performance Period, the Performance Based Equity Grant will be forfeited on August 14, 2025. Subject to Section 4 below, the Performance Based Equity Grant shall be subject to Executive's continuous service, whether as an employee, officer, and/or member of the Board (service in any such capacity, "Service") through the vesting date. The Performance Based Equity Grant will be subject to Parent's Equity Incentive Plan and any applicable grant agreement, which will be consistent with the terms of this Section 3.3.2. For the avoidance of doubt, Executive will not pay any purchase price for the RSUs.

Page 2 of 19

- 3.3.3 **Ongoing Equity-Based Award.** Executive will be eligible for ongoing annual equity awards under Parent's Equity Incentive Plan commensurate with his position and in accordance with the program applicable to other similarly situated executive officers for periods after August 13, 2025.
- 3.4 **Paid Time Off.** Executive shall be eligible for paid time off ("*PTO*") in accordance with the Company's PTO policies as they may exist from time to time, and as required by any state and local law. The Executive will also be entitled to all paid holidays given by the Company to its executives.
- 3.5 **Other Employment Benefits.** Executive shall be allowed to participate in the Company's other benefit plans and programs on the same basis as other executives of the Company, subject to the eligibility requirements of such plans or programs. Such benefit plans and programs may be adopted, modified or terminated by the Company from time to time in its sole discretion and may include, without limitation, medical, health and dental care, life insurance, disability protection, 401(k) and retirement plans. In addition, Executive will be afforded unlimited booking access as a guest to any of the properties of the Company at no cost to him, and the Company will provide Executive with a reimbursement for any imputed taxes that may result from such access for up to 60 nights annually.
- 3.6 **Expense Reimbursement.** The Company shall reimburse Executive for out-of-pocket expenses reasonably incurred by Executive in the performance of Executive's duties under this Agreement, subject to the Company's policies regarding expense reimbursement as they may exist from time to time.
- 4. **TERMINATION.** Executive's employment with the Company may be terminated as provided in this Section 4. This Agreement shall terminate upon the termination of Executive's employment with the Company; *provided, however*; that termination of this Agreement shall not relieve either party of obligations under this Agreement which by their terms are to be performed after termination.
 - 4.1 Termination by Company for Cause. The Company may terminate Executive's employment for Cause at any time upon written notice to Executive, effective immediately or upon such later date as may be specified in the notice. As used in this Agreement, "Cause" shall mean Executive's: (a) conviction of (including plea of guilty or no-contest to) any felony or any crime involving dishonesty; (b) material violation of law, or act of fraud in connection with Executive's employment; (c) refusal to comply with any lawful written directive of the Board that results in material harm to the Companies and that is not cured (if capable of cure, as determined by the Board in its reasonable judgment) within ten (10) days after written notice to Executive identifying the issue and what performance is expected to cure the same; (d) material breach of Executive's fiduciary duty or duty of loyalty to the Company; (e) material breach of this Agreement, the Confidentiality Agreement, or any other contract with the Company and that is not cured (if capable of cure, as determined by the Board in its reasonable judgment) within ten (10) days after written notice to Executive identifying the breach and what performance is expected to cure the same; (f) material violation of any written Company policy and that is not cured (if capable of cure, as determined by the Board in its reasonable judgment) within ten (10) days after written notice to Executive identifying the violation and what performance is expected to cure the same; or (g) conduct by the Executive constituting a willful and material act of misconduct in connection with the performance of the Executive's duties, including, without limitation (A) dishonesty to the Company with respect to any material matter; or (B) misappropriation of funds or property of the Company or any of its subsidiaries or affiliates other than the occasional, customary and de minimis use of Company property for personal purposes. Notwithstanding the foregoing, no act or omission by the Executive shall be grounds for a Cause termination if such act or omission was taken by the Executive in a reasonable belief that Executive was acting in the best interest of the Company.



- 4.2 **Termination by Company Without Cause.** The Company may terminate Executive's employment without Cause at any time upon written notice to Executive, effective immediately or upon such later date as may be specified in the notice. For all or any part of the period between the date of such notice and the effective date of such notice, the Company may, at its sole discretion, require Executive to work from home or other remote location, relieve Executive of all or any part of Executive's duties, place Executive on paid administrative leave, or any combination thereof. Any involuntary termination by the Company of the Executive's employment under this Agreement which does not constitute a termination for Cause under Section 4.1 or does not result from the Disability of the Executive under Section 4.3 will be deemed a termination without Cause.
- 4.3 Termination by Executive. Executive may terminate Executive's employment at any time for any reason, including, but not limited to, Good Reason, upon written notice to the Company, effective immediately or upon such later date as may be specified in the notice. For all or any part of the period between the date of such notice and the effective date of such notice, the Company may, at its sole discretion, require Executive to work from home or other remote location, relieve Executive of all or any part of Executive's duties, place Executive on paid administrative leave, or any combination thereof. For purposes of this Agreement, "Good Reason" will mean that the Executive has complied with the "Good Reason Process" (hereinafter defined) following the occurrence of any of the following events: (i) a material diminution in the Executive's responsibilities, authority or duties; (ii) a material diminution in the Executive's compensation and benefits as described in Section 3 of this Agreement; (iii) a material change in the geographic location at which the Executive provides services to the Company; (iv) a material breach of this Agreement or any other agreement related to Executive's employment as Chief Executive Officer between Executive and any of the Companies by any of the Companies; or (v) the failure of any acquirer of the Company, to agree to the terms of this Agreement. "Good Reason Process" will mean that (i) the Executive reasonably determines in good faith that a "Good Reason" condition has occurred; (ii) the Executive notifies the Company in writing of the first occurrence of the Good Reason condition within 60 days of his knowledge of the first occurrence of such condition; (iii) the Executive cooperates in good faith with the Company's efforts, for a period not less than 30 days following such notice (the "Cure Period"), to remedy the condition; (iv) notwithstanding such efforts, the Good Reason condition continues to exist; and (v) the Executive terminates his/her employment within 60 days after the end of the Cure Period. If the Company cures the Good Reason condition during the Cure Period, Good Reason will be deemed not to have occurred. For the avoidance of doubt, Executive may terminate his employment under this Agreement with or without Good Reason, but may maintain his role as Chairman of the Board, and/or as a Board member.

Page 4 of 19

- 4.4 Termination upon Death or Disability. Executive's employment will terminate automatically upon Executive's death; provided, however, that Executive's death shall not terminate the rights of his heirs or estate to compensation earned by Executive under this Agreement at the time of his death. The Company may terminate Executive's employment for Disability at any time upon written notice to Executive or Executive's legal representative, with immediate effect. As used in this Agreement, "Disability" means that, by reason of any medically determinable physical or mental impairment which can be expected to result in death or to last for a continuous period of not less than 12 months, Executive (a) is unable, with reasonable accommodation, to engage in any substantial gainful activity or (b) has been receiving income replacement benefits for a period of not less than 120 days under an accident and health plan covering employees of the Company. If any question arises as to whether during any period the Executive is disabled so as to be unable to perform with reasonable accommodation the essential functions of the Executive's then existing position or positions the Executive may, and at the request of the Company will, submit to the Company a certification in reasonable detail, by a physician selected by the Company trained in the disability affecting Executive to whom the Executive or the Executive's guardian has no reasonable objection, as to whether the Executive is so disabled or how long such disability is expected to continue. Such certification will for the purposes of this Agreement be conclusive of the issue. The Executive will cooperate with any reasonable request of the physician in connection with such certification. If such question will arise and the Executive unreasonably fails to submit such certification, the Company's determination of such issue will be binding on the Executive. For the avoidance of doubt, a termination by the Company for Disability shall not constitute a termination by the Company without Cause. Nothing in this Agreement will be construed to waive the Executive's rights, if any, under existing law including, without limitation, the Family and Medical Leave Act of 1993, 29 U.S.C. §2601 et seq. and the Americans with Disabilities Act, 42 U.S.C. §12101 et seq.
- 4.5 **Termination by Agreement.** Executive's employment with the Company may be terminated at any time by written agreement of the parties.

Page 5 of 19

4.6 **Payment upon Termination.**

- 4.6.1 <u>Accrued Payments</u>. Upon the termination of Executive's employment with the Company, Executive shall be entitled to payment of (a) earned but unpaid Base Salary through the date of termination, (b) any earned but unpaid Annual Bonus from a previous year, (c) reimbursement of business expenses incurred during employment in accordance with Section 3.6 of this Agreement, and (d) any benefits accrued and vested as of the date of termination in accordance with the applicable benefit plans (collectively, the "*Accrued Payments*"). Executive shall not be entitled to severance pay or other similar termination payment or benefit of any kind except as expressly provided in Section 4.6.2 of this Agreement.
- 4.6.2 <u>Severance</u>. If the Board terminates Executive's employment (actually and not constructively) without Cause, or if Executive terminates employment for Good Reason and Executive has completed at least 180 days as the Chief Executive Officer under this Agreement, in each case regardless of whether Executive continues in Service as a member of the Board, and provided that Executive (x) is in material compliance with this Agreement and the Confidentiality Agreement and (y) executes and returns to the Company a complete release of all claims against the Company and related persons in a form reasonably acceptable to the Company that becomes effective and irrevocable within sixty (60) days after the effective date of such termination (*"Termination Date"*), the Company shall, in addition to payment of the Accrued Payments:
 - (a) provide Executive with a lump sum cash severance payment equal to \$1,100,000, which shall be paid within 10 days following the effective date of the release; and
 - (b) cause all of Executive's unvested equity (including any unvested portion of the One Time Equity Grant and the Performance Based Equity Grant) to vest in full immediately as of the effective date of the release, notwithstanding the terms of the Equity Incentive Plan and the applicable award agreements. For avoidance of doubt the Performance Based Equity Grant will only vest if the performance-based targets have been achieved in the Performance Period.
- 4.6.3 <u>Change of Control Defined</u>. As used herein, "*Change of Control*" shall mean: (a) any change in the ownership or control of the common stock of the Parent which results in more than 50% of the issued and outstanding common stock of the Parent being owned or controlled by a person or entity, or a group of persons or entities, who did not own or control more than 50% of the issued and outstanding common stock of Parent as of the date of this Agreement; (b) the merger or consolidation of Parent with another entity such that more than 50% of the issued and outstanding equity interests of the surviving entity is owned or controlled by a person or entity, or a group of persons or entities, who did not own or control more than 50% of the issued and outstanding common stock of Parent as of the date of this Agreement; or (c) the sale of all or substantially all of the operating assets of Parent.

Page 6 of 19

4.6.4 <u>Resignation from Chief Executive Officer Role and Continuation as Chair</u>. For the avoidance of doubt, for purposes of any equity awards granted to Executive, including those under Sections 3.3.1, 3.3.2 and 3.3.3, Executive will be considered to be in continued employment or service as a Service Provider (as defined in the Equity Incentive Plan) and will continue to be eligible to vest in such awards so long as he remains in Service in any capacity. Without limiting the foregoing, if the Executive resigns as Chief Executive Officer of Parent but remains as Chair of the Board of or as a member of the Board, the Executive will be entitled to continue vesting in the Initial Grant and the Performance Based Equity Grant subject to rules of having achieved the performance price over the closing period per 3.3.2. If the Executive is no longer the Chief Executive Officer but still the Chair of the Board of Directors, the Executive will be eligible for an annual grant.

5. ADDITIONAL PROVISIONS

5.1 **Notices.** All notices to be given to a party hereto shall be properly given (a) on the date the notice is hand-delivered, (b) on the day after the notice is deposited with UPS or FedEx for overnight delivery to the address shown below or such other address as the party may have designated by notice to the other party, or (c) on the date received as evidence through "read receipt" via electronic mail, addressed to such party at the address shown below or such other address as the party may have designated by notice to the other party:

The Company:	Executive:
Inspirato LLC	Payam Zamani
Attn: Legal Dept	102 Muir Lane
1544 Wazee St	Alamo, CA 94507
Denver, CO 80202	payam@zamani.org
legal@inspirato.com	

- 5.2 **Severability.** The parties explicitly acknowledge and agree that the provisions of this Agreement are both reasonable and enforceable. However, the provisions of this Agreement are severable, and the invalidity of any one or more provisions shall not affect or limit the enforceability of the remaining provisions. Should any provision be held unenforceable for any reason, then such provision shall be enforced to the maximum extent permitted by law.
- 5.3 **Modification; Waiver**. Except for judicial modification as provided in Section 5.6, this Agreement cannot be amended or modified except by a writing signed by each of the parties. No waiver of any provision shall be deemed to have occurred unless memorialized in a writing signed by the waiving party. If either party should waive any breach of any provision of this Agreement, such party will not thereby be deemed to have waived any preceding or succeeding breach of the same or any other provision of this Agreement.

Page 7 of 19

- 5.4 **Compliance with Section 409A.** The intent of the parties is that payments and benefits under this Agreement comply with Section 409A of the Internal Revenue Code of 1986 ("*Section 409A*"), to the extent subject thereto, and accordingly, to the maximum extent permitted, this Agreement shall be interpreted to be in compliance with Section 409A. The payment of any annual bonus is intended to be a "short term deferral" under Section 409A and any amount payable shall be paid in a lump sum on a date determined by The Company before the end of the "short term deferral" period" with respect to such bonus. To the extent required to avoid an accelerated or additional tax under Section 409A, amounts reimbursable to Executive under this Agreement for expenses shall be paid to Executive on or before the last day of the year following the year in which the expense was incurred and the amount of expenses eligible for reimbursement during one year may not affect the amounts reimbursable in any subsequent year. Notwithstanding any other provision in this Agreement or in any other document, The Company shall not be responsible for the payment of any applicable taxes incurred by Executive pursuant to this Agreement, including with respect to compliance pursuant to Section 409A. The Company makes no representation that any or all of the payments and benefits described in this Agreement will be exempt from or comply with Section 409A.
- 5.5 **Governing Law; Venue.** This Agreement shall be governed by the laws of the State of Colorado and applicable federal law, without regard to any state's principles regarding conflict of laws. Any action arising out of or relating to this Agreement shall be brought only in the state or federal courts in or for Denver, Colorado, and Executive and the Company hereby irrevocably waive any right that they might have to challenge the selection of those forums, including but not limited to challenges to personal jurisdiction, venue, or the convenience of the forum.
- 5.6 **Attorney Fees.** In the event of a breach or threatened breach of this Agreement, the non-breaching party shall be entitled to recover such party's reasonable attorney fees incurred as a result of such breach or threatened breach.
- 5.7 **Binding Effect; Assignment.** This Agreement shall be enforceable by the Company and its successors and assigns and shall be binding against Executive and Executive's heirs, beneficiaries and legal representatives. The Company may assign this Agreement to any parent, subsidiary or affiliated company or successor in interest. Executive may not assign this Agreement.
- 5.8 **Taxes.** All payments pursuant to this Agreement shall be subject to withholding for taxes as required by applicable law.
- 5.9 **Construction.** This Agreement shall be deemed to have been drafted jointly by the parties, and no ambiguity in the Agreement shall be construed against either the Company or Executive.
- 5.10 **Titles and Headings.** Titles and headings in this Agreement are for purpose of reference only and shall not limit, define or otherwise affect the provisions of this Agreement.

Page 8 of 19

5.11 **Complete Agreement.** This Agreement (along with the Confidentiality Agreement) is the entire agreement between the parties regarding the matters addressed herein, and it and supersedes and replaces all prior agreements, representations, negotiations or discussions between the parties regarding such matters, whether written or oral. This Agreement may be signed in counterparts, including fax counterparts, and all counterparts together shall constitute one fully-executed agreement.

WHEREUPON, the parties have executed this Agreement on the dates shown below, to be effective as of the Effective Date.

EXECUTIVE:

THE COMPANY: INSPIRATO LLC

By:

Robert Kaiden, Chief Financial Officer

Date:

Page 10 of 19

Date:

Payam Zamani By:

EXHIBIT 1

EMPLOYEE PROPRIETARY RIGHTS AND INVENTIONS ASSIGNMENT AGREEMENT

This Employee Proprietary Rights and Inventions Assignment Agreement ("Agreement") is made and entered into by and between Inspirato LLC (the "Company"), and the employee whose name and signature appear below ("Employee") as of the date of Employee's signature below:

WHEREAS, Employee is employed or has been offered employment with the Company to provide services to the Company and/or to one or more of the other Companies (as defined below) in a position in which Employee will or may have access to the Companies' intellectual property, trade secrets and other confidential information;

WHEREAS, the Companies' intellectual property, trade secrets and other confidential information are valuable assets of the Companies; and

WHEREAS, the Company, on behalf of itself and the other Companies, hereby acknowledges and understands that the Employee owns and operates other businesses, is a co-inventor of patents, and continues to innovate and build things as part of these other businesses he owns and operates.

NOW, THEREFORE, in consideration of the employment described above and other good and valuable consideration, the parties agree as follows:

1. CONFIDENTIALITY.

1.1 Definition of "Affiliate." As used in this Agreement, "**Affiliate"** means a legal entity that (a) owns or controls another legal entity, (b) is owned or controlled by one or more other legal entities or natural persons, or (c) is under common ownership or control with another legal entity. As used in this definition, "control" (including its correlative meanings, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise).

1.2 Definition of the "Companies." As used in this Agreement, the "Companies" means the Company and all of its direct and indirect Affiliates.

1.3 Definition of "Confidential Information." As used in this Agreement, "**Confidential Information**" means all non-public intellectual property of the Companies, all of the Companies' "**trade secrets**" as defined in the Colorado Uniform Trade Secrets Act (C.R.S §7-74-101 et seq.), and all other non-public information of the Companies relating to the business of the Companies, including but not limited to all non-public information regarding the Companies': organization, operations, and management; revenues, expenses and finances; existing and prospective contracts and business arrangements; policies and procedures; employees and contractors, including payroll, medical and other personnel records; customers, including customer lists and customer needs and preferences; vendors and service providers; business and marketing plans and strategies; data and datasets; software, hardware and information systems; and patents, trademarks and other intellectual property, in each case, obtained by Employee during the period that Employee is employed with the Company. Confidential Information does not include information that (i) arises from Employee's general training, knowledge, skill, or experience, whether gained from any Companies or not; (ii) is readily ascertainable to the public; (iii) Employee otherwise has a right to disclose as legally protected; (iv) has become generally publicly known through no fault of Employee; (v) was previously known by Executive before his employment by the Company; (vi) was given by any Companies to any third-party without the requirement of confidentiality; or (vii) is or was developed independently without any breach of this Agreement.

Page 11 of 19

1.4 Confidentiality and Nondisclosure. Employee hereby acknowledges and agrees that all Confidential Information which Employee receives or learns while employed by the Company shall be considered the exclusive property of the Company. Without the written consent of the Company, Employee shall not, directly or indirectly, disclose or use any Confidential Information for the benefit of any person other than the Companies. The obligations set forth in this paragraph are in addition to, and not in lieu of, any obligations of Employee otherwise provided by applicable law, such as trade secret statutes, fiduciary duties, and the like. Notwithstanding the foregoing or anything in this Agreement to the contrary, Employee may disclose Confidential Information to the extent required under applicable law or pursuant to any order or requirement of a court, administrative agency or other governmental body.

1.5 Return of Company Property. Employee agrees that upon termination of Employee's employment with the Company, for whatever reason and whether voluntary or involuntary, or at any time upon the Company's reasonable written request, Employee will promptly surrender to the Company all property of the Companies in Employee's possession, custody or control, including but not limited to any copies of materials that incorporate or are derived from Confidential Information, and certify in writing to the Company that Employee has done so.

1.6 Compliance with Pre-Existing Duties. Employee represents and warrants that Employee's employment with the Company does not and will not breach any agreement with any former employer of Employee, including any confidentiality agreement or noncompetition agreement with a former employer. Employee shall not, during his/her employment with the Company, improperly use or disclose to any of the Companies any proprietary information or trade secrets belonging to any former employer or any other third party to whom Employee owes a duty of nondisclosure.

1.7 Information from Third Parties. Employee acknowledges that the Companies have received and will continue to receive confidential or proprietary information from third parties which the Companies must maintain in confidence and protect from unauthorized disclosure or use, including but not limited to information regarding resort and other properties and the Companies' members and their families and guests. Without the written consent of the Company, Employee shall not, directly or indirectly, disclose or use for the benefit of any person other than the Companies any such information, except where such disclosure or use is: (a) in connection with and in furtherance of Employee's work on behalf of the Companies, (b) not otherwise contrary to applicable laws regarding trade secrets, confidential information or intellectual property; and (c) not contrary to any agreement between the third party and any of the Companies of which Employee has knowledge.¹

2. DEVELOPMENTS.

2.1 Developments. As used herein "Development" means all products of human intelligence which have been protected or could be protected by Intellectual Property Rights (as defined hereafter), all embodiments thereof (including, without limitation, all software, hardware, information, data, documentation, materials, ideas, discoveries, concepts, processes, formulae, techniques, designs, formats, methodologies, algorithms, programs, know-how, tools, and other technology), all inventions, conceptions, developments, discoveries, creations, or works of authorship included therein, and all updates, upgrades, enhancements, modifications, derivatives, improvements and translations thereto, thereof or thereon, and all Intellectual Property Rights therein and relating thereto.

2.2 Intellectual Property Rights. As used herein, the term "Intellectual Property Rights" means all worldwide intellectual property and proprietary rights, including, without limitation, all trade secrets, patents and patent applications, copyrights, mask works, trademarks, trade names, service marks, trade dress, moral rights, rights in datasets and databases, contractual rights, and all other intellectual property and proprietary rights recognized by the laws of any jurisdiction or country, whether registered or unregistered.

¹ Note to Draft: Given the significant overlap in Executive's business activities, it is expected that he may bring on employees from other ventures to work with the Company. For this reason a nonsolicit is not feasible here.

Page 12 of 19

2.3 Company Developments. As used herein, the term "Company Developments" means all Developments exclusively relating to the business of the Companies made, conceived, reduced to practice, created, or developed by Employee in whole or in part while employed by the Company and solely in the course of the performance by Employee of his duties for the Company; provided, however, that, notwithstanding the foregoing or anything in this Agreement to the contrary, Company Developments shall not include any and all (a) Prior Developments as defined in Section 2.6 (Prior Developments) of this Agreement, or (b) Developments that are (i) not exclusively related to the business of the Companies, (ii) made, conceived, reduced to practice, created or developed, alone or jointly with others, wholly on Employee's own time and without use of personnel or resources of any of the Companies, or (iii) made, conceived, reduced to practice, created or developed, alone or jointly with others, wholly on Employee's own time and without use of personnel or resources of any of the Companies, or (iii) made, conceived, reduced to practice, created or developed, alone or jointly with others, as part of the Employee's other businesses owned or operated by the Employee (subsections (a) through (b), and any and all Intellectual Property Rights in or to any such Developments, hereinafter referred to as "Company Excluded Developments"). The parties hereby acknowledge and agree that, notwithstanding anything in this Agreement to the contrary, (A) as between Employee, on the one hand, and the Companies, on the other hand, the Company Excluded Developments are and shall be the sole and exclusive property of Employee, (B) Employee hereby retains and reserves any and all rights, titles, and interests in and to the Company Excluded Developments, and (C) the Companies shall have no rights to such Company Excluded Developments.

2.4 Assignment of Company Developments. All Company Developments shall be the sole and exclusive property of the Company. Employee agrees to and hereby does irrevocably assign, now and in the future (when any such Company Developments or any Intellectual Property Rights therein or related thereto are first made, conceived, reduced to practice, created, or developed, as applicable), to the Company (or to a third party if directed by the Company) all of Employee's right, title, and interest in and to any and all Company Developments (and all Intellectual Property Rights therein and related thereto) made, conceived, reduced to practice, created, or developed by Employee (either alone or with others) while employed by the Company. Following termination of Employee's employment with the Company, Employee retains no rights to use the Company Developments or any such Intellectual Property Rights.

2.5 Obligation to Keep Company Informed. During the period of Employee's employment, Employee will promptly and fully disclose to the Company in writing all Company Developments authored, conceived, or reduced to practice by Employee, either alone or with others. For the avoidance of doubt, in no event shall the Employee be required to disclose any Company Excluded Developments to any Company hereunder.

2.6 Prior Developments. Employee agrees that Employee will not incorporate, or permit to be incorporated, Prior Developments (defined below) in any Company Developments without the Company's prior written consent. In addition, Employee agrees that Employee will not incorporate into any Company software or otherwise deliver to the Company any software code licensed under any open source software license (including the GNU GPL or LGPL or any other license that, by its terms, requires or conditions the use or distribution of such code on the disclosure, licensing, or distribution of any source code owned or licensed by the Company). Employee has, or has caused to be, alone or jointly with others, made, conceived, reduced to practice, created or developed certain Developments prior to the commencement of Employee's employment by the Company, in which Employee has an ownership interest or which Employee has a license to use, and that Employee wishes to have excluded from the scope of this Agreement (collectively referred to as "**Prior Developments**"). If, in the course of Employee's employment with the Company, Employee hereby grants the Company a non-exclusive, perpetual, fully-paid and royalty-free, non-transferable, irrevocable and worldwide license, with rights to sublicense through multiple levels of sublicensees, to reproduce, make derivative works of, distribute, publicly perform, and publicly display in any form or medium, whether now known or later developed, make, have made, use, sell, import, offer for sale such Prior Development, but solely to the extent necessary for the operation of the Company's business.

Page 13 of 19

2.7 Protection and Enforcement of Intellectual Property Rights and Assistance. The Company will have the right, at its own expense, and solely in its own name, to apply for, prosecute and defend its rights in the Company Developments and all Intellectual Property Rights therein. During the period of Employee's employment and thereafter, upon the Company's reasonable written request and at the Company's sole cost and expense, Employee will execute such documents and take such other actions as the Company may reasonably request in order to secure the Company's ownership of the United States and foreign Intellectual Property Rights relating to Company Developments in all countries.

2.8 Records. Employee agrees to keep and maintain adequate and current records (in the form of notes, sketches, drawings and in any other form that is required by the Company) of all Company Developments made by Employee during the period of Employee's employment by the Company, which records shall be available to, and remain the sole property of, the Company at all times.

2.9 Special Notice to Employee. This Section 2 does not require Employee to assign any of his or her rights in any Development which qualifies fully for protection under California Labor Code §2870; Chapter 19 Section 805 of the Delaware Code; Chapter 765 Section 1060/2 of the Illinois Compiled Statutes; Kansas Statutes Annotated Section 44-130; Minnesota Statutes Annotated Section 181.78; North Carolina General Statutes Section 66-57.1; Revised Code of Washington Section 49.44.140; or any other equivalent law. You acknowledge that you have received and read the state law notices in Appendix A to this Agreement.

3. PROTECTED RIGHTS.

Notwithstanding any other provision of this Agreement, nothing in this Agreement (or any other agreement signed by Employee) shall restrict Employee's right to (a) report violations of law to law enforcement officials; (b) give truthful testimony under oath in a judicial, administrative, or arbitral proceeding; (c) file a charge with, make truthful statements to, cooperate with investigations by, or assist others in proceedings before governmental agencies (including the U.S Equal Employment Opportunity Commission, the National Labor Relations Board and the U.S Securities and Exchange Commission); (d) speak with an attorney representing Employee; (e) discuss the facts related to any claim of sexual assault or sexual harassment; (f) engage in whistle-blower activity protected by the Securities Exchange Act of 1934, the Dodd-Frank Wall Street Reform and Consumer Protection Act, or any rules or regulations issued thereunder (including Rule 21F-17); or (g) file or disclose any facts necessary to receive unemployment insurance, Mediciad, or other public benefits to which Employee may be entitled. In addition, 18 U.S.C. §1833(b) provides as follows, and nothing in this Agreement or any other agreement, or any Company policy, is intended to conflict with this statutory protection: "(1) An individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. (2) An individual and use the trade secret information in the court proceeding, if the individual (A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret, except pursuant to court order."

<u>State-Specific Rights</u>: If Employee is employed in a state listed on <u>Appendix B</u>, then the applicable provisions on <u>Appendix B</u> shall apply in lieu of any conflicting provision in this Agreement.

Page 14 of 19

4. NONDISPARAGEMENT. While employed by the Company, Employee shall not intentionally publicly disclose, issue any public statements, or otherwise cause to be publicly disclosed any information which is designed, intended or might reasonable be anticipated to disparage the Company, its officers or directors, its business, its products or its customers, in their capacity as such. There is no disparagement under this Section 4 unless, at the time of the disclosure or statement complained of, Executive actually knows the affected party (or parties), personnel, product(s) or customer(s) in question are officers, directors, businesses, products, or customers of the Company. Notwithstanding the foregoing, nothing in this Agreement shall prevent, restrict, prohibit, limit or preclude Employee from: (i) reporting violations of law to law enforcement officials; (ii) giving truthful testimony under oath in a judicial, administrative, or arbitral proceeding; or (iii) making truthful statements to governmental agencies such as the EEOC or SEC.

5. SCOPE AND REMEDIES. Employee acknowledges and agrees that any breach by Employee of any provision of this Agreement will cause the Company irreparable injury and damage and that the Company shall therefore be entitled to, in addition to all other remedies available to it, injunctive and other equitable relief (without the necessity of posting a bond) to prevent or stop such breach and to secure the enforcement of this Agreement.

6. GENERAL

6.1 Severability. If any provision of this Agreement is found to be invalid or unenforceable, the remaining provisions will be unaffected and will remain enforceable according to their terms.

6.2 Modification; Waiver. Except for judicial modification as expressly provided herein, this Agreement cannot be amended or modified except by a writing signed by each of the parties. No waiver of any provision shall be deemed to have occurred unless memorialized in a writing signed by the waiving party. If either party should waive any breach of any provision of this Agreement, such party will not thereby be deemed to have waived any preceding or succeeding breach of the same or any other provision of this Agreement.

6.3 No Effect on At-Will Employment Status. This Agreement is not intended to, and shall not be construed to, grant any employment rights to Employee beyond that of at-will employment or, if applicable, as outlined in Employee's employment agreement as executed by the Employee and a corporate officer of the Company.

6.4 Survival. The provisions of this Agreement shall survive the termination of this Agreement and the termination of Employee's employment with the Company.

6.5 Governing Law and Venue. This Agreement shall be governed by the laws of the State of Colorado and applicable federal law, without regard to any state's principles regarding conflict of laws. Any action arising out of or relating to this Agreement shall be brought only in the state or federal courts in or for Denver, Colorado, and Employee and the Company hereby irrevocably waive any right that they might have to challenge the selection of those forums, including but not limited to challenges to personal jurisdiction, venue, or the convenience of the forum.

6.6 Binding Effect; Assignment. This Agreement shall be enforceable by and binding against the Company and its successors and assigns and shall be binding against Employee and Employee's heirs, beneficiaries and legal representatives. The Company may assign this Agreement to any parent, subsidiary or affiliated company or successor in interest. Employee may not assign this Agreement.

6.7 Titles and Headings. Titles and headings in this Agreement are for purpose of reference only and shall not limit, define or otherwise affect the provisions of this Agreement.

6.8 Complete Agreement. This Agreement, together with Employee's employment agreement as executed by the Employee and a corporate officer of the Company, is the entire agreement between the parties regarding the matters addressed herein, and it and supersedes and replaces all prior agreements, representations, negotiations or discussions between the parties regarding such matters, whether written or oral. This Agreement may be signed in counterparts, including fax counterparts, and all counterparts together shall constitute one fully-executed agreement.

Page 15 of 19

EMPLOYEE:	INSPIRATO, LLC
	By:
Signature	Signature
Print Name	Print Name, Title
Date	Date
Page	16 of 19

Appendix A

California Labor Code Section 2870

(a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer s equipment, supplies, facilities, or trade secret information except for those inventions that either: (1) Relate at the time of conception or reduction to practice of the invention to the employer s business, or actual or demonstrably anticipated research or development of the employer; or (2) Result from any work performed by the employee for the employer.

(b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

Chapter 19 Section 805 of the Delaware Code

Any provision in an employment agreement which provides that the employee shall assign or offer to assign any of the employee's rights in an invention to the employee's employer shall not apply to an invention that the employee developed entirely on the employee's own time without using the employer's equipment, supplies, facility or trade secret information, except for those inventions that: (1) Relate to the employer's business or actual or demonstrably anticipated research or development; or (2) Result from any work performed by the employee for the employer.

To the extent a provision in an employment agreement purports to apply to the type of invention described, it is against the public policy of this State and is unenforceable. An employer may not require a provision of an employment agreement made unenforceable under this section as a condition of employment or continued employment.

Chapter 765 Section 1060/2 of the Illinois Compiled Statutes

(1) A provision in an employment agreement which provides that an employee shall assign or offer to assign any of the employee's rights in an invention to the employer does not apply to an invention for which no equipment, supplies, facilities, or trade secret information of the employer was used and which was developed entirely on the employee's own time, unless (a) the invention relates (i) to the business of the employer, or (ii) to the employer's actual or demonstrably anticipated research or development, or (b) the invention results from any work performed by the employee for the employer.

Kansas Statutes Annotated Section 44-130

Any provision in an employment agreement which provides that an employee shall assign or offer to assign any of the employee's rights in an invention to the employer shall not apply to an invention for which no equipment, supplies, facilities or trade secret information of the employer was used and which was developed entirely on the employee's own time, unless:(1) The invention relates to the business of the employer or to the employer's actual or demonstrably anticipated research or development; or (2) the invention results from any work performed by the employee for the employer.

Minnesota Statutes Annotated Section 181.78

Any provision in an employment agreement which provides that an employee shall assign or offer to assign any of the employee's rights in an invention to the employer shall not apply to an invention for which no equipment, supplies, facility or trade secret information of the employer was used and which was developed entirely on the employee's own time, and (1) which does not relate (a) directly to the business of the employee or (b) to the employer's actual or demonstrably anticipated research or development, or (2) which does not result from any work performed by the employee for the employer. Any provision which purports to apply to such an invention is to that extent against the public policy of this state and is to that extent void and unenforceable.

North Carolina General Statutes Section 66-57.1

Any provision in an employment agreement which provides that the employee shall assign or offer to assign any of his rights in an invention to his employer shall not apply to an invention that the employee developed entirely on his own time without using the employer's equipment, supplies, facility or trade secret information except for those inventions that (i) relate to the employer's business or actual or demonstrably anticipated research or development, or (ii) result from any work performed by the employee for the employer. To the extent a provision in an employment agreement purports to apply to the type of invention described, it is against the public policy of this State and is unenforceable. The employee shall bear the burden of proof in establishing that his invention qualifies under this section.

Revised Code of Washington Section 49.44.140

A provision in an employment agreement which provides that an employee shall assign or offer to assign any of the employee's rights in an invention to the employer does not apply to an invention for which no equipment, supplies, facilities, or trade secret information of the employer was used and which was developed entirely on the employee's own time, unless (a) the invention relates (i) directly to the business of the employer, or (ii) to the employer's actual or demonstrably anticipated research or development, or (b) the invention results from any work performed by the employee for the employer. Any provision which purports to apply to such an invention is to that extent against the public policy of this state and is to that extent void and unenforceable.

Page 18 of 19

APPENDIX B:

State-Specific Rights

For employees who are employed in one of the following states, the applicable provisions for that state apply in lieu of any conflicting provision in the Agreement.

California	Nothing in this Agreement prevents Employee from discussing or disclosing information about unlawful acts in the workplace, such as harassment or discrimination or any other conduct that Employee has reason to believe is unlawful.
	The California Uniform Trade Secrets Act (Cal. Civil Code §§3426-3426.11) shall govern instead of the Delaware Uniform Trade Secrets Act (6 Del. Code §2001 et seq.).
	The jury waiver in Section 6.5 shall not apply.
Illinois	Nothing in this Agreement prevents Employee from making truthful statements or disclosures regarding unlawful employment practices.
Maine	Nothing in this Agreement prevents Employee from disclosing or discussing discrimination, including harassment, occurring between employees or between an employer and an employee (a) in the workplace, (b) at work-related events coordinated by or through the employer, or (c) off the employment premises.
	Employee retains the right to provide testimony or evidence, file claims or make reports to any federal or state agency that enforces employment or discrimination laws, including, but not limited to, the Maine Human Rights Commission and the Department of Labor.
New Mexico	Nothing in this Agreement prevents Employee from disclosing a claim of sexual harassment, discrimination or retaliation occurring in the workplace or at a work-related event coordinated by or though the Company.
Oregon	Nothing in this Agreement prevents Employee from disclosing or discussing conduct (a) that constitutes discrimination prohibited by ORS 659A.030 (Discrimination because of race, color, religion, sex, sexual orientation, gender identity, national origin, marital status, age or expunged juvenile record prohibited), including conduct that constitutes sexual assault, or by ORS 659A.082 (Discrimination against person for service in uniformed service prohibited) or 659A.112 (Employment discrimination), and (b) that occurred between employees or between an employer and an employee in the workplace or at a work-related event that is off the employment premises and coordinated by or through the employer, or between an employer and an employee off the employment premises ("Discrimination").
Washington	Nothing in this Agreement (or any other agreement signed by Employee) prevents Employee from disclosing or discussing conduct, or the existence of a settlement involving conduct, that Employee reasonably believed under Washington state, federal, or common law to be illegal discrimination, illegal harassment, illegal retaliation, a wage and hour violation, or sexual assault, or that is recognized as against a clear mandate of public policy.

Page 19 of 19

Separation and Release Agreement

This Separation and Release Agreement ("<u>Agreement</u>") is made between Inspirato LLC, a Delaware limited liability company (the "<u>Company</u>") and Bradley Handler ("<u>Executive</u>") (each a "party" and together the "parties"):

WHEREAS, the Executive was terminated without cause by the Company as Executive Chairman of Inspirato Incorporated, a Delaware corporation that is the parent company of Inspirato LLC;

WHEREAS, the Executive has separated from employment with the Company effective August 13, 2024 ("Separation Date");

WHEREAS, the parties wish to ensure an amicable separation and to provide for the release in full of all claims by the Executive;

WHEREAS, the parties mutually agree that Executive's Employment Agreement signed on June 30, 2021 (the "<u>Employment Agreement</u>") is terminated as of the Separation Date and that this Agreement supersedes and replaces the Employment Agreement, including but not limited to any provisions addressing severance pay and benefits upon termination;

NOW, THEREFORE, the parties agree as follows:

1. <u>Separation Benefits</u>. Provided that the Executive complies with all conditions described in Section 3 of this Agreement (the "<u>Conditions</u>"), the Company shall provide the following separation benefits to the Executive:

(a) **Severance Pay**. The Company will pay Executive severance pay in the gross amount of \$216,000.00 (Two Hundred Sixteen Thousand Dollars and 00/100), less applicable local, state, and federal tax withholdings ("<u>Severance Pay</u>"). Severance Pay payments will be deposited directly into Executive's bank account(s) as Executive has designated in the Company's payroll service. Severance Pay will be paid in thirty-six (36) substantially equal bimonthly installments, each in the gross amount of \$6,000 (Six Thousand Dollars and 00/100), in accordance with the Company's payroll practice for Colorado employees ("<u>Severance Pay Installments</u>"). The first Severance Pay Installment shall be paid on January 15, 2025 and shall continue for 35 payroll cycles thereafter, subject to all conditions herein. Severance Pay will not be subject to voluntary employee deferral or employer matching contributions pursuant to any pension or other retirement plan. Each Severance Pay Installment pursuant to this section is intended to constitute a separate payment for purposes of Treasury Regulation Section 1.409A-2(b)(2).

(b) **COBRA**. If the Executive timely elects and is eligible for Continuation Coverage as defined herein, the Company shall pay directly to the Company's COBRA provider or group health plan provider for premiums to continue the medical, dental and vision insurance coverage (if any) of Executive and Executive's eligible dependents pursuant to the continuation- coverage provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 or comparable state law ("<u>Continuation Coverage</u>") from the Separation Date through the earlier of (i) the eighteenth (18) month anniversary of the Separation Date; (ii) the Executive's eligibility for group medical plan benefits under any other employer's group medical plan; or (iii) the cessation of the Executive's continuation rights under COBRA. Provided, however, if the Company determines that it cannot pay such amounts to the group health plan provider or the COBRA provider (if applicable) without potentially violating applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then the Company shall convert such payments to payroll payments directly to the Executive for the time period specified above. Such payments shall be subject to tax-related deductions and withholdings and paid on the Company's regular payroll dates.

Page 1 of 8

(c) **Equity Securities.** Notwithstanding the terms of Inspirato Incorporated's 2021 Equity Incentive Plan or any applicable award agreements, 105,441 (one hundred and five thousand four hundred forty-one) of Executive's unvested Restricted Stock Units shall become fully vested on the eighth (8) calendar day after Executive signs and returns the signed original of this Agreement to the Company.

(d) **Taxes**. The Company may withhold taxes or report taxable income from benefits provided pursuant to this Section 1. The Company makes no representation concerning tax consequences or tax liability that may be incurred by Employee from payments made pursuant to this Agreement. Employee and the Company acknowledge that nothing herein constitutes tax advice to the other party.

2. <u>Release in Full of All Claims</u>. In exchange for the promises described in Section 1 of this Agreement, the Executive, for himself/herself and his/her heirs, assigns and personal representatives, fully and completely releases the Company and its parents, subsidiaries and affiliated entities and all predecessors and successors thereto, and all benefit plans thereof, and all of their respective shareholders, members, partners, directors, officers, managers, employees, attorneys, administrators and agents (each a "<u>Releasee</u>" and collectively the "<u>Releasees</u>") from any and all claims or causes of action that the Executive may have against the Releasees, known or unknown, including claims or causes of action that relate in any way to the Executive's employment with any Release or the termination thereof, from the beginning of time through the date the Executive signs this Agreement (each a "<u>Released Claims</u>"), including but not limited to the following:

(a) federal, state or local laws prohibiting discrimination (including harassment and retaliation) in employment, such as: (i) the Age Discrimination in Employment Act ("<u>ADEA</u>"), the Older Workers Benefit Protection Act, and Executive Order 11141, which prohibit discrimination based on age; (ii) Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1866 (42 U.S.C. § 1981), the Equal Pay Act, and Executive Order 11246, which prohibit discrimination based on race, color, national origin, religion, or sex; (iii) the Genetic Information Nondiscrimination Act, which prohibits discrimination based on disability; (v) the Americans With Disabilities Act and Sections 503 and 504 of the Rehabilitation Act of 1973, which prohibit discrimination based on disability; (v) the National Labor Relations Act, which prohibits discrimination for engaging in certain concerted protected activity; (vii) the Occupational Safety and Health Act and the Mine Safety and Health Act, which prohibit discrimination for engaging in certain safety-related activity; (vii) the Sarbanes Oxley Act, which prohibits discrimination for engaging in certain whistleblowing activity; and (viii) any state or local law that prohibits discrimination on any of the bases described above;

(b) federal, state or local laws regarding wages and hours, including laws regarding minimum wage, overtime compensation, wage payment, vacation pay, sick pay, compensatory time, commissions, bonuses, and meal and break periods wages, such as the Fair Labor Standards Act and the Colorado Wage Claim Act (C.R.S. 8-4-101 et seq.);

(c) other employment laws, including but not limited to: (i) the Family and Medical Leave Act and analogous state and local laws, which require employers to provide leaves of absence under certain circumstances; (ii) the Worker Adjustment and Retraining Notification Act (WARN) and analogous state and local laws, which require advance notice of certain workforce reductions; (iii) the Employee Retirement Income Security Act, which protects employee benefits (among other things); and (iv) the Uniformed Services Employment and Reemployment Rights Act, which requires employers to provide military leave under certain circumstances; and

Page 2 of 8

(d) any common law theory, including but not limited to breach of contract (expressed or implied), promissory estoppel, wrongful discharge, outrageous conduct, defamation, fraud or misrepresentation, tortious interference, invasion of privacy, negligent hiring or supervision, or any other claims based in contract, tort or equity.

Notwithstanding the foregoing, the Released Claims do not include claims for breach of this Agreement, claims related to rights to indemnification or insurance the Executive has pursuant to contractual arrangements with the Company or its parent, corporate documents of the Company or its parent, claims that arise after the Executive signs this Agreement, claims for vested pension benefits, claims for workers' compensation benefits or unemployment compensation benefits, and any other claims that cannot by law be released by private agreement. In addition, this release does not prevent the Executive from filing: (i) a lawsuit to challenge the effectiveness of a release of claims of age discrimination under the ADEA; or (ii) a charge with a governmental agency, including but not limited to the U.S. Equal Employment Opportunity Commission ("EEOC") and the U.S. Securities and Exchange Commission ("SEC"), but the Executive is waiving his/her right to recover any monetary or injunctive relief pursuant to any such charge. Notwithstanding the foregoing, this Agreement does not prevent the Executive from recovering an award from or by a governmental agency for providing information.

Executive acknowledges and agrees that the Executive is releasing both known and unknown claims and waives the benefit of any statute purporting to prevent the Executive from releasing unknown claims, including but not limited to the protection of Cal. Civ. Code Section 1542, which states:

A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party.

3. <u>Conditions</u>. The Executive shall comply with the following terms, as conditions of payment, and in the event that the Executive fails to satisfy these conditions, the Company shall have no obligation to provide any separation benefits pursuant to Section 1 and shall be entitled to a refund of any separation benefits previously paid:

(a) *Execution and Return of Agreement.* The Executive shall sign this Agreement and return the signed original of the Agreement to the Company within forty-five (45) days after the Executive receives the Agreement and shall not revoke it, except as otherwise described in Section 6, below.

(b) *Property*. Excluding the two computers and associated monitors at the Executive's home, the Executive shall return all Company property in the Executive's possession, custody or control on or before the Separation Date, including but not limited to all motor vehicles, computer hardware, office equipment, tools, telephones, credit cards, keys, card keys, and the originals and all copies of all documents, files, computer software and electronic data (including any and all copies of the Company's proprietary software, commensurate source code and engineering designs); provided, however, that the Executive may retain copies of documents reflecting the Executive's compensation and benefits from the Company. By signing this Agreement, the Executive represents and warrants that the Executive has fully complied with this Section 3(b).

(c) *Non-Disparagement*. Subject to Section 8, to the fullest extent permitted by law, neither the Executive nor the Company shall engage or in any conduct or make any statement calculated to or likely to have the effect of undermining, disparaging or maligning the Executive, the Releasees or the Company's business, products, services, customers or clients. By signing this Agreement, both the Executive and the Company represents and warrants that neither the Executive or the Company have made statements on or after the Separation Date that would violate this Section, if made after this Agreement has become effective. By signing the General Release, the Executive and the Company represent and warrant they have fully complied with this Section 3(c).

Page 3 of 8

(d) Confidential Information.

(i) Subject to Section 8, Executive shall not disclose to any third party, or use for the benefit of the Executive or any third party, any Confidential Information. For purposes of this Agreement, "<u>Confidential Information</u>" shall mean: (A) all trade secrets of the Releasees, as that term is defined in the Colorado Uniform Trade Secrets Act, C.R.S. 7-74-101 et seq.; (B) all intellectual property of the Releasees, including but not limited to all inventions, discoveries, ideas or processes that have been or could be protected by patent, trademark, copyright or similar protections; (C) all communications or information to or from counsel for any of the Releasees that constitute attorney work product or are protected by attorney-client privilege; and (D) all other non-public information concerning the business or operations of the Releasees, including but not limited to information concerning organization, management, finances, business plans and strategies, clients and customers, relationships with contractors and vendors, proprietary or specialized computer software, employees, products and services, equipment and systems, methods, processes and techniques, and prospective and executed contracts and other business arrangements.

(ii) In response to any subpoena, court order or other legal process purporting to require disclosure of Confidential Information, the Executive shall: (A) immediately notify the Company; (B) take all lawful steps, at the Company's expense, to resist the subpoena, court order or other process unless otherwise directed by the Company; and (C) cooperate fully, at the Company's expense, with all lawful efforts by the Company to protect the Confidential Information from disclosure.

(iii) Notwithstanding the foregoing, 18 U.S.C. §1833(b) provides, in part: "(1) An individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. (2) An individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual and use the trade secret information in the court proceeding, if the individual (A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret, except pursuant to court order." Nothing in this Agreement, any other agreement executed by the Executive, or any Company policy is intended to conflict with this statutory protection.

(e) Other Agreements. Subject to Section 8, both the Executive and the Company shall fully comply with all other agreements between the Executive and the Company (or any parent, subsidiary or affiliate of the Company or predecessor or successor thereto), including but not limited to: (i) the Proprietary Rights and Inventions Assignment Agreement attached hereto as Appendix 1 and any other agreements regarding confidentiality, protection of intellectual property or nonsolicitation; (ii) any and all indemnification agreements, arrangements or provisions applicable to the Executive including in the Company's Bylaws and that certain Board service related Indemnification Agreement between the Company and the Executive substantially in the form attached hereto as Appendix 2; (iii) the 2011 Founder Club Use Benefit attached hereto as Appendix 3; and (iv) that certain Lease Agreement between Executive and the Company for a unit at 71 Wentworth Street in Charleston, South Carolina.

Page 4 of 8

(f) *Cooperation and Assistance*. For a period of one year from and after the Separation Date, the Executive shall cooperate with and assist the Company, without out-of-pocket expense to the Executive, by sharing the knowledge the Executive has gained during the course of employment with the Company as reasonably requested by the Company.

(g) *Board Resignation.* Executive shall resign from the Inspirato Incorporated Board of Directors effective as of the execution date of the Agreement.

4. <u>No Other Claims</u>. The Executive represents and warrants that:

(a) Executive has no Released Claims pending against the Company or any other Releasee and has not assigned or transferred any Released Claim to anyone;

(b) Executive has been timely paid all compensation owed for services rendered through the Separation Date, including all salary, wages, bonuses, commissions, overtime compensation (if applicable) and payment for all accrued but unused vacation, and has timely received all meal periods and rest breaks to which the Executive may have been entitled;

(c) Executive has been fully reimbursed for all business expenses incurred by the Executive for which the Executive was entitled to reimbursement;

(d) Executive has not suffered any work-related injury or illness as an employee of the Company or any other Releasee and is not aware of any facts or circumstances that would give rise to a workers' compensation claim by the Executive against the Company or any other Releasee; and

(e) Executive has not suffered any sexual harassment or sexual abuse as an employee of the Company or any other Releasee and is not aware of any facts or circumstances that would give rise to such a claim by the Executive against the Company or any other Releasee.

5. <u>Acknowledgements</u>. By signing this Agreement, the Executive acknowledges and agrees that:

(a) the consideration described in Section 1 of this Agreement is consideration to which the Executive would not otherwise be entitled, but for the signing of this Agreement;

(b) Executive has been advised to consult with legal counsel and a tax professional, including with regard to Section 409A, as defined below, about this Agreement and has been given an opportunity to do so;

(c) Executive has been given the amount of time specified in Section 3(a) within which to consider this Agreement before signing it, any changes to this Agreement did not restart the consideration period, and if the Executive has signed this Agreement in less than the specified consideration period, the Executive has done so voluntarily;

Page 5 of 8

- (d) Executive is not relying on any promises or representations of any kind, except those set forth in this Agreement; and
- coercion.

(e)

Executive has signed this Agreement voluntarily, of the Executive's own free will, and without any threat, intimidation or

6. <u>Revocation</u>. The Executive may revoke this Agreement by delivering written notice of revocation to the Company by email, delivery or U.S. Mail addressed as follows, which notice must be received not later than the seventh (7th) calendar day following the Executive's signing of this Agreement, and this Agreement shall not become effective until the seven-day revocation period has expired without revocation by the Executive:

Inspirato LLC 1544 Wazee Street Denver, CO 80202 Email: <u>dprobst@inspirato.com</u> Attn: Danielle Probst

7. <u>Confidentiality</u>. Subject to Section 8, the existence and terms of this Agreement are strictly confidential and shall not be disclosed by the Executive to anyone except (a) the Executive's spouse, attorneys and tax advisors, and then only after securing their agreement to be bound by this provision; or (b) in response to inquiry from a taxing authority or otherwise as required by law.

8. Protected Rights. Notwithstanding any other provision of this Agreement, nothing in this Agreement (or any other agreement signed by Executive) shall restrict Executive's right to (a) report violations of law to law enforcement officials; (b) give truthful testimony under oath in a judicial, administrative, or arbitral proceeding; (c) file a charge with, make truthful statements to, cooperate with investigations by, or assist others in proceedings before governmental agencies (including the U.S Equal Employment Opportunity Commission, the National Labor Relations Board and the U.S Securities and Exchange Commission); (d) speak with an attorney representing Executive; (e) discuss the facts related to any claim of sexual assault or sexual harassment; (f) engage in whistle-blower activity protected by the Securities Exchange Act of 1934, the Dodd-Frank Wall Street Reform and Consumer Protection Act, or any rules or regulations issued thereunder (including Rule 21F-17); or (g) file or disclose any facts necessary to receive unemployment insurance, Medicaid, or other public benefits to which Executive may be entitled. In addition, 18 U.S.C. §1833(b) provides as follows, and nothing in this Agreement or any other agreement, or any Federal or State trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law in a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual and use the trade secret information in the court proceeding, if the individual (A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret pursuant to court order."

State Specific Rights: In the event of any conflict between a provision of this Agreement and applicable state law, state law will govern. Nothing in this Agreement limits Employee's ability to disclose or discuss, either orally or in writing, any alleged discriminatory or unfair employment practice. This Agreement does not prohibit disclosure of information that arises from Employee's general training, knowledge, skill, or experience, whether gained on the job or otherwise, information that is readily ascertainable to the public, or information that Employee otherwise has a right to disclose as legally protected conduct.

Page 6 of 8

9. <u>Invalidity of Release</u>. If any provision of Section 2 of this Agreement is held to be invalid or unenforceable and the Executive is permitted to and does assert any Released Claim against a Releasee, the Company shall be entitled to an immediate refund of one hundred percent (100%) of all payments made pursuant to Section 1 of this Agreement (except that Executive may retain one hundred dollars (\$100)), in addition to any other remedy available to the Company under law or equity; provided, however, that this provision shall not apply to a claim of age discrimination under the ADEA unless ordered by a court of law.

10. <u>Severability</u>. If any provision of this Agreement is held to be invalid or unenforceable, the remaining provisions shall be unaffected and shall continue in full force and effect.

11. <u>No Admission</u>. The parties agree that this Agreement is not an admission, and shall not be construed as an admission, by either Party of any violation of law or other wrongdoing of any kind.

12. <u>Attorney Fees and Costs</u>. In any litigation, arbitration or other proceeding arising out of or relating to this Agreement, the prevailing party shall be entitled to recover his/her/its reasonable attorney fees and costs; provided, however, that this provision shall not apply to a claim of age discrimination under the ADEA or a suit challenging the validity of a release of age discrimination claims under the ADEA.

13. <u>Choice of Law; Forum Selection; Waiver of Jury Trial</u>. This Agreement shall be governed by the laws of the State of Colorado, without regard to any state's principles regarding conflict of laws. Any action arising out of or relating to this Agreement or the Released Claims shall be brought only in the state or federal courts in or for Denver, Colorado, and Executive and the Company hereby waive any right that they might have to challenge the selection of those forums, including but not limited to challenges to personal jurisdiction, venue, or the convenience of the forum. <u>Executive and the Company hereby irrevocably waive their respective rights to a jury trial with respect to any action or claims arising out of or relating to this Agreement or the Released Claims. Executive understands and agrees that any action or claims arising out of or relating to this Agreement or the Released Claims shall be heard only by a judge and not by a jury and that Executive is giving up Executive's right to have any such action or claims heard by a jury.</u>

14. <u>Code Section 409A</u>. This Agreement is intended to comply with Section 409A of the Internal Revenue Code of 1986, as amended ("<u>Section 409A</u>"), or an exemption thereto, and payments may only be made under this Agreement upon an event and in a manner permitted by Section 409A or an exception thereto. Accordingly, this Agreement shall be interpreted in a manner consistent with the requirements of Section 409A to the extent applicable. Any payments under this Agreement that may be excluded from Section 409A either as a short-term deferral or as separation pay due to an involuntary separation from service shall be excluded from Section 409A to the maximum extent possible. All separation payments to be made upon the termination of employment hereunder may only be made upon a "separation from service" within the meaning of Section 409A. Notwithstanding any other provision in this Agreement or in any other document, the Company shall not be responsible for the payment of any applicable taxes incurred by the Executive pursuant to this Agreement, under Section 409A or otherwise. The Company makes no representation that any or all of the payments and benefits described in this Agreement will be exempt from or comply with Section 409A, and the Executive agrees that the Company is not responsible for any liabilities arising to the Executive under 409A.

15. Entire Agreement. This Agreement is the entire agreement between the parties regarding the subjects addressed herein, and it supersedes and replaces all prior agreements, representations, negotiations or discussions between the parties regarding such matters, whether written or oral, including, but not limited to, the Employment Agreement by and between the Executive and the Company dated June 30, 2021. This Agreement may not be modified or amended, nor may any term or provision hereof be waived or discharged, except in a writing signed by both parties. This Agreement may be executed in counterparts, including fax counterparts or in electronic format via DocuSign or similar electronic signature service, and all counterparts together shall constitute one fully- executed agreement.

Page 7 of 8

16. <u>Receipt of Separation Information</u>. Executive acknowledges and agrees that Executive has received the separation information attached hereto as Exhibit A in accordance with the Older Workers Benefit Protection Act.

IN WITNESS WHEREOF, the parties have executed this Agreement on the dates stated below.

NOTE TO EXECUTIVE:

Sign and return by the deadline specified in Section 3(a). Do not sign before last day of employment.

INSPIRATO LLC

By:	Robert Kaiden
Its:	Chief Financial Officer
Date:	
EXEC	UTIVE
By:	
	Brad Handler
Date:	
Page 8 of 8	

<u>Exhibit A</u>

Separation Information pursuant to the Older Workers Benefit Protection Act

See attached.

<u>Appendix 1</u>

Proprietary Rights and Inventions Assignment Agreement

See attached

<u>Appendix 2</u>

Form of Indemnification Agreement

See attached.

Appendix 3

2011 Founder Club Use Benefit

Founders: Brad and Brent Handler

- Membership for life with no annual dues
- Usage of \$60,000 annually or 60 days of usage annually (whichever comes first) at any Inspirato property, hotel or experience offered on the Inspirato web site, for life
- \$60,000 increases annually by CPI plus 4.5% (from 2011)

Separation and Release Agreement

This Separation and Release Agreement ("<u>Agreement</u>") is made between Inspirato LLC, a Delaware limited liability company (the "<u>Company</u>") and Eric Grosse ("<u>Executive</u>") (each a "party" and together the "parties"):

WHEREAS, the Executive was employed by the Company as Chief Executive Officer;

WHEREAS, the Executive has separated from employment with the Company effective August 13, 2024 ("Separation Date");

WHEREAS, the parties wish to ensure an amicable separation and to provide for the release in full of all claims by the Executive;

WHEREAS, the parties mutually agree that Executive's Employment Agreement signed on September 25, 2023 (the "Employment Agreement") is terminated as of the Separation Date and that this Agreement supersedes and replaces the Employment Agreement, including but not limited to any provisions addressing severance pay and benefits upon termination;

NOW, THEREFORE, the parties agree as follows:

1. <u>Separation Benefits</u>. Provided that the Executive complies with all conditions described in Section 3 of this Agreement (the "<u>Conditions</u>"), the Company shall provide the following separation benefits to the Executive:

(a) **Severance Pay**. The Company will pay Executive severance pay in the gross amount of \$550,000.00 (Five Hundred Fifty Thousand Dollars and 00/100), representing payment of twelve (12) months' base salary, less applicable local, state, and federal tax withholdings ("<u>Severance Pay</u>"). Severance Pay payments will be deposited directly into Executive's bank account(s) as Executive has designated in the Company's payroll service. Except as provided in this Section 1(a), Severance Pay will be paid according to the Payment Schedule outlined in Section 1(b).

(b) **Payment Schedule.** Severance Pay will be paid in thirty-six (36) bimonthly installments according to the Payment Schedule as defined below, in accordance with the Company's payroll practice for Colorado employees ("<u>Severance Pay Installments</u>"). Severance Pay Installments shall begin to accrue and, subject to the occurrence of the First Payment Date (as defined below), be paid out on August 30, 2024 and continue for 35 payroll cycles thereafter, subject to all conditions herein The first twelve (12) Severance Pay Installments shall be in the gross amount of \$11,458.33; the following twenty-four (24) Severance Pay Installments shall be in the gross amount of \$17,187.50. The first payment of Severance Pay under this Agreement shall be made to Executive on the first regularly-scheduled Company payday for Colorado employees that is at least eight (8) business days after Executive signs and returns the signed original of this Agreement to the Company, unless revoked in accordance with Section 6 below (the "<u>First Payment Date</u>"). The first payment of Severance Pay under this Agreement shall include a catch-up payment to cover any Severance Pay Installments that accrued prior to the First Payment Date. Severance Pay will not be subject to voluntary employee deferral or employer matching contributions pursuant to any pension or other retirement plan. Each Severance Pay Installment pursuant to this section is intended to constitute a separate payment for purposes of Treasury Regulation Section 1.409A-2(b)(2).

Page 1 of 7

(c) **COBRA**. If the Executive timely elects and is eligible for Continuation Coverage as defined herein, the Company shall pay directly to the Company's COBRA provider or group health plan provider for premiums to continue the medical, dental and vision insurance coverage (if any) of Executive and Executive's eligible dependents pursuant to the continuation-coverage provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 or comparable state law ("*Continuation Coverage*") from the Separation Date through the earlier of (i) the eighteen (18) month anniversary of the Separation Date; (ii) the Executive's eligibility for group medical plan benefits under any other employer's group medical plan; or (iii) the cessation of the Executive's continuation rights under COBRA. Provided, however, if the Company determines that it cannot pay such amounts to the group health plan provider or the COBRA provider (if applicable) without potentially violating applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then the Company shall convert such payments to payroll payments directly to the Executive for the time period specified above. Such payments shall be subject to tax-related deductions and withholdings and paid on the Company's regular payroll dates.

(d) **Equity Securities.** Subject to the Conditions, Executive will be given a one-time grant of 166,667 Restricted Stock Units (the "<u>RSU Grant</u>"). The RSU Grant will become fully vested on the Company's quarterly vesting date that is at least eight (8) business days after Executive signs and returns the signed original of this Agreement to the Company.¹ Upon vesting, the RSU Grant will be settled and released in the method and manner used by the Company.

(e) **Taxes**. The Company may withhold taxes or report taxable income from benefits provided pursuant to this Section 1. The Company makes no representation concerning tax consequences or tax liability that may be incurred by Employee from payments made pursuant to this Agreement. Employee and the Company acknowledge that nothing herein constitutes tax advice to the other party.

2. <u>Release in Full of All Claims</u>. In exchange for the promises described in Section 1 of this Agreement, the Executive, for himself/herself and his/her heirs, assigns and personal representatives, fully and completely releases the Company and its parents, subsidiaries and affiliated entities and all predecessors and successors thereto, and all benefit plans thereof, and all of their respective shareholders, members, partners, directors, officers, managers, employees, attorneys, administrators and agents (each a "<u>Releasee</u>" and collectively the "<u>Releasees</u>") from any and all claims or causes of action that the Executive may have against the Releasees, known or unknown, including claims or causes of action that relate in any way to the Executive's employment with any Releasee or the termination thereof, from the beginning of time through the date the Executive signs this Agreement (each a "<u>Released Claim</u>" and together the "<u>Released Claims</u>"), including but not limited to the following:

(a) federal, state or local laws prohibiting discrimination (including harassment and retaliation) in employment, such as: (i) the Age Discrimination in Employment Act ("<u>ADEA</u>"), the Older Workers Benefit Protection Act, and Executive Order 11141, which prohibit discrimination based on age; (ii) Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1866 (42 U.S.C. § 1981), the Equal Pay Act, and Executive Order 11246, which prohibit discrimination based on race, color, national origin, religion, or sex; (iii) the Genetic Information Nondiscrimination Act, which prohibits discrimination based on disability; (v) the Americans With Disabilities Act and Sections 503 and 504 of the Rehabilitation Act of 1973, which prohibit discrimination based on disability; (v) the National Labor Relations Act, which prohibits discrimination for engaging in certain concerted protected activity; (vii) the Occupational Safety and Health Act and the Mine Safety and Health Act, which prohibit discrimination for engaging in certain safety-related activity; (viii) the Sarbanes Oxley Act, which prohibits discrimination for engaging in certain whistleblowing activity; and (viii) any state or local law that prohibits discrimination on any of the bases described above;

¹ The Company's quarterly vesting dates fall on the 20th day of the second month of each quarter.

Page 2 of 7

(b) federal, state or local laws regarding wages and hours, including laws regarding minimum wage, overtime compensation, wage payment, vacation pay, sick pay, compensatory time, commissions, bonuses, and meal and break periods wages, such as the Fair Labor Standards Act and the Colorado Wage Claim Act (C.R.S. 8-4-101 et seq.);

(c) other employment laws, including but not limited to: (i) the Family and Medical Leave Act and analogous state and local laws, which require employers to provide leaves of absence under certain circumstances; (ii) the Worker Adjustment and Retraining Notification Act (WARN) and analogous state and local laws, which require advance notice of certain workforce reductions; (iii) the Employee Retirement Income Security Act, which protects employee benefits (among other things); and (iv) the Uniformed Services Employment and Reemployment Rights Act, which requires employers to provide military leave under certain circumstances; and

(d) any common law theory, including but not limited to breach of contract (expressed or implied), promissory estoppel, wrongful discharge, outrageous conduct, defamation, fraud or misrepresentation, tortious interference, invasion of privacy, negligent hiring or supervision, or any other claims based in contract, tort or equity.

Notwithstanding the foregoing, the Released Claims do not include claims for breach of this Agreement, claims related to rights to indemnification or insurance the Executive has pursuant to contractual arrangements with the Company or its parent, corporate documents of the Company or its parent, claims that arise after the Executive signs this Agreement, claims for vested pension benefits, claims for workers' compensation benefits or unemployment compensation benefits, and any other claims that cannot by law be released by private agreement. In addition, this release does not prevent the Executive from filing: (i) a lawsuit to challenge the effectiveness of a release of claims of age discrimination under the ADEA; or (ii) a charge with a governmental agency, including but not limited to the U.S. Equal Employment Opportunity Commission ("EEOC") and the U.S. Securities and Exchange Commission ("SEC"), but the Executive is waiving his/her right to recover any monetary or injunctive relief pursuant to any such charge. Notwithstanding the foregoing, this Agreement does not prevent the Executive from recovering an award from or by a governmental agency for providing information.

Executive acknowledges and agrees that the Executive is releasing both known and unknown claims and waives the benefit of any statute purporting to prevent the Executive from releasing unknown claims, including but not limited to the protection of Cal. Civ. Code Section 1542, which states:

A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party.

3. <u>Conditions</u>. The Executive shall comply with the following terms, as conditions of payment, and in the event that the Executive fails to satisfy these conditions, the Company shall have no obligation to provide any separation benefits pursuant to Section 1 and shall be entitled to a refund of any separation benefits previously paid:

(a) *Execution and Return of Agreement*. The Executive shall sign this Agreement and return the signed original of the Agreement to the Company within forty-five (45) days after the Executive Receives the Agreement and shall not revoke it, except as otherwise described in Section 6, below.

Page 3 of 7

(b) *Property*. The Executive shall return all Company property in the Executive's possession, custody or control on or before the Separation Date, including but not limited to all motor vehicles, computer hardware, office equipment, tools, telephones, credit cards, keys, card keys, and the originals and all copies of all documents, files, computer software and electronic data (including any and all copies of the Company's proprietary software, commensurate source code and engineering designs); provided, however, that the Executive may retain copies of documents reflecting the Executive's compensation and benefits from the Company. By signing this Agreement, the Executive represents and warrants that the Executive has fully complied with this Section 3(b).

(c) *Non-Disparagement.* Subject to Section 8, to the fullest extent permitted by law, Executive shall not engage or in any conduct or make any statement calculated to or likely to have the effect of undermining, disparaging or maligning the Releasees or the Company's business, products, services, customers or clients. By signing this Agreement, Executive represents and warrants that Executive has made no statements on or after the Separation Date that would violate this Section, if made after this Agreement has become effective. By signing the General Release, Employee represents and warrants that Employee has fully complied with this Section 3(c).

(d) Other Agreements. Subject to Section 8, the Executive shall fully comply with all other agreements between the Executive and the Company (or any parent, subsidiary or affiliate of the Company or predecessor or successor thereto), including but not limited to the Proprietary Rights and Inventions Assignment Agreement attached hereto as Appendix 1, and any other agreements regarding confidentiality, protection of intellectual property or nonsolicitation (each an "Other Agreement").

(e) *Cooperation and Assistance*. For a period of one year from and after the Separation Date, the Executive shall cooperate with and assist the Company, without out-of-pocket expense to the Executive, by sharing the knowledge the Executive has gained during the course of employment with the Company as reasonably requested by the Company.

(f) *Board Resignation.* Executive shall resign from the Inspirato Incorporated Board of Directors effective as of the execution date of the Agreement.

4. <u>No Other Claims</u>. The Executive represents and warrants that:

(a) Executive has no Released Claims pending against the Company or any other Releasee and has not assigned or transferred any Released Claim to anyone;

(b) Executive has been timely paid all compensation owed for services rendered through the Separation Date, including all salary, wages, bonuses, commissions, overtime compensation (if applicable) and payment for all accrued but unused vacation, and has timely received all meal periods and rest breaks to which the Executive may have been entitled;

(c) Executive has been fully reimbursed for all business expenses incurred by the Executive for which the Executive was entitled to reimbursement;

(d) Executive has not suffered any work-related injury or illness as an employee of the Company or any other Releasee and is not aware of any facts or circumstances that would give rise to a workers' compensation claim by the Executive against the Company or any other Releasee; and

(e) Executive has not suffered any sexual harassment or sexual abuse as an employee of the Company or any other Releasee and is not aware of any facts or circumstances that would give rise to such a claim by the Executive against the Company or any other Releasee.

Page 4 of 7

5. <u>Acknowledgements</u>. By signing this Agreement, the Executive acknowledges and agrees that:

(a) the consideration described in Section 1 of this Agreement is consideration to which the Executive would not otherwise be entitled, but for the signing of this Agreement;

(b) Executive has been advised to consult with legal counsel and a tax professional, including with regard to Section 409A, as defined below, about this Agreement and has been given an opportunity to do so;

(c) Executive has been given the amount of time specified in Section 3(a) within which to consider this Agreement before signing it, any changes to this Agreement did not restart the consideration period, and if the Executive has signed this Agreement in less than the specified consideration period, the Executive has done so voluntarily;

- (d) Executive is not relying on any promises or representations of any kind, except those set forth in this Agreement; and
- (e) Executive has signed this Agreement voluntarily, of the Executive's own free will, and without any threat, intimidation or coercion.

6. <u>Revocation</u>. The Executive may revoke this Agreement by delivering written notice of revocation to the Company by email, delivery or U.S. Mail addressed as follows, which notice must be received not later than the seventh (7th) calendar day following the Executive's signing of this Agreement, and this Agreement shall not become effective until the seven-day revocation period has expired without revocation by the Executive:

Inspirato LLC 1544 Wazee Street Denver, CO 80202 Email: dprobst@inspirato.com Attn: Danielle Probst

7. <u>Confidentiality</u>. Subject to Section 8, the existence and terms of this Agreement are strictly confidential and shall not be disclosed by the Executive to anyone except (a) the Executive's spouse, attorneys and tax advisors, and then only after securing their agreement to be bound by this provision; or (b) in response to inquiry from a taxing authority or otherwise as required by law.

8. Protected Rights. Notwithstanding any other provision of this Agreement, nothing in this Agreement (or any other agreement signed by Executive) shall restrict Executive's right to (a) report violations of law to law enforcement officials; (b) give truthful testimony under oath in a judicial, administrative, or arbitral proceeding; (c) file a charge with, make truthful statements to, cooperate with investigations by, or assist others in proceedings before governmental agencies (including the U.S Equal Employment Opportunity Commission, the National Labor Relations Board and the U.S Securities and Exchange Commission); (d) speak with an attorney representing Executive; (e) discuss the facts related to any claim of sexual assault or sexual harassment; (f) engage in whistle-blower activity protected by the Securities Exchange Act of 1934, the Dodd-Frank Wall Street Reform and Consumer Protection Act, or any rules or regulations issued thereunder (including Rule 21F-17); or (g) file or disclose any facts necessary to receive unemployment insurance, Medicaid, or other public benefits to which Executive may be entitled. In addition, 18 U.S.C. §1833(b) provides as follows, and nothing in this Agreement or any other agreement, or any Federal or State trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal; and (B) does not disclose the trade secret information in the court proceeding, if the individual (A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret pursuant to court order."

Page 5 of 7

State Specific Rights: In the event of any conflict between a provision of this Agreement and applicable state law, state law will govern. Nothing in this Agreement prevents Employee from discussing or disclosing information about unlawful acts in the workplace, such as harassment or discrimination or any other conduct that Employee has reason to believe is unlawful. Executive has the right to consult with an attorney regarding this Agreement.

9. <u>Invalidity of Release</u>. If any provision of Section 2 of this Agreement is held to be invalid or unenforceable and the Executive is permitted to and does assert any Released Claim against a Releasee, the Company shall be entitled to an immediate refund of one hundred percent (100%) of all payments made pursuant to Section 1 of this Agreement (except that Executive may retain one hundred dollars (\$100)), in addition to any other remedy available to the Company under law or equity; provided, however, that this provision shall not apply to a claim of age discrimination under the ADEA unless ordered by a court of law.

10. <u>Severability</u>. If any provision of this Agreement is held to be invalid or unenforceable, the remaining provisions shall be unaffected and shall continue in full force and effect.

11. <u>No Admission</u>. The parties agree that this Agreement is not an admission, and shall not be construed as an admission, by either Party of any violation of law or other wrongdoing of any kind.

12. <u>Attorney Fees and Costs</u>. In any litigation, arbitration or other proceeding arising out of or relating to this Agreement, the prevailing party shall be entitled to recover his/her/its reasonable attorney fees and costs; provided, however, that this provision shall not apply to a claim of age discrimination under the ADEA or a suit challenging the validity of a release of age discrimination claims under the ADEA.

13. <u>Controlling Law; Venue</u>. This Agreement shall be governed by the laws of the State of California, Executive's principal work location as of the Separation Date. Any action arising out of or relating to this Agreement shall be brought only in the state or federal courts in or for Los Angeles, California and the Executive and the Company hereby waive any right that they might have to challenge the selection of those forums, including but not limited to challenges to personal jurisdiction, venue, or the convenience of the forum.

14. <u>Code Section 409A</u>. This Agreement is intended to comply with Section 409A of the Internal Revenue Code of 1986, as amended ("<u>Section 409A</u>"), or an exemption thereto, and payments may only be made under this Agreement upon an event and in a manner permitted by Section 409A or an exception thereto. Accordingly, this Agreement shall be interpreted in a manner consistent with the requirements of Section 409A to the extent applicable. Any payments under this Agreement that may be excluded from Section 409A either as a short-term deferral or as separation pay due to an involuntary separation from service shall be excluded from Section 409A to the maximum extent possible. All separation payments to be made upon the termination of employment hereunder may only be made upon a "separation from service" within the meaning of Section 409A. Each amount to be paid or benefit provided under this Agreement shall be construed as a separate identified payment for purposes of Section 409A. Notwithstanding any other provision in this Agreement or in any other document, the Company shall not be responsible for the payment of any applicable taxes incurred by the Executive pursuant to this Agreement, under Section 409A or otherwise. The Company makes no representation that any or all of the payments and benefits described in this Agreement will be exempt from or comply with Section 409A, and the Executive agrees that the Company is not responsible for any liabilities arising to the Executive under 409A.

Page 6 of 7

15. <u>Entire Agreement</u>. This Agreement is the entire agreement between the parties regarding the subjects addressed herein, and it supersedes and replaces all prior agreements, representations, negotiations or discussions between the parties regarding such matters, whether written or oral, including, but not limited to, the Employment Agreement by and between the Executive and the Company dated June 30, 2021. This Agreement may not be modified or amended, nor may any term or provision hereof be waived or discharged, except in a writing signed by both parties. This Agreement may be executed in counterparts, including fax counterparts or in electronic format via DocuSign or similar electronic signature service, and all counterparts together shall constitute one fully- executed agreement.

16. <u>Receipt of Separation Information</u>. Executive acknowledges and agrees that Executive has received the separation information attached hereto as Exhibit A in accordance with the Older Workers Benefit Protection Act.

IN WITNESS WHEREOF, the parties have executed this Agreement on the dates stated below.

NOTE TO EXECUTIVE:

Sign and return by the deadline specified in Section 3(a). Do not sign before last day of employment.

INSPIRATO LLC

By:	
	Robert Kaiden
Its:	Chief Financial Officer
Date:	August 13, 2024
EXECUTIVE	
By:	
	Eric Grosse
Date:	August 13, 2024
Page 7 of 7	